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ABSTRACT

Critical components in coal-fired power plants such as final superheater heat exchangers experience
severe conditions associated with high metal temperatures and high temperature gradients during base
load and transient operations. Such adverse conditions could significantly reduce the life span of the
components, especially due to the requirement of greater plant flexibility that is an essential part of the
global power system transformation. Integrated thermofluid process models can be employed to obtain
a better understanding of the relationship between the operational conditions and the metal
temperatures. Thus, a methodology was developed to model radiant superheater heat exchangers in
steady state and transient operations.

The methodology is based on a network approach which entails solving the transient one-dimensional
forms of the conservation equations for mass, energy and momentum. The model building blocks
account for the convective thermal resistance on the steam side, the conductive thermal resistances of
the tube wall and scaling or fouling on the tube walls, as well as the convective and radiative thermal
resistances and direct radiation on the flue gas side. The model captures the physical layout of the tube
passes in a tubesheet via the arrangement of the network building blocks. It is also possible to connect
tubesheets together across the width of the boiler as per the arrangement in a real plant.

The modelling methodology was first used to develop a process model of a convective cross-flow
primary superheater heat exchanger with complex flow arrangement. The dual-tube 12-pass
superheater was discretized along the flue gas flow path as well as along the steam flow path. The
model was qualitatively validated using real plant data from literature and for reference purposes also
systematically compared to conventional lumped parameter models. The ability of the model to analyse
the effect of ramp rate during load changes on the tube metal temperature was demonstrated, as well
as the ability to determine the maldistribution of flow and temperature on the steam and flue gas sides.

The methodology was also applied to model a U-shaped radiant superheater heat exchanger. Due to
the challenges associated with obtaining comprehensive real plant data in an industrial setting, a
validation methodology was proposed that is based on a combination of plant design C-schedules and
a boiler mass and energy balance, as well as limited plant measurements. The consistent comparisons
with C-schedule data provide evidence of the validity of the model, which was further demonstrated
via the comparisons with real plant data. The model allows prediction of the steam mass flow and
temperature distribution going into the outlet stub headers as well as the main outlet headers for
different inlet flow and temperature distributions on the steam and flue gas sides. These results were
compared to detail real-plant measurements of the outlet header temperatures. The model also allows
prediction of the metal temperatures along the length of the tubes which cannot readily be measured
in the plant. The model was applied to demonstrate the impact of different operational conditions on
the tube metal temperatures.

Such integrated process models can be employed to study complex thermofluid process phenomena
that may occur during intermittent, transient and low load operation of power plants. In addition, such
models could be useful for predictive and preventative maintenance as well as online condition
monitoring.
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Chapter 1. Introduction

1. INTRODUCTION

Boiler tube failures are a leading cause of unplanned capacity losses in South African coal fired power
plants, as illustrated in Figure 1-1. These boiler tube failures are predominantly on the components that
operate under high pressures and temperatures, i.e. the superheater and reheater heat exchangers. The
high pressure and temperature steam escaping from a failed tube can cut the tubes around it hence
increasing the maintenance and replacement costs. In addition to component replacement costs, the
down time of a plant leads to revenue losses for the utility. In many cases operational anomalies are

major contributors to these tube failures.

16

14 -

0 I I I I I

Boiler Tubes Combustion Chamber ID fans Steam circuit Gas cleaning

—_ —_
S \S]

% Contribution to the UCLF
oo

Figure I-1: A breakdown of the top five contributions of boiler subsystems or components to the Unplanned

Capacity Loss Factor (UCLF) in Eskom Coal fired fleet, Year to Date January 2015 [5].

According to Babcock & Wilcox, for coal-fired boilers it is possible to repair a tube leak and put the
boiler back online, only to be forced offline within a short time by another tube leak. This emphasizes
the need to identify and correct the root cause of the problem because a tube failure may be a symptom
of other problems. Some knowledge of the different kinds of tube failures (their visual characteristics)
and their causes does assist in the root cause analyses of the problem. These failures include the
following: caustic attack, oxygen pitting, hydrogen damage, acid attack, stress corrosion cracking,

waterside corrosion fatigue, superheater fireside ash corrosion, high-temperature oxidation, water
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fireside corrosion, fireside corrosion fatigue, graphitization, dissimilar metal weld failure, tube erosion,

short-term overheat, long-term overheat and thermal fatigue [6].

Short-term overheating is the ductile rupture of tube metal due to elevated temperatures. These high
temperatures may be caused by lack of cooling from water or steam flow, especially during boiler
start-ups. For example, if the condensate in boiler superheater tubes is not evaporated, this condensate
will act as a blockage, hence starving certain sections of the tubes from cooling by steam [6]. This type
of failure is characterised by a thin lipped “fish mouth” opening along the length of the tube as shown

in Figure 1-2.

Figure 1-2: Thin lipped “fish mouth” opening characterizing short-term overheating.

Long-term overheating in tube metals, also known as creep failure, occurs after months or years of
operation in elevated temperatures. This occurs especially in water walls, superheater and reheater
tubes where during normal operation the metal temperature will degrade due to the high temperatures
and pressures over the life of the component. Typical operation hours of such tube materials are about
200 000 hours. After such a time, the tube experiences creep rupture which is characterized by a thick-

lipped opening as shown in Figure 1-3.

Heat exchangers in the convective pass of a coal-fired power plant boiler are made up of tubes which
form the elements or tube sheets. The steam flowing in these tubes are either collected directly in a set
of main headers, or in so-called stub headers from where it is fed to the main headers, as illustrated in

Figure 1-4.
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Main header

/ Connecting pipe

Stub header

Left hand side

™_Heat exchanger
element

Right hand side

Figure 1-4: A three-dimensional CAD drawing of a superheater heat exchanger with sub headers.

Final superheater outlet headers are susceptible to failure because they operate under high pressures
and temperatures. Some headers operate at temperatures close to 600 °C. These headers are typically
made of P11 and P22 materials [7]. Even though these headers are designed to operate below the

temperature limit and by taking creep failure into account, over time the material degrades, creep
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damage accumulates, and material failure is inevitable. Damage is aggravated when the boiler
undergoes cyclic operations such as start-ups, shut-down and load changes, because the time to failure
due to creep can be accelerated. This is due to the fact that the cyclic operation introduces additional

damage mechanisms, namely thermal fatigue and oxide notching [7].

During cyclic operations, the variation of the steam temperature induces a temperature gradient along
the thickness of the header wall. Such a temperature gradient in turn introduces thermal stresses. When
there is continuous variation of the temperatures it results in cycling thermal stresses which then induce
thermal fatigue in the radial direction. The introduction of more intermittent renewable energy sources
on the electricity supply grid means that coal-fired plants are forced to change its mode of operation
from base load to two-shifting or low load variable operation [8, 9]. This is also the case in South

Africa where times of intermittent surplus capacity amid weaker demand are becoming more prevalent.

Since subsystems like superheaters have elements connected along the length of the header, if the
temperatures in these elements are different such that certain sections of the header expand more than
others, bending stresses can be introduced. If these bending stresses are cycled, then thermal stresses
in the axial direction can be introduced leading to tube-to-header weld cracks as well as failure on the

header itself, as shown in Figure 1-5.

Figure 1-5: Failed header due to bending stresses [10].

According to Nakoneczny & Schultz [7] cracking in high temperature superheater headers has been
found to occur in almost every weld position, as well as the ligament area between tube stub bore
holes. Figure 1-6 shows some of the locations on a header that are susceptible to cracking. In general,

header damage can be classified as repairable or non-repairable. Repairable damage mainly consists
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of cracks or any failure that can be repaired through welding. Non-repairable damage consists of

ligament (or bore hole) cracks which will require a replacement of the header.

Tube to Hoader Thermowell tug Handhole
Fitting

Outlet Tubes

Figure 1-6: Locations on the header that are susceptible to cracking [7].

Combustion characteristics, steam flow and boiler load variations are three factors related to boiler
operations that influence ligament damage in headers [7]. In terms of the influence from combustion,
the arrangements of the burners on the furnace walls has an influence on the flue gas flow and
temperature leaving the furnace to the convective pass. There are two configurations of burners,
namely wall-fired and tangentially fired. For wall-fired boilers, the burners are on the front wall and/or
the rear wall depending on the size of the unit. For tangentially fired boiler the burners are situated at
the corners of the furnace. For each configuration there is an associated Furnace Exit Gas Temperature
(FEGT) and flow profile which have an effect on the heat absorption of the downstream heat
exchangers in the convective pass. An example of a typical FEGT profile for a wall fired boiler is

shown in Figure 1-7c.
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c¢) Furnace flue gas exit temperature

Flux (kW/m?)
-160.5

-1249

-89.40

-53.79

\

-18.19

Flux (kW/m?)
T 219.6

|

1176

15.54

-86.50

-188.5

665.7

630.8

595.9

Figure 1-7: An example of a furnace exit gas temperature profile resulting from a wall-fired boiler. a) left

wall flux, b) front wall flux and c) furnace exit gas temperature [11].

Since some of the heat exchangers in the convective pass are enclosed by the water walls and heat is

also absorbed on these water-cooled walls, the temperature of the tubesheets near the side of the heat

exchangers is lower than that of the inner tubesheets. Slagging on the furnace water walls and fouling

on the heat exchangers in the convective pass also affect the temperature and flow distribution on the

superheaters and reheaters. Fouling on the superheaters or reheaters is mostly experienced by the tube

sheets in the middle of the heat exchanger, thus contributing to the creation of an M-type steam

temperature profile on the outlet header. An example of such an M-type steam temperature profile is

given in Figure 1-8.

University of Cape Town



Tube Leg

Temgerature

1070F
{577C)

Temperature

Chapter 1. Introduction

teft End

Figure 1-8: Steam temperature variation in a header [7].

Tube Leg Location

Right End

In terms of the effect of boiler loads, Figure 1-9 demonstrates typical variations of tube leg temperature

and average header temperature corresponding to load changes. For example, in the case of a load

ramp up (from O to A) the boiler firing rate must increase to maintain the boiler pressure. In this period

the boiler is temporarily over-fired to compensate for the effect of increased steam mass flow and

boiler pressure drop. Therefore, during this period the boiler tube leg temperature will be higher than

the average header temperature. This temperature difference on the header results in localized stresses

which are much greater than those related to the steam pressure. The temperature differences also occur

during ramp down. These stresses contribute to header crack initiation, especially along the bore hole

penetration, which eventually lead to premature header failure [7].
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o Boiler Load, MW

— — — — ——— —————

Temperature, F

g
i

Time

Figure 1-9: Superheater tube leg temperature variation with load [12].

There are some other quasi-steady state operational phenomena which can influence the lifetime of
some components, which will be addressed below. In order to generate more power or less power from
a unit of a coal fired power plant, the steam mass flow rate is increased or decreased, respectively as
alluded to above. If slag builds up on the waterwalls of the combustion chamber, the furnace gas exit
temperature increases. This implies that the heat exchangers downstream in the convective pass
(superheaters, reheaters and economizer) might experience higher temperatures than they were
designed for, hence an increase in the amount of heat absorbed. Thus, in order to maintain the designed
outlet superheated steam temperature attemperation water has to be increased. If a heat exchanger in
the convective pass is fouled, the amount of heat it absorbs drops thus more coal must be burnt to get
the desired outlet steam temperature. Once the fouled heat exchanger has been cleaned using soot
blowers, the heat it absorbs increases thus less coal must be burnt. All these factors that have an
influence on the performance of the heat exchangers in the convective pass emphasise the importance
of capturing the behaviour of these heat exchangers. This will help to improve the quality of control
for the superheated steam temperature as well as to analyse the effect of such factors on the conditions

of the tubes and headers of the heat exchanger.

University of Cape Town 8



Chapter 1. Introduction

1.1. Problem statement

From the discussion above it should be evident that in coal-fired power plants, final stage superheaters
may experience severe conditions associated with high metal temperatures, high temperature gradients
and cycling during transient operations. The outlet headers of the final superheaters are especially
susceptible to these problems. Failure of such components results in an increase in maintenance costs
for the plant. So, as a first step to determine how the operational conditions affect the life span of the
components, a better understanding is required of the relationship between the operational conditions
and the metal temperatures. For this, it is necessary to be able to model the thermofluid processes of

final stage radiant superheaters in steady state and transient operations.

There is therefore a need for a thermofluid process model that can predict the flow and heat transfer of
the steam inside the heat exchanger tubes and headers, the heat conduction through the tube and header
walls, as well as the flue gas flow outside the tubes, in an integrated manner. The results should be
provided in such a way that it facilitates the study of the resultant thermal stresses in the tube and
header walls to better understand, predict and manage how the operational conditions affect the life

span of the components.

The model should strike an appropriate balance between simplicity and accuracy so that it may be
applied in real-life to gain an understanding of typical problems and to investigate the risks and impacts
associated with intermittent operation, load following and variations in coal quality. This, together
with appropriate and sufficiently accurate measurements of key parameters, could inform decision
making for operating and maintenance strategies to improve component availability and reliability,

and to reduce the cost of power produced.

1.2. Project hypothesis

The hypothesis of this research is that a network approach can be employed to construct integrated
thermofluid process models of radiant superheater heat exchangers that provide results of outlet steam
flows and temperatures that can be used as boundary values for Computational Fluid Dynamics (CFD)
and Finite Element Method (FEM) analyses to determine the resultant stresses in headers. This will in
turn serve to better understand, predict and manage how steady state and transient operational

conditions affect the life span of these components.
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1.3. Project objectives

The overall objective of this project is to develop a modelling methodology for radiant superheaters
heat exchangers in steady state and transient operations. The methodology should integrate the
thermofluid phenomena occurring in the superheater, which includes the fluid flow and heat transfer

phenomena.
The enabling objectives of this work are:

¢ Develop a methodology to model the thermofluid processes of a heat exchanger in the
convective pass of a coal-fired boiler.

+ Apply the methodology in modelling a specific heat exchanger.

¢ Develop a methodology to validate the developed thermofluid process model.

+ Verify and validate the developed heat exchanger model.

¢ Investigate the effect of operational anomalies on the tube metal and steam temperatures.

¢ Investigate the effect of operational anomalies on the flue gas and steam flow distribution.

+ Investigate the effect of transients such as load changes on the tube metal and steam

temperatures.

1.4. Project scope

The scope of this project is to develop a modelling methodology for radiant superheaters based on the
one-dimensional network approach supported with high-level models to obtain input data for the heat
exchanger models. These high-level models are applied to the other sub-systems of the boiler that are
not of direct interest in the analyses, but still influences some of the conditions in the heat exchanger

of interest.

The purpose of this study is not to produced three-dimensional computational fluid dynamics models
of the superheater heat exchangers in coal fired boilers, or to determine the actual stresses in the

headers.

1.5. Format of the report

Chapter 1 presented the problem statement, the project hypothesis and the objectives of the project.

Chapter 2 presents a review of literature associated with the modelling of heat exchangers, mainly in

coal-fired boilers, for both steady state and transient operations.
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In Chapter 3 the network modelling methodology is developed. Chapter 4 presents a demonstration
case study, where the network methodology is applied to a convective primary superheater heat

exchanger. The geometrical and performance data for this case study was obtained from the literature.

Chapter 5 explores the flue gas side modelling in more detail and considers some limitations associated

with the one-dimensional network approach when modelling three dimensional phenomena.

Chapter 6 discusses further refinement of the modelling methodology, which is then applied to a
radiant final superheater heat exchanger. The performance data for this case study was obtained from
real plant measurements. The chapter also presents the high-level models needed for generating
additional flue gas side input data to the radiant superheater model, as well as the validation of the
modelling methodology based on plant design C-schedule data. Chapter 7 then presents the application

case study of the refined model, where the real plant data is employed.

Chapter 8 concludes the report and presents some recommendations for future work.
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2. LITERATURE REVIEW

This chapter presents a review of literature on the modelling of heat exchangers with a focus on heat
exchangers in the convective pass of coal-fired boilers. The chapter begins by looking at simplified
modelling methods which are mainly used to assess the overall performance of the heat exchanger.
Two-dimensional (2-D) models which model one tubesheet/serpentine/element are then reviewed.
Since a heat exchanger is a three-dimensional (3-D) system, some 3-D modelling methodologies and
techniques are also reviewed. In addition, literature on the simulation of the dynamic behaviour of heat
exchangers in coal fired boilers is reviewed. The chapter finishes with the review of literature dealing

with the modelling of the flue gas side of the heat exchangers.

2.1. Lumped heat exchanger modelling

Thermo-fluid process models are often employed to determine the performance of a heat exchanger
and sometimes also for online condition monitoring. In many instances the complete heat exchanger
is lumped together and viewed as a generally counter-current flow, co-current flow, cross-flow or

hybrid flow arrangement.

The most common simplified or lumped methodologies for modelling heat exchangers are the
Logarithmic Mean Temperature Difference (LMTD) method and the effectiveness-number of transfer
unit (NTU) method [13, 14]. For the LMTD method an experimentally determined correction factor is
used to account for different types of heat exchangers or different flow arrangements. For the
effectiveness-NTU method, different correlations are employed for the effectiveness-NTU relationship
of different heat exchanger layouts. Kays and London [15] graphically presented the relationship

between the NTU and the effectiveness for several different heat exchangers.

Diez et al. [16] highlighted the importance of lumped models for online calculations. Online
calculations require simple models which are quick to run. They modified the conventional LMTD
lumped model and integrated it with offline CFD predictions to improve the online modelling
technique. The LMTD model was modified to account for direct radiation fluxes. Cantrell and Idem
[17] employed the effectiveness-NTU method to the model the online performance of heat exchanger
assemblies in a typical coal-fired boiler. The model calculated instantaneous heat transfer in different
sections of the boiler to determine the local cleanliness factor. Information obtained from such models

was used to make informed decisions on which section of the convective pass was to be cleaned.
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There are limitations to these lumped models, especially if a detail analysis of the heat exchanger
performance is required. They also cannot model with accuracy heat exchangers with complex flow
arrangements. These models are usually not set up in such a way that the flow and temperature
maldistribution amongst the tubes and within the headers can be identified. They also do not account

for pressure drop and/or momentum conservation.

2.2. Single tubesheet modelling

In addressing the limitation of lumped models in capturing the geometry of a heat exchanger, more
detailed models of one whole tubesheet of the heat exchanger may be developed. For these models the
assumption is that there is no variation of flow and temperature amongst the different tubesheets along

the width of the boiler.

Taler et al. [4] used mass and energy conservation equations to develop a one-dimensional (1-D)
transient numerical heat transfer model for a dual-tube 12 pass primary superheater heat exchanger
tubesheet of a 50-MW coal-fired utility boiler. The superheater tubesheet was discretised into a
network of 1-D control volumes in the flow path connected such that they capture the complex flow

arrangements as shown in Figure 2-1.
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Figure 2-1: Heat exchanger flow arrangement and division into control volumes [4].

The flue gas side flow was assumed to be flowing in parallel flow channels, hence no mixing. For each

pass the two tubes in parallel were lumped together and modelled as a single larger tube with equivalent

University of Cape Town 13



Chapter 2. Literature review

flow and heat transfer areas. The transient form of the 1-D partial differential equations for mass and
energy conservation were employed for each control volume, including the heat transfer between the
steam and flue gas streams. They employed a finite-volume method (FVM) to obtain the discretized
equations and an explicit (forward) Euler method was then proposed for the temporal integration. The
Gauss-Seidel method was used to iteratively solve the system of equations. However, they did not

present results of transient analyses for this case.

The model used the convection heat transfer coefficient correlation given by Trojan and Taler [18] and
Taler et al. [19]. Radiation heat transfer was accounted for through an effective heat transfer coefficient
which was proposed by Taler and Taler [20]. This radiation heat transfer coefficient was then added
to the convection heat transfer coefficient resulting in an equivalent heat transfer coefficient on the
flue gas side. In the initial version of the model, momentum conservation was not taken into account.
In an updated version of the model presented by Trojan and Taler [18] the pressure drops were also
taken into account. The simplifying assumptions applied in the model by these authors are as follows:
the steam and flue gas flow is 1-D; the flue gas flows in channels with no mixing; the physical
properties of fluids are functions of temperature; axial conduction in tube wall is negligible; fluid
conduction in the direction of flow is negligible; the flue gas and steam temperatures and the flue gas
velocity are assumed to be uniform over the cross section at the inlet; and heat transfer coefficients on

the inner and outer surfaces are uniform across the width of the heat exchanger [18].

The model of Taler et al. has the capability to account for scaling and an ash deposit layer of constant
thickness. The ash deposit layer was accounted for separately, but the scaling layer can only be
accounted for through a corrected internal heat transfer coefficient. Great care needs to be taken in
accounting for the geometry on the inside of the tube if scaling is present. The model was verified
against an analytical model of a one tube heat exchanger. It was then applied on a convective
superheater of a coal-fired boiler, simulating the distribution of steam, tube wall, soot layer and flue
gas temperatures on the different tubes in steady state for different scenarios. Their results were
compared to real plant data for a single design case. They assumed a uniform thickness distribution of
the ash fouling layer and to obtain good comparison with plant data they adjusted the fouling layer
thickness in the model. This demonstrates one of the major difficulties in using real plant data for
validation. The distribution and thickness of the fouling layer is not known and practically impossible

to measure, and it changes continuously during operation.
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Trojan and Taler [18] also investigated the effect of an uneven flue gas temperature distribution across
the width of the superheater. This was done by modelling the one tubesheet that experienced the highest
flue gas temperature and then comparing the results to that of a tubesheet from the case with uniform
flue gas temperatures which was experiencing lower temperatures. In these models, the relationship
between the steam mass flow and pressure drop was highlighted. In order to maintain the same pressure
drop in both these models the steam mass flow rate in the case with higher flue gas temperature was
reduced. This therefore approximates the effect of steam mass flow maldistribution. However, they
did not connect all the different tubesheet models across the width of the header and include the
integrated solution of the momentum conservation equations. Thus, the flow rate needs to be specified
for each of the flow increments and the maldistribution in the flow cannot be predicted, either on the
steam side or on the flue gas side. They also adjusted the fouling layer thermal resistance until they
obtained a good comparison between the calculated and measured steam outlet temperature for steady

state scenarios. They did not present any results of transient analysis.

Taler et al. [21] and Trojan et al. [22] combined the superheater model of Taler et al. [4] with an
evaporator model to model the larger boiler for analysing the impact of ash and scaling. The model
was applied as a boiler slagging and fouling simulator and was used successfully together with on-line
measurements to guide soot blower operations for the combustion chamber and superheater heat

exchangers in Skawina Power Plant in Poland.

The radiation heat transfer coefficient of Taler and Taler [20] was derived on the basis of a diffusivity
model of radiation heat exchange. Their radiation heat transfer coefficient calculation method was
compared to other methods using the commercial CFD package FLUENT 6.2 based on the Discrete
Ordinates method and the Discrete Transfer Radiation Method. It showed good accuracy for a wide
range of flue gas temperatures, making it suitable for both radiant type and convective type heat

exchangers in the convective pass of a coal-fired boiler.

Trojan [23] modelled in detail the same superheater as that of Taler et al. [4] using the commercial
CFD package, ANSYS-CFX. HyperMesh 11 was used to create the mesh for the CFD calculations.
For verification purposes two meshes were used: a mesh of 7 809 105 elements and a slightly finer
mesh of 8 490 342 elements. The k - £ turbulence model was used to model the flue gas and steam
flows. The model only focused on one tubesheet. Local and mean values of flue gas temperature after
the primary superheater, tube wall temperatures and outlet steam temperatures were determined for

both clean and fouled tubes. The outlet steam temperature results were verified against results from
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the Taler et al. [4] model as well as validated against plant measurement data. The model clearly
showed areas with the maximum tube wall temperatures. The comparison of the results from the two
models with different meshes was satisfactory proving that the meshes was selected properly. Taler et
al. [24] used the CFD superheater model of Trojan [23] to verify a numerical superheater model based
on the energy conservation equation similar to the model of Taler et al. [4]. The heat exchanger overall

performance results of the two models compared well.

Coelho [25] presented the theory of a 2-D mathematical model for simulating flow and heat transfer
in the convection chamber of a utility boiler in steady state. The flow field simulation was based on
the mass and momentum conservation equations in conjunction with the CFD porous media approach
to account for tube bundles and control dampers. The k — £ eddy viscosity/diffusivity model was
employed to cater for turbulence. The pressure drop of the cross flow through the tube bundles was
estimated using the correlation of Zukauskas [26]. Energy balances were performed along the tubes to
account for convective heat transfer and fouling resistance on both sides of the tube, as well as radiation
on the gas side. A highly simplified zonal method presented by Coelho [27] was used to incorporate

radiation heat transfer. The flow and heat transfer are iteratively coupled by the solution algorithm.

The simplified formulation for calculating radiative heat transfer based on the zonal method presented
by Coelho [27] conceptually divided the convection chamber into cavities and heat exchangers.
Fictitious planes were placed immediately upstream and downstream of every heat exchanger as shown
in Figure 2-2. A cavity was defined as a region in the convective pass not occupied by heat exchangers
and enclosed by fictitious planes and the walls of the boiler. In the general formulation, the following
simplifying assumptions were adopted: neglecting scattering; gray medium; uniform chemical
composition of the gas; uniform temperature and emissivity of the walls bounding a cavity or heat
exchanger; as well as uniform gas temperature within a cavity. Instead of calculating the direct
exchange areas, a simplified formulation was derived through assuming that the tubes were infinitely
long, ignoring the wall bounding a heat exchanger and using the average mean beam lengths to perform

the radiative calculations.

Coelho [28] applied the mathematical model on the convective pass of an oil-fired 250 MWe power
station. The computational domain consisted of a chain of heat exchangers, namely the final
superheater, primary superheater, reheater and economiser. The predicted gas temperatures at a few
access ports and the exit water/steam temperatures of each heat exchanger were presented. The steam

temperatures were compared to experimental data and good agreement was obtained. The model was
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further used to perform a sensitivity study to investigate the influence of some input variables which

included the inlet conditions, the angle of the plates of the control dampers and the fouling resistances.
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Figure 2-2: A schematic of a heat exchanger with the fictitious planes for the computation of radiation heat

transfer [25].

Xu et al. [29] developed a single tube model for evaluating the wall temperature profiles of superheater
and reheater tubes in power plants. The model uses the FVM to discretise the 2-D heat conduction
equation. They ignored circumferential heat conduction in the tube wall. In addition, the change of
diameter and material of one tube was ignored. Empirical correlations were used to account for heat
transfer on both the flue gas and steam side. Radiation heat transfer on the flue gas side was accounted
for through an effective heat transfer coefficient. Momentum and mass conservation equations were
not solved on either of the two streams. They investigated the effect of inner fouling, steam temperature

distribution and gas temperature distribution on the tube wall temperature.

2.3. Complete heat exchanger modelling

Developing a complete model of a heat exchanger in the convective pass of a boiler enables the
simulation of maldistribution of flow and temperature on the outlet headers of these heat exchangers.
With certain models the so called “M-type” outlet steam temperature profile can be modelled. Such

complete heat exchanger models have to capture the 3-D phenomena present in the heat exchanger.

Prieto et al. [30] developed a steady state 3-D numerical model for simulating flow and heat transfer

in a reheater of a utility boiler. The model was developed in the commercial CFD software code
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PHOENICS V3.1. The porosity concept was employed to model the tube bundles. In addition,
turbulence was accounted for. On the gas side the model solved the conservation equations of mass,
momentum and energy. However, on the tube side only the energy equation was solved. Thus, steam
velocity was derived from the specified mass flow rates and the geometry of the serpentines. They
employed empirical correlations to incorporate convection and radiation heat transfer. The outside heat
transfer coefficient accounts for both convection and radiation heat transfer. The model was developed
such that it can use measured values of velocity, temperatures and gas composition in the reheater as
inlet/outlet boundary conditions. The focus of the model was on the characterisation of the thermal

behaviour of tube serpentines (tubesheets).

In an accompanying paper, Prieto et al. [31] presented the application of the model to investigate the
maldistribution of flow and temperature across the width of the reheater. At first the model was run
with an assumption of uniform steam mass flow rate in all the tubes. When the temperature results at
the outlet of each serpentine were compared to experimental data the distribution was different as
shown in Figure 2-3. This clearly showed that the assumption of a uniform steam mass flow
distribution was not valid. In order to obtain a temperature distribution from the model that compares
well with the experimental results, as shown in Figure 2-4, the model had to be re-run with different
steam mass flow rates in each serpentine. This is one of the limitations of a model based on a
commercial CFD code because in order for it to solve, the velocity or mass flow has to be specified.
Thus, the mass flow distribution must be known beforehand to be able to produce a realistic

temperature distribution.
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Figure 2-3: A comparison between calculated and experimental temperature results from each serpentine at

the outlet of the reheater for a uniform steam mass flow rate assumption on the model [31].
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Figure 2-4: A comparison between calculated and experimental temperature results from each serpentine at

the outlet of the reheater for a non-uniform steam mass flow rate assumption on the model [31].

Gonzalez et al. [32] did experimental work on the same reheater to determine the causes of the
temperature maldistribution. They investigated the effect of load changes, orientation of burners and
the activation of burners. Their results demonstrated that the outlet temperature for reheater tubes
decreases with decrease in load as shown in Figure 2-5. However, the temperature maldistribution
across the width of the reheater still persisted even under low loads. Likewise, variations in the burner

configuration did not change the temperature maldistribution at the outlet of the reheater.
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Figure 2-5: Variation of temperature profile with load [32].

Gomez et al. [33] developed a mathematical model on PHOENICS V2.2 that can be used to simulate
the fluid flow and heat transfer in the convective pass of a power station boiler. The model allowed the

integration of the simulation of the shell side, tube side and the metal thermal fields. On the gas side
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the model solved the continuity, momentum and enthalpy conservation equations. The tubes were
represented as sub-grid features based on the distributed resistance concept developed by Patankar and
Spalding [34]. On the tube side, only the energy conservation equation was solved. Steam mass flow
was calculated locally using geometrical calculations. The model was verified for two distinct aspects.
Firstly, it was checked whether the geometrical features of the model such as connectivity options and
the conservation of the overall tube-side mass flow were working correctly. Secondly, the model was
verified against analytical models for geometrically simple configurations (i.e. single pass cross flow
heat exchanger configuration). These simple configurations were analysed using the LMTD or the
effectiveness-NTU method. The comparison of the predicted overall heat transfer rates was

satisfactory.

The model was then applied on the convective chamber of a 350MWe power station boiler, beginning
at the furnace exit where uniform temperatures and velocity profiles were assumed. In this model, the
tube side heat transfer resistances were neglected. For the gas side, the semi-empirical approach
proposed by Cortes et al. [35] was used. This semi-empirical approach is a formulation of lumped
models coupled with energy balances in a section by section analysis. For the model results, on the gas
side 2-D flow fields and temperature contours for two different load cases were shown. For the
corresponding load cases, the tube side temperatures were also shown. Despite plant measurements
being available for the safe operation and control of the utility boiler, it was discovered that such
information was neither complete nor reliable enough to validate a thermofluid process model. Thus,
a plan to improve and extend the instruments and measurement procedure was designed and
implemented as detailed by Diez et al. [36]. Heat transferred and outlet steam temperatures at different

load cases for each heat exchanger in the convective pass compared well to the gathered plant data.

The rationale of the measurement procedure of Diez et al. [36] was to improve the quality and
reliability of experimental data obtained from large utility boilers for validating thermofluid models.
This is done because there are limitations associated with laboratory scale methods, since the physical
processes involved in a coal-fired boiler are not necessarily scalable. Building pilot demonstration
plants is also not an option due to very high investment and maintenance costs. Thus, the only way to
get experimental validation data is to obtain measurements from the full-scale utility boiler itself. The
limitation of this approach is that data acquired under industrial operation conditions in large boilers
lack the quality and completeness provided by laboratory data. Thus, a methodology to improve that
was devised. The methodology involves the examination of the reliability of the existing process data,

and the design of new instruments and measurement procedures to gather additional information which
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might supplement or improve the quality of the available data. For some aspects of data acquisition
which may prove difficult to measure such as the gas flow, simple Mass and Energy Balances (MEB)

were recommended.

Xu et al. [37] developed a thermal load deviation model to simulate the temperature maldistribution at
the outlet of superheaters and reheaters across the width of a utility boiler. The model was based on
the power plant in situ thermodynamics parameters, thermal deviation theory and flow rate deviation
theory. The thermal deviation model is obtained by an iterative procedure based on three parts; the
structural deviation model, the flow rate deviation model and the thermal deviation model. The
structural deviation describes the geometry of the superheater or reheater. The flow rate deviation
model simulates the flow rates and the heat absorbed by each tube in the heat exchanger. The thermal
deviation model accounts for heat absorbed on the outer surfaces of the tubes. Thus, this model mainly
focuses on energy balance and flow analysis. It does not solve the momentum and continuity equations

on both streams. The calculated results of this model compared well with experimental data.

Tilton and Ratnam [38] developed a 1-D modelling methodology based on the network paradigm for
simulating and predicting the performance of the tube-side of the heat exchangers in the convective
pass. In this methodology the tube banks are subdivided into a number of modules. For each module,
a set of equations consisting of mass, pressure and energy balances were derived, which were then

solved using a sparse matrix method. Their model was not validated against real plant data.

2.4. Transients

Mathematical models that can capture the dynamic behaviour of heat exchangers can range from
lumped analytical models to custom developed numerical models as well as software based numerical
models. In broad terms these models are sometimes referred to as lumped-parameter and distributed-
parameter models. The details captured by a dynamic model depends on the purpose of the model as
highlighted in the earlier sections of this chapter. Most of the dynamic models that were developed in

the early 1950s were focused on the steam outlet temperatures for process control purposes.

Paynter and Takahashi [39] developed a modelling method based on statistical principles to
approximate exact solutions for evaluating the dynamic response of counter-flow and parallel flow
heat exchangers. Gvozdenac [40, 41] developed transient response analytical models for gas-to-gas
parallel flow, counter flow and cross flow heat exchangers. The model solution was based on the
Laplace transform method and the modified Bessel functions. Temperature distribution results for both

the fluids and the wall were shown. Yin and Jensen [42] also developed an analytical model to simulate
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the transient response of a heat exchanger in which one fluid is single phase and the other has a constant
temperature. In their model, the dynamic behaviour of the heat exchanger is approximated by an
integral method assuming that the single-phase fluid and wall temperature distribution can be

expressed by a combination of known initial and final temperature distributions and a time function.

Enns [43] developed and compared dynamic models of a superheater of a boiler in a steam power
plant. Some of the developed models were based on the lumped-parameter approach and others based
on the distributed-parameter approach. For all the models, steam outlet temperature was of interest.
Also, they did not solve instantaneous flow output, pressure drop and tube temperature. In addition,

all the models assumed that heat addition to the tubes was independent of the tube temperature.

Kang et al. [44] developed a nonlinear lumped-parameter dynamic model of a power plant boiler
superheater. Their model had high precision and low computational complexity which implied that it
can simulate operating conditions when large disturbances such as large load changes occur as well as
can satisfy the requirements for control system design. The model consisted of three parts which are
the pressure-flowrate channel model, the enthalpy-temperature channel model and the steam
thermodynamics parameter model. The model was applied on a final superheater of a 600MW boiler
including its attemperation stage. The superheater was simplified to a one tube representation.
Reasonable dynamic results of steam pressure and temperature were observed upon imposed

disturbances.

Zima [45] developed a numerical solution of transient flow in parallel and counter flow superheaters.
The numerical solution entailed iteratively solving the heat transfer equations from an energy balance
on the control volumes of the superheater based on the implicit finite difference method. These
equations were then solved using the method of lines coupled with Gear’s method. Time and space
temperature distribution for the flue gas, tube metal and steam were determined. This numerical
solution was compared to a mathematical model which solved the 1-D transient forms of the
conservation equations of mass, momentum and energy on the steam side. For this mathematical model
the flue gas and steam temperatures were calculated by solving the energy balances over control
volumes as with the first model. Again, in this model, the iterative implicit finite difference method
was employed except that the momentum equation was solved on a staggered grid. The comparison

between the results of these two models was satisfactory.

Zima [46] then applied the mathematical model that solves the 1-D conservation equations of mass,

momentum and energy on the steam side on superheaters of a coal fired boiler. Each superheater was
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modelled as a 1-D counter-flow heat exchanger. This simplification was achieved through assuming
that the conditions in each tube element along the length of the header do not vary leading to a 2-D
superheater. This 2-D superheater was then approximated using the 1-D model. The results of this
model were compared to experimental data obtained from the boiler during acceptance tests, start-up

and shutdown. The comparison was satisfactory and hence validated the model.

Zima [47, 48] applied the mathematical model that only solves the energy conservation equation and
neglects the mass and momentum conservation equations on the platen superheater and economizer of
a coal-fired boiler. For the platen superheater outlet steam temperature results compared well with
plant data during a shutdown. Also, the economizer feed water temperature results compared well with
plant data from acceptance tests. This model is important in the aspect of digital temperature control
of the outlet steam temperature because it is quick to solve. In addition, Zima [48] successfully
compared the model with analytical solutions for a tube with the temperature step function of the fluid

at the tube inlet and for a case with the heat flux a step function on the outer surface.

Although the models of Zima are applicable for digital outlet steam temperature control, the
simplifications of heat exchangers with complex geometry to a counter flow or parallel flow limits the

capability of the model to assess damage caused by the transients on a specific tube or region.

Gwebu [49] and Gwebu et al. [50] investigated the possibility of modelling the dynamic behaviour of
a boiler heat exchanger using an integrated thermo-fluid process modelling software Flownex. The
emphasis of the work was placed on modelling the dynamic behaviour of a tube since by mass a boiler
heat exchanger consists of more tubes than headers. Thus, most of the dynamic behaviour comes from
the tubes. The focus was on radial heat transfer on the thickness of the tube, thus solving the 1-D heat
conduction equation. In the investigation analytical solutions based on the Bessel function were used
to verify a custom developed numerical solution based on the FVM. The verified numerical solution
was then used to verify the software Flownex. Flownex allows the use of time depended boundary
conditions on either side of the tube thickness. The modelling strategy systematically showed the
limitations of each method, beginning with the analytical solution and then the custom developed
numerical solution. However, limited transient cases were studied using Flownex to fully demonstrate
its ability to capture transients of fluids flowing on each side of the tube as well as the propagation of

those transients through the tube thickness.
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2.5. Flue gas side flow modelling

In a typical coal-fired boiler’s convective pass heat exchanger, the flue gas flow is complex due to the
presence of numerous tubes in the path of the flow. These densely populated tubes increase the
resistance to the flow. The arrangement of these tubes in the heat exchanger influences the extent of

mixing of the flue gas within the heat exchanger.

In this section, a review of literature on modelling the flue gas pass in boiler heat exchanger or the
shell side of a shell and tube heat exchanger is presented. The literature is systematically presented in
accordance with the timeline of the development of the tools used. It begins by analysing the
application of conventional Computational Fluid Dynamics (CFD); the application of the full Navier-
Stokes equations and its limitations. The limitations of the conventional CFD are then addressed by
adapting the full Navier-Stokes equations through the CFD porous media approach. The porous media
approach is founded upon the distributed resistance concept. For further advantages, this distributed
resistance concept is adapted into a 1-D network approach, which is used to model two-dimensional

and three-dimensional phenomena.

2.5.1. Conventional Computational Fluid Dynamics (CFD)

According to Runchal [51], the origins of Computational Fluid Dynamics can be traced as far back as
the 1928’s or even earlier. Back then, many researchers developed concepts to what eventually became
CFD. Despite being around for a couple of decades before, the wide availability and accessibility of
digital computers as a tool in the early 1960’s assisted in the development of CFD. In addition, the
work done by a team led by Professor Brian Spalding at Imperial College in London on the Navier-
Stokes equations and computer codes also contributed immensely in the development of CFD. In 1969,
they published their work in a book [52] which promoted the use of CFD as an engineering tool. This
led to this team being credited for creating CFD as an engineering tool, since they applied CFD in
solving engineering problems [51, 53]. Furthermore, their work significantly contributed to the

development of most of the commercially CFD software packages available today.

Spalding’s team reinvented the upwind concept which fostered the thinking in terms of fluxes rather
than the state of variables at a given grid point [54]. This followed the work of Barakat and Clark [55]
and Gentry et al. [56]. However, the upwind concept was first put forward by Courant et al. [57]. This

concept gave rise to the tank-and-tube analogy where a node (control volume) is viewed as a tank
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exchanging fluxes with adjacent tanks via tubes (grid). This led to the development of the Finite

Volume concept which is widely applied today.

Due to the tremendous improvement in computational power over the years, researchers continued to
improve CFD and its applications. It is applied across a wide variety of industrial sectors including
aerospace, defence, automotive, medical, chemical, manufacturing and processing, environmental, fire

and safety, marine, nuclear power and fossil-fuel power industries [53].

Since computers of the 1960’s were not computationally powerful, novelty in generating concepts to
simplify modelling of certain industrial equipment was key. This equipment included the shell and
tube heat exchanger used mostly in the nuclear and fossil fuel power industry. Since the shell and tube
heat exchanger has a complex geometry including a complex arrangement of tubes and baffles, it
would have been computationally expensive to model it through conventional CFD over the whole 3-
D computational domain. Conventional CFD refers to solving the full Navier - Stokes equations. Thus,
other concepts were developed as alternatives to model such equipment. These concepts included the

Subchannel Analysis and the Distributed Resistance concept (porous media) [58, 59, 60].

The subchannel analysis is suitable to model flows around obstacles that form a regular pattern such
as a tube bank in a heat exchanger or a rod bundle in nuclear fuel elements. In this analysis, the space
in between a cluster of three or four tubes is referred to as a subchannel. In a computational domain,
one grid is equivalent to one subchannel [59]. The foundation of the method is that it explicitly assumes
that one of the velocity components (e.g. axial direction) in a certain direction is dominant compared
to the other velocity components (e.g. cross flow) [58, 60]. Thus, only the dominant direction is
rigorously solved, hence the only computationally expensive direction. Originally, the subchannel
analysis was intended for design calculations. However, it has since been used extensively in the
nuclear industry, especially for reactor safety analysis such as modelling flow blockage in the fuel

assembly, as seen in references [61, 62, 63].

The subchannel method can also be used to characterise pressure drops for certain tube arrangements,
thus getting resistance correlations for the distributed resistance method in cases where empirical
correlations do not exist. The distributed resistance method entails less details than the subchannel
analysis since it combines the porosity concept and the distributed resistance concept, thus allowing

for a courser computational grid. It is addressed in detail in the next subsection.
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2.5.2. CFD porous media approach: The porosity and distributed resistance concept

The distributed resistance method, just like the subchannel analysis, is an alternative method to
modelling engineering equipment which often have solid obstacles such as rods, tubes and baffles
through which the fluid flows. This equipment include heat exchangers, condensers, cooling towers,

steam generators, nuclear fuel rod-bundle and chemical reactors.

One of the advantages of the distributed resistance method over the subchannel analysis is its economy
on computer solving time and storage. This is because the distributed resistance method is based on
the “continuum” approach, implying that its computational domain does not depend on the geometry
of the obstacles in the equipment. The effect of the obstacle geometry is introduced through the
porosity to account for variation in volume and the distributed resistance factor to account for
resistance to fluid flow as well as the heat transfer to or from the fluid [64]. In essence, the flow through
the equipment is treated as fluid flowing through a porous media, thus this method is also called the
CFD porous media approach. This method was developed and published in 1974 by Patankar and

Spalding [34]. The development and further use of the distributed resistance method is traced below.

In 1974, Patankar and Spalding [34] developed the distributed resistance concept to solve the
differential equations that govern the flow of a continuous fluid. They initiated the use of this concept
to model shell and tube heat exchangers. This approach regards the space within the shell as uniformly
filled with fluid, through which a resistance to fluid motion is distributed on a fine scale. This resistance
is due to the presence of tubes and baffles in the shell. Mathematically, this resistance is introduced in
the governing equations by introducing a representative factor. They showed the governing equations
in Cartesian coordinate system. An advantage of this approach is that it allows the use of a coarse grid
which saves computer solving time and storage. In the paper, they presented simulation results for both
steady state and transient regimes. However, no attempts were made to compare their results to

experimental data.

AbuRomia et al. [65] applied the distributed resistance method in their detailed model to obtain the
flow distribution on the shell side of the Clinch River Breeder Reactor Plant Intermediate Heat
Exchangers. In the study, they used a lumped and a detailed model to analyse the axial/cross flow on
the shell side of the heat exchangers. The lumped model utilized an overall pressure drop and flow
distribution approach to determine the magnitude of the axial and cross flow components as a function
of the baffle configurations. The detailed model used a fundamental approach in numerically solving

the governing equations of fluid mechanics on a turbulent flowing fluid over a discretised
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computational domain. The distributed resistance concept was applied to the momentum equation to
account for the presence of the tube bundles and baffles. This detailed model resulted in a complete
map of the fluid and pressure distribution within the tube bundle. It was concluded that the radial
distribution of the cross flow can only be accurately obtained using the detailed model. This radial

distribution is the mechanism responsible for fluid mixing in the tube bundle.

Patankar and Spalding [64] applied the distributed resistance method to model the three-dimensional
flow and heat transfer in a steam generator. They employed the method using equations based on the
cylindrical coordinate system. The porosity was explicitly included in the equations to account for the
presence of the tubes in the nominal volume. They successfully calculated and plotted the three-
dimensional velocity and enthalpy fields. Their results were not validated against experimental data,
thus they recommended it for future research. The recommended future research also included work

on empirical correlations for the distributed resistance factor.

Butterworth [66] used equations for analysing flow for anisotropic porous media to show how pressure
drop data for one-dimensional flow in tube bundles may be generalised in order to provide a framework
for analysing real flow problems which are often multidimensional in nature. The equations use a flow
resistance tensor which accounts for flow resistance or permeability. For turbulence encountered in
most tube-bundles the tensor components depend on the magnitude and direction of the superficial
velocity vector. However, a simplifying assumption was made such that the tensors depend only on
the magnitude of the velocity and not the direction. This resulted in isotropic flow properties for planes
perpendicular to tubes arranged in square and equilateral triangle arrays. Thus, implying that the
pressure drop for flow in one direction can be predicted using data from the flow in another direction.

This was shown to be reasonable by comparing to experimental data.

Singhal and Spalding [67] presented the URSULA?2 computer program for modelling multi-phase flow
and heat transfer in steam generators. The mathematical formulation of the program was based on the
Inter-Phase Slip Analyser algorithm and the distributed resistance concept, described in reference [64].
Spalding [68] presented the mathematical formulation of the fundamental governing equations of fluid
flow used in the URSULA2 program. These equations incorporated the porosity and distributed
resistance and were written in cylindrical-polar coordinates. The diffusion or turbulent mixing terms
were neglected in the momentum conservation equation. The porosity or void fraction was calculated
from the fluid density and the resident enthalpy. Empirical correlations were used to account for the

resistances including the solid to fluid resistances.
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Marchand et al. [69] used the URSULA2 computer program to perform two-phase, transient, three-
dimensional flow and heat transfer calculations for the thermal-hydraulic performance of a steam
generator. The transient was due to a power ramp from 100 to 60 percent. They were satisfied with the
trends of the results; however, they did not quantitatively verify these results with experimental data.
Singhal et al. [70] also applied the URSULA?2 computer program to predict the thermo-hydraulic
performance of a PWR steam generator using three different models; the homogeneous, the two-fluid
and the algebraic-slip models. The predictions of the three different models showed significant
differences especially on the circulation ratio and the flow details on the hot side of the steam generator.
However, the URSULA?2 program was not validated with experimental data thus its results were

interpreted with caution.

Sha [60] highlighted the distributed resistance method in conjunction with porosity. This was presented
in a summary of methods that can be used in rod-bundle thermal-hydraulic analysis together with the
subchannel analyses and the benchmark rod-bundle thermal analysis using a body fitted coordinate
system. The differences between the methods and the assumptions employed and shortcomings of each
methods were clearly shown. It was concluded that the distributed resistance method eliminates the
limitations of the subchannel analysis in modelling large flow disturbances. Furthermore, the
distributed resistance method is applicable to a wide variety of geometries compared to the subchannel

analysis. Thus, it presents a most cost-effective way of developing a computer code.

Sha et al. [71] developed a multidimensional thermal hydraulic computer code to analyse heat
exchangers. The analysed heat exchangers were the shell and tube heat exchangers for liquid metal
services e.g. Liquid Metal Fast Breeder Reactor intermediate heat exchangers and steam generators.
The concept of distributed resistance, volumetric porosity and surface permeability were used on the
shell side to modify the continuum Navier-Stokes equation to cater for the effect introduced by the
presence of obstacles such as tubes and baffles. In essence, this modelled the flow in anisotropic porous
media. The code was tested using experimental data. The comparison between experimental

predictions and the model predictions were satisfactory.

Srikantiah and Singhal [72] presented a computer program called ATHOS which was developed for
design and performance analysis of steam generators in the nuclear power industry. This program can
model the geometry and internal structure including the tube bundle, tube support plates and baffles of
the steam generator in considerable details. Due to symmetry, only one half of the steam generator was

modelled. The porosity concept and distributed resistance concept were applied on the secondary side.
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In addition, on the secondary side diffusion and turbulence were assumed to be negligible as well as
the dissipation and spatial component of the pressure work in the energy equation. The final flow field

for steady state simulations were plotted successfully.

Hu [73] used the distributed resistance approach to model tube bundle flow during a steam generator
wet layup. The model used the concept of anisotropic porous media flow within the Darcy regime to
predict the flow pattern of the tube bundle flow. The distributed resistance (fluid continuum) approach
for porous media was established by introducing porosity which is the ratio of the fluid volume to the
total volume. In general, a continuum approach is valid when a representative elementary volume is
large enough to give a continuous statistical average of the fluid property and is small enough so that

it may be considered as a material point in space [74].

In 1984, Patankar [59] established that the general heat transfer community did not sufficiently
understand the computational activities associated with the distributed resistance concept when applied
on the shell side flow in heat exchangers. Thus, he wrote a paper providing a review of the basic
concepts associated with this type of analyses together with some illustrative examples. He explained
that the shell side space is regarded as continuum which is under the action of distributed resistance
due to the presence of obstacles such as tubes. In essence, the shell side flow is treated as a flow through
a porous material. The presence of obstacles in the flow is accounted for via the porosity and the

resistance formulas. The resistance formulas can be determined using empirical correlations.

Rhodes and Carlucci [75] presented a comparison between numerical predictions using the porous
media concept, and measurements of the two-dimensional isothermal shell-side velocity distributions
in a model heat exchanger. The computations and measurements were done with and without tubes
present in the model to assess the efficacy of the porous media theory. The porosity, which accounts
for flow volume reduction due to the tube bundle, was assumed to be locally isotropic. Their study
demonstrated feasibility and validity of the porous media approach in modelling the fluid flow over
obstacles. It also highlighted a discrepancy between the measured and predicted results from modelling
the tube-to-baffle leakage flow. However, it was concluded that the discrepancy might have been due

to the method used to estimate the leak rate from the velocity measurements.

Theodossiou et al. [76] performed a numerical study on the shell side of an experimental heat
exchanger with an isothermal, steady flow distribution and incompressible fluid. The numerical study
was two dimensional. The computations were done for both cases with or without tubes in the model,

for Reynolds numbers less than 10 000. The distributed resistance concept was used to account for the

University of Cape Town 29



Chapter 2. Literature review

presence of tubes and baffles in the model. The resulting equations from the distributed resistance
concept were solved using a semi-implicit predictor type scheme. The computations were run using
different meshes in order to validate the code. For further validation of the distributed resistance
concept, the numerical results were compared to experimental data. This comparison produced a good

qualitative agreement.

Karayannis and Markatos [77] applied the distributed resistance method encapsulated in the program
HERCULES to rate shell and tube heat exchangers. The HERCULES program ran under the
PHOENICS CFD software package. They recommended the use of the CFD based approach to model
the shell side flow field over the traditional methods which had simplifications. These simplifications
included the assumptions made for mechanical design (leakages & bypass flows), thermal process

(viscosity & density changes), fouling and corrosion amongst others.

Prithiviraj and Andrews [78] developed a three-dimensional, conservative, fully implicit, collocated
control volume-based calculation technique for simulation of flow, heat transfer and turbulence on the
shell side of shell and tube heat exchangers. This computer code, called HEATX was based on the
distributed resistance method and the k — £ turbulent model. Wall functions of Launder and Spalding
[79] were used to compute momentum and heat transfer coefficients close to plane surface walls.
HEATX also employed the volume porosities and non-homogeneous surface permeabilities to account
for the presence of obstructions such as tubes and baffles in the computational domain. To account for
distributed resistance across tubes in cross flow, pressure drop empirical correlations from Zukauskas
[26] were used. In addition, to account for tubes in an axial flow, pressure drop correlations of Rehme
[80] were used. HEATX was validated against standard benchmarks such as the driven cavity,

backward facing step and sudden pipe expansion.

Prithiviraj and Andrews [81] presented the foundation and fluid mechanics behind the HEATX
computer code. Equations governing fluid flow, heat transfer and turbulence were shown including the
empirical correlations used to account for distributed resistance due to the presence of tubes and
baffles. The code was employed to model an E shell type heat exchanger. The results agreed with
experimental data. Prithiviraj and Andrews [82] presented the heat transfer part of the HEATX
computer code. The shell side is coupled with the tube side of the heat exchanger. HEATX was
employed to simulate a shell and tube heat exchanger. Both pressure drop and temperature difference

results agreed well with experimental data.
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Hueng et al. [83] used the distributed resistance method incorporated in the PHOENICS CFD software
package through subroutines to model a shell and tube heat exchanger. Their main objectives were to
investigate the effect of different turbulence models on the velocity field and the heat transfer
coefficient as well as perform design optimisation calculations on the heat exchanger. Four different
turbulence models were used; the constant turbulent viscosity model, the ¥ — £ turbulence model, the
Chen — Kim k - ¢ turbulence model and the k - ¢ and constant turbulent viscosity model. They
found that the effect of different turbulence models on the velocity distribution and heat transfer was
insignificant. Thus, implying that a simple turbulence model could be used without compromising the
accuracy of results. In the design optimisation of baffles in the heat exchanger, they reported that the
optimum size of the baffle is about 70% of the height of the heat exchanger. This optimum size had a

higher heat transfer rate and relatively lower pressure drop.

Coelho [25] presented the theory of a two-dimensional mathematical model for simulating flow and
heat transfer in the convective pass of a utility boiler in steady state. The flow field simulation was
based on the mass and momentum equations in conjunction with the distributed resistance concept and
porosity to account for tube bundles and control dampers. The k - £ eddy viscosity/diffusivity model
was employed to cater for turbulence. The pressure drop of the cross flow through the tube bundles
was estimated using the correlation of Zukauskas [26]. Energy balances were performed along the
tubes to account for convective heat transfer and fouling resistance on both sides of the tube, as well
as radiation on the gas side. The flow and heat transfer are iteratively coupled by the solution algorithm.
Coelho [28] applied the mathematical model on the convection chamber of an oil-fired 250 MWe
power station. The computational domain consisted of a chain of heat exchangers, namely; the final
superheater, primary superheater, reheater and economiser. The results of predicted 2D velocity field

were presented, however these results were not compared to experimental data.

Mirzabeygi and Zhang [84] used the distributed resistance approach to account for the flow resistance
due to the presence of tubes in the shell side of a three-dimensional model of a condenser. The
numerical model for this condenser was based on Eulerian — Eulerian two-phase model for solving the

conservation equations.

It is evident that the distributed resistance method or CFD porous media approach to model the shell
side of a shell and tube heat exchanger has been used extensively in the past four decades. Nowadays,
this method is available in most commercial CFD software packages, either directly or through

subroutines. Important components of this method that need careful considerations are; the choice of
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empirical correlations used to account for the presence of tubes in the momentum and energy equation,

porosity, and whether to account for turbulence or not.

2.5.3. Integrated Systems CFD analysis

In thermal-fluid design of systems consisting of many components connected together (e.g. nuclear
reactors, coal fired power plants, etc.) engineers are continuously faced with two major challenges.
The first challenge is on the ability to predict the performance of each individual component in the
system. The second challenge is to predict the performance of the integrated system consisting of all
the sub-systems which interact with one another [85, 86]. The modelling and analyses techniques used
range from simplified one-dimensional models which do not capture all the physical phenomena [87]
to the detailed large scale computational fluid dynamics models, which captures most of the significant

physical phenomena but cannot simulate the entire plant as an integrated model [88].

An integrated Systems Computational Fluid Dynamics (SCFD) Approach was developed such that it
can tackle these challenges in thermal-fluid design of complex systems [86]. The underlying
philosophy of the SCFD methodology is that for the modelling and analyses of large thermal-fluid
systems it must be possible to link models of various levels of abstraction and various degrees of
complexity together to simulate the complete integrated system. The level of abstraction and the degree
of complexity of the models are determined by the nature and details of the simulation as well as the
details and character of the information required. This approach provides a progressive analysis of the
problem, since the complexity of the simulation can be increased gradually and selectively and
eventually also the level of abstraction, until the required complexity and levels of abstraction are
reached [86]. In the context of superheater modelling, the SCFD allows the gas side to be modelled
using CFD and then be integrated to the tube side which is modelled using one-dimensional techniques.
Over the past two decades, this philosophy has gradually evolved from a mainly 1-D network approach

[89], to a more advanced strategy [90] and finally a comprehensive methodology [91].

du Toit et al. [91] presented the SCFD in which a network code, Flownex, served as the main
framework to interconnect models of the various components in a system together and to control the
solution as well. They applied the SCFD to model the thermo-fluid phenomena of the Pebble Bed
Modular Reactor (PBMR) reactor core and core structures. The fundamental governing equations were
re-formulated into forms that were suitable for use in the SCFD code. The model was based on a
simplification philosophy which was to strike a balance between accuracy and simplicity as well as

refrain from simply developing a detailed CFD code that does not allow quick integrated plant
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simulations. The numerical results were compared to experimental data and a good comparison was
observed. Rousseau et al. [92] validated the transient thermal-fluid SCFD model of the PBMR reactor.
The Flownex results for four transient cases were compared with experimental results as well as results
from the Thermix/DIREKT code. For the Flownex model quicker computer simulation time were

observed thus illustrating that a fine balance between accuracy and simplicity was achieved.

In 2006, du Toit et al. [93] highlighted the SCFD approach. They qualified it with an illustration of its
use on two examples; the simulation of the power cycle of the PBMR power plant and the analysis of
a coal-fired boiler. Du Toit and Kruger [94, 95] applied the integrated SCFD approach in simulating a
coal-fired boiler furnace and evaporators. Flownex was used to model the quality of the water-steam
mixture in the evaporators as well as the main framework for the simulation. This Flownex model was
linked to a Flo++ 3-D CFD model of the gas flow and heat generation in the furnace. The CFD model
showed gas flow patterns and heat distribution in the furnace while the Flownex model showed the
temperatures and steam quality inside the evaporators. The integrated SCFD proved to have an
advantage over uncoupled models which often rely on assumptions in order to couple certain

phenomena, since in this method the coupled phenomena are iteratively solved.

du Toit and Rousseau [86] presented an overview of the modelling of the flow and heat transfer in a
packed bed High Temperature Gas-Cooled Reactor using the SCFD approach. The developments in
the design and layout of the PBMR gave rise to need for the simulation of a wide range of physical
phenomena. Thus, a comprehensive 2-D axisymmetric pebble bed reactor was developed through
reducing and recasting the conservation equations into a form usable with a network code. This re-
formulation of the conservation equations resulted in a collection of one-dimensional models that can
be linked together to build a comprehensive multidimensional model of the reactor. The initial
validation of this model entailed comparing the predicted results to steady state and transient
experimental results. A good comparison between simulated and experimental results was obtained
which showed that for the cases considered the SCFD approach accounted for all the important
physical phenomena. A fine balance between speed, accuracy and simplicity was achieved from this

SCFD model.

2.5.4. One dimensional network approach applied to three dimensional phenomena

In the previous subsection, the integrated system CFD approach was presented as a technique that has

an advantage of combining the detailed nature of the CFD results with the simulation of a complete
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plant. However as shown with some applications of the SCFD, it is also possible to take advantage of

the simulation speed presented by the use of the 1-D approaches to model 3-D phenomena.

In 1994, Tilton and Ratnam [38] developed a methodology for integrating the results of 2-D or 3-D
gas flow models with that of a 1-D gas and tube flow model. The 1-D model used the network approach
in which the tube banks were divided into modules. The gas flow and tube flow modules were
explicitly interconnected. The interconnection between these modules can be split or mixed. They
stated that a set of equations was derived for each individual heat exchanger module, consisting of the
conservation equation; mass, momentum and energy equation. However, they did not show these
equations only referring readers to semi-empirical methods in literature. They further stated that the
set of equations were solved using the sparse-matrix method within a network solution code. As part
of the method, the gas flow results of the 1-D model were compared to results from a CFD gas flow
model based on the distributed resistance method. The CFD results were used to determine the pattern
of the gas flow distribution in the 1-D model. The method was applied on the convective pass of a

boiler. However, the model was not validated against experimental data.

2.6. Summary of literature review
From the literature review the following conclusions are drawn:

¢ In many of the simplified models reviewed the solution of the momentum conservation
equations are not included, which implies that the flow maldistribution cannot be obtained as
part of the solution.

¢ For the CFD models, where the momentum equation is solved on the flue gas side, the inlet
mass flow or velocity distribution must be specified. Thus, the flow maldistribution on both
the flue gas and the steam side cannot be obtained automatically.

¢ There are few available dynamic models that capture the detail geometry of the heat exchanger.
Usually, lumped parameter models are used in dynamic simulations. Lumped parameter
models cannot capture the dynamic behaviour of different tubes or regions within a heat
exchanger.

¢ Several 1-D models have been applied for modelling heat exchangers in coal-fired boilers to
reasonable levels of detail.

¢ Some models based on the 1-D approach modelled the flue gas as if it was flowing in parallel

channels. Thus, they do not account for mixing.
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¢ The need for large boiler heat exchanger model verification and validation is highlighted in
literature. However, it is not clear how to do this in a structured and methodical way.

¢ The distributed resistance method or CFD porous media approach has been used extensively
by researchers and engineers to model heat exchangers in an industrial setting.

¢ Only a few researchers attempted to use the 1-D network approach to model 3-D phenomena.
This approach aims to take advantage of the increased simulation speed presented by the 1-D

network approach.
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3. ONE DIMENSIONAL NETWORK METHODOLOGY

Disclaimer: The conceptual idea of the modelling methodology was introduced in reference [1] which was published as
part of the proceedings of the 7" IASTED International Conference on Modelling, Simulation and Identification, Calgary,
2017. Further development of the methodology was published in the journal Heat Transfer Engineering [2]. The candidate
is indebted to his supervisor together with whom the concept was developed.

The network approach employed here is encapsulated in an integrated process modelling software suite
called Flownex Simulation Environment (Flownex SE). The Flownex SE software platform (shown in

Figure 3-1) was developed by M-Tech Industrial [96] in South Africa.

3.1. Flownex software overview

This section is included for completeness and serves to introduce a reader who is not familiar with the
Flownex software, the platform on which the models have been built. Readers who are already familiar

with the platform may wish to proceed directly to section 3.2.

This software platform is built on a system-level one-dimensional thermo-fluid network methodology
[97, 98]. In its library this software contains several pre-configured components such as pipes, heat
exchangers, flow resistors, compressors, pumps, turbines, etc which can be systematically connected
together in any complex manner to model a complete thermo-fluid system. The thermo-fluid system
can be a complete power plant, heating or cooling system or a subsystem of a thermo-fluid system. In
addition, a range of “primitive” heat transfer components are also available. These components include
1-D planar conduction heat transfer elements, surface convection heat transfer elements as well as
surface and spatial thermal radiation heat transfer elements which can be connected to create any
complex heat transfer network. Flownex is also capable of handling user-defined re-usable compound
components built from systematically combining the pre-configured components or primitive

components in its library [98, 99].

The software also allows flexible coupling with other software tools which gets inputs from Flownex
to solve and send results back to Flownex and vice versa. These software tools include Matlab and
Simulink or any other tool with an Application Programming Interface (API) functionality and have
specific capabilities to embed models or calculations developed in Excel, Mathcad, Engineering
Equation Solver (EES), Relap or C# scripts. In addition, Flownex contains tools such as a system

designer (similar to goal seeking), optimizer and sensitivity analysis [49, 98, 99].
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Figure 3-1: Flownex Simulation Environment process modelling platform (M-Tech Industrial, 2014).

Flownex solves the transient form of the fundamental conservation equations together with built-in
fluid property relations and component characteristics representative of all the types of components.
In a network made of these components, control elements can be added to obtain a complete integrated
dynamic system simulation model of a plant, sub system or component. A range of working fluids are
available. These include the homogeneous model of water-steam according to the International
Association for the Properties of Water and Steam (IAPWS) formulations. The Flownex solver
provides as output the fluid properties such as temperature, pressure or quality at each point in the
system together with the mass flow rates, heat transfer rates and power for all components. Some
properties are a product of the solution scheme and others are obtained through post-processing [98,

99].

The integrated process modelling capability encapsulated within Flownex entails the simultaneous
solution of the transient one-dimensional forms of the conservation equations for mass, energy and
momentum, combined with the applicable closure relations, boundary values and initial values. The
closure relations include models for the component specific characteristics and all the modes of heat
transfer as well as the fluid property relationships. The network methodology can be described in terms
of the node and element schematic shown in Figure 3-2. By convention, in a network methodology

diagram, elements are represented by circles and nodes represented by squares.

In this approach an element is essentially a control volume which may represent any type of physical
component such as a pipe, valve, heat exchanger, boiler, turbine, etc. Each element has one inlet and
one outlet and the properties within the element are assumed to be represented by a single weighted

average value between the inlet and the outlet. An element may also represent a single subdivision or
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increment of a physical component that is discretized into several control volumes. It may therefore

represent a pipe increment or heat exchanger increment [98].

Node

J—O—

Element

Figure 3-2: Network of nodes and elements applied in Flownex [98].

The nodes represent the connection points between elements, which may also be a physical reservoir
or tank. Nodes may therefore have multiple inlets and outlets with the properties within a node assumed
to be uniform and represented by a single averaged value. In the solution of the integrated network all
the fluid volume, and therefore all the fluid mass, is assumed to be contained within the nodes, but the
net change of momentum between all the inlets and outlets of a node is assumed to be negligible. Mass
may be added to or removed from a node via mass sources or sinks. Energy may also be added to or

removed from a node via power and heat transfer terms.

On each element, half of the volume is allocated to the inlet and outlet nodes respectively, implying
that mass and energy can only accumulate in nodes, not in elements. However, elements accommodate
the total (stagnation) pressure changes due to power inputs and outputs or frictional losses occurring
within the specific component. The mass and energy conservation equations are therefore solved for
each of the nodes, while momentum conservation is solved for each of the elements. In addition,
Flownex solves for the mass flow rates due to pressure drops, and determines the resultant operating

points of pumps, fans, compressors and turbines [98].

The spatially integrated transient partial differential forms of the mass and energy conservation
equations without internal sources that are solved for each node are given in simplified format by

ap

5 =é(2m,--2me) (3.1

LI (S i ey [ 2, 20
ot ,OV(zmi(hOi-l-gZi) sz(hoe+gZE)+Q W+V(at h“atjj G2
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Here p is the density and /, the specific stagnation or total enthalpy of the fluid defined as A +1v*.
h is the specific static enthalpy, v the weighted average velocity in the control volume, V the
volume, ¢ the time, 2 the mass flow rate, Z the elevation and p the static pressure. Wherever there
is no subscript it refers to the averaged property value within the control volume, while the subscripts

i and e refer to the properties at the inlets and outlets respectively. Q is the rate of heat transfer to

the control volume, W is the power output from the control volume and g is the gravitational

acceleration.

The transient one-dimensional form of the momentum conservation equation (for incompressible flow)

that is solved for each element is given by

om A 0
a_’? :Z((pol‘ _pOe) +10g(zi _Ze) +Apy, —Apy, +Lv§} (3.3)

Here p, is the total pressure defined as p +1 pv®, A the average cross-sectional flow area, L the
representative length Ap,,, and Ap,, are the total pressure increase or decrease due to work and losses

respectively. For compressible and homogeneous two-phase flows eq. (3.3) will contain additional
terms. For elements such as pipes, ducts, bends and valves the total pressure loss term is often
expressed in the form of a loss coefficient, K multiplied by the dynamic pressure represented as

follows

|
Op,, =K @ pv’ (3.4)
Equations (3.1), (3.2) and (3.3) may be integrated over a discrete time step A: using an Euler

integration scheme to obtain

po=p +((1—a)Sf +aSC)At (3.5)
for the new density,

Iy =1 +((1-a)S: +as,) & (3.6)
for the new total enthalpy and,

=it +((1-a) s, +as, ) 3.7)
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for the new mass flow rate through the components. The superscript o refers to the values at the

previous time. The terms S_(z), S, (¢) and S, (¢) are the time dependent source terms emanating

from the mass, energy and momentum conservation equations, respectively [98]. The weighting factor
a which determines the characteristics of the integration scheme is between 0.0 (fully explicit) and
1.0 (fully implicit) for the source terms at the previous time step and current time step. This therefore
determines the degree of forward or backward Euler integration. A weighting factor of 0.7 is often

applied since it produces a good balance between accuracy and stability.

As illustrated schematically in Figure 3-3, the network methodology essentially reduces to the
calculation of new values for the density and total enthalpy in each node based on the “old” values
(with superscript 0 ) and the mass flow rates into and out of the nodes. The mass flow rates are dictated
by the elements connected to the nodes. With the enthalpy and density known within the node, the

pressure may be derived from the appropriate fluid property relationship p = f (ph) The pressures

in the nodes in turn dictate the new values of the mass flow rate in each of the elements. The primary

dependent variables that are solved for are p,,, i, and p .

7\

Component Mass & Energy Momentum Component
Characteristics Conservation Conservation Characteristics
' p=1(p°.5.(1)) .
Q m=f(m°,S, (1)) Pow
W hy= f(43,S,(1)) Apy;
h=f(h) o =f(p)
\ Fluid
Properties
p=f(p.h)

Figure 3-3: Schematic of Flownex transient network solution methodology [98].

The numerical solution scheme applied in Flownex is an Implicit Pressure Correction Method (IPCM).
It is based on combining the mass and momentum conservation equations to obtain a total pressure

matrix which, after a Newton-Raphson-type linearization, produces a total pressure correction solution
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matrix. A more detailed description of the numerical solution scheme is presented by

Greyvenstein [100].

It is important to note that while the conservation equations will be the same for all the nodes and
elements, the component characteristics ( Q W, Ap,,, andAp,, ) will be totally different, depending on

the component of interest. The level of complexity at which each component is modelled is really
determined by the level of complexity that is encapsulated in the component characteristic
formulations. Interestingly, there can be a very wide variation in the level of complexity, even for the
same type of component operating with the same working fluid and the same boundary values [98,

101].

Besides solving the transient form of the conservation equations, Flownex also has the ability to
directly solve the steady-state form of the equations, i.e. with the 0/ Ot terms set equal to zero. This
implies that the steady-state solution can be obtained directly and quickly without having to run a

transient from a given set of initial conditions until steady-state is eventually reached.

In terms of heat transfer, Flownex is capable of handling conductive, convective and radiative heat
transfer. Flownex solves both the Cartesian form and cylindrical form of the 2-D transient heat

conduction equation. The 2-D Cartesian form of the transient heat conduction equation is given by

pca_Tzi(ka_T]+i kd_T (3.8)
ot Ox\ Ox) 0y Oy
Solving the Cartesian form of the equation implies that the area discretisation is done using standard

averages [97]. This capability allows Flownex to solve for heat conduction through any shape,

provided the shape is sufficiently discretised.

For the cylindrical form, the circumferential heat conduction is neglected, thus it solves radial and
axial conduction heat transfer. The cylindrical form of the 2-D heat conduction heat is given by

pca_Tzli(rka_TJ.Fi(ka_TJ (39)
ot ror\ 9or) 0z\' o0z '

with k the thermal conductivity, ¢ the specific heat capacity and I is the radius of the component.
Solving the cylindrical form of the equation implies that linear area change discretisation is used [97].
The heat conduction equation, Cartesian or Cylindrical, is solved using the FVM on a grid with half

control volumes at the end as illustrated in Figure 3-4.
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The convection heat transfer is accounted for through empirical correlations. This convective heat
transfer is set to be constant throughout an element or an increment if the element is discretised axially
[96]. Radiation heat transfer is accounted for through solving the standard radiation heat transfer

equation given by

Q.4 =AFo(T, ~T) (3.10)
with A the outer fouled tube surface area and F the view factor.

Material 1

3 Nodes

Half
control
volume

Figure 3-4: Representation of the discretisation of the thickness of the component on the heat transfer element

[49].

Figure 3-5 shows the basic Flownex components which are a pipe element, a composite heat transfer
element, a boundary condition and a node, while Figure 3-6 presents the primitive heat elements for

conduction, convection and radiation.

The pipe element shown in Figure 3-5a) can be used to model flow in pipes and ducting with non-
constant, arbitrary cross-sectional area. It takes into account both friction and secondary losses. The
technical capabilities of this component include the ability to handle gas, liquid and two-phase flows,
the ability to model both steady-state and transient flows, the capability of handling the effects of
gravity as well as the capability of allowing the user to discretise the computational domain into a
number of increments. The 1-D convective acceleration term in the momentum equation is retained in

the calculations. For transient flows, the inertia terms are also retained in the calculations [102].
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B SI

a) b) c) d)

Figure 3-5: Flownex components; a) Pipe, b) Composite Heat Transfer Element, c) Boundary Condition and
d) Node.

The Composite Heat Transfer element (CHT) shown in Figure 3-5b) is used to model heat transfer in
a solid structure resulting from an interaction with its surroundings. The CHT can account for
conduction, convection and radiation heat transfer to and from the solid structure. It can account for
different solid materials which are modelled by dividing the structure into different material layers for
the conduction heat transfer. The capabilities of modelling conduction heat transfer on both Cartesian
and cylindrical coordinate systems are described in the preceding subsection. In essence, the CHT
element can account for both thermal resistance and thermal inertia. Its ability to allow for the
interaction of the solid structure with the surroundings implies that it can connect to flow elements or

flow nodes with either single or two-phase flows [102].

The Boundary Condition component shown in Figure 3-5c¢) is used to specify the boundary conditions
of a network. This defines the interface between the network and the surroundings. These may be inlet
conditions, outlet conditions, etc. The boundary condition component can only be connected to a node
component in which the specified conditions are applied. The conditions that can be specified at the
boundary include pressure (static or total), temperature, quality, enthalpy and mass source. The nodes
(illustrated in Figure 3-5d)) are end points of elements. However, nodes can also be used to model

tanks or reservoirs [102].

a) b) c)

Figure 3-6: Primitive heat transfer element; a) conduction, b) convection and c) surface and spatial radiation

heat transfer elements.

To allow for flexibility in modelling certain heat transfer problems, a combination of two or more
primitive heat transfer elements is preferred over the CHT element. The purely Conduction primitive
heat transfer element shown in Figure 3-6a) is used only for linear, 1-D heat transfer in mostly solid

structures. Figure 3-6b) shows the Convection primitive heat transfer element which serves as a heat
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exchange link between a fluid and a solid structure. The Convection primitive heat transfer element
can be connected to either a non-discretised flow element or a flow node on one side and on the other
side it is always connected to a solid node. It cannot be connected between two flow nodes or two flow
elements or a combination of a flow element and a flow node. On the Convection heat transfer element,
the user specifies the heat transfer area as well as the method for calculating the convection heat
transfer coefficient. The heat transfer coefficient can either be calculated using the Dittus-Boelter
correlation or Gnielinski correlation or be specified as a constant. Figure 3-6¢) presents the Surface
Radiation primitive heat transfer element that is used to model radiation heat transfer between surfaces.
The Surface Radiation element can be connected between two solid nodes or between a solid node and
a flow node or between two flow nodes. The radiation heat transfer element cannot be connected to a
flow element. In the calculation, the radiation heat transfer element solves the standard radiation heat

transfer equation shown in the preceding subsection [102].

St

Flue gas M Flue gas

— o o >

Heat transfer Heat transfer
T P ‘ .
| i |
— N o—— P ——n
a) Steam b) Steam

Figure 3-7: Examples of simple Flownex networks in which heat is transferred from a flue gas stream to a
steam stream: a) Complete standalone Flownex network and b) Compound component which represents the

repeatable part of a network.

Figure 3-7a) presents a simple Flownex setup where by convention (because of heat transfer arrow)
heat is transferred from the flue gas stream to the steam stream via the CHT element. If the heat transfer
direction is opposed to the prescribed direction, then the heat transfer value will be a negative. On both
streams, inlet temperature and pressures are prescribed and a mass sink at the outlet of each flow
element is prescribed. If this simple Flownex network is repeatable in a bigger network, for instance if
the computational domain is discretised to smaller increments for better accuracy, then the repeatable

part can be lumped together to form a Compound component as shown in Figure 3-7b). This
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Compound component is then used a building block for the bigger network. Geometrical information
can differ between different Compound components. It can be seen that the compound component no
longer has the nodes and boundary conditions at both the inlet and outlet of each stream. These
components are replaced by the open fibre component (small squares at each end of both streams)
which implies that the compound component can be treated like an element on which boundary

conditions are not specified.

3.2. Tube modelling methodology

In order to model a specific heat exchanger in the convective pass of a coal fired boiler using the
network approach encapsulated in Flownex, a specific methodology was developed. This developed
methodology is generic such that it can be applicable to any heat exchanger of interest in the convective
pass of a coal-fired boiler. In this methodology the basic building blocks remain the same, while only
the geometrical information and tube arrangements will vary according to the specific heat exchanger

geometry.

Since most heat exchangers in the convective pass of a coal-fired boiler are mostly made of plain tubes,
the methodology is centred around modelling a tube or a bundle of tubes. Now consider a plain tube

with steam flowing inside and flue gas cross-flowing over it as shown in Figure 3-8.

Steam flow
 ———  ———
o o
¢ °
L °
o — —_—
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— o © e ©
L °
° °
°
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Figure 3-8: Representation of a tube with steam flowing inside and flue gas cross-flowing over it.

In order to correctly model the fluid flow and heat transfer in this tube on Flownex one should focus
on the cross-section of the thickness of the tube as shown in Figure 3-9. The directions of the arrows

shown in Figure 3-9 represent the assumed direction of heat transfer, which may initially be unknown.
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The actual direction of the heat transfer will be part of the solution and if it is different from the
assumed direction, the calculated value of the heat transfer will simply be negative. This approach
ensures that the directions of the heat transfer are always correctly calculated rather than being

specified, since the directions may in fact change during a transient event.

The heat transfer through the pipe wall is assumed to be axially symmetric and for the thin-walled

superheater tube it is assumed that heat conduction is in the radial direction only. The vertical arrows
in Figure 3-9 represent the steam (st) and flue gas ( fg ) flow paths respectively. 4, , and &, , are

the total enthalpy of the steam at the inlet and outlet respectively. 4, .. and h, ,, are the total enthalpy

of the flue gas inlet and outlet, respectively.
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Figure 3-9: Schematic of a single heat exchanger pipe increment (not to scale) [1].

The thermal inertia associated with the mass contained in the control volume of each node is also
accounted for by solving the transient form of the energy conservation equation for the node. For

example, for the cross-hatched control volume in Figure 3-9 the energy conservation would be

07;7 - Qpi +Qp0

o (pve), G-11)

with ( ,OVC)p the heat capacity of the pipe wall material within the control volume, and Q'pi and on

the heat transfer rates into the control volume from the inner and outer pipe wall surface temperature
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nodes respectively. The conductive heat transfer rates are calculated using a simple Fourier conduction

heat transfer equation in cylindrical form.

The representation of the fluid flow and heat transfer in a tube on Flownex is shown in Figure 3-10.
This represent the display of the compound component which is a heat exchanger increment. Such heat
exchanger increments are connected one after the other until the desired geometry and level of
discretisation is achieved. The compound component can represent one tube or a tube bundle. The

lumping of tubes into one is demonstrated in Figure 3-11.

Steam inlet

Flue gas inlet l Flue gas outlet

Steam outlet

Figure 3-10: Representation of a heat exchanger increment [1].

}

/) SIMPLE

i

Figure 3-11. Schematic of bundling inline tubes into one tube (the four tubes were selected for demonstration

purposes) [49].
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If the heat exchanger element represents a tube bundle, then great care must be taken in accounting for
surface area, free flow area and diameter. The surface area is important for correctly capturing the heat
transfer into the tubes, thus it should be the sum of all the surface areas in the grouped tubes. The free
flow area is important to characterise the amount flowing in the tube bundle thus it should also be the
sum of the free flow areas of all the tubes. On the other hand, the hydraulic diameter should be the
same as that of a single tube since it is used in the calculation of the heat transfer coefficients. The
hydraulic diameter is kept the same as that of one tube through manipulating the inlet

circumference/perimeter on the Flownex inputs using the following equation

4A
P :% (3.12)

tube

with P, A and D,,, the perimeter, free flow area and hydraulic diameter, respectively.

freeflow

Figure 3-12 shows the heat exchanger increment sub-network within Flownex which is embodied in

the compound component shown in Figure 3-10.

Inner pipe surface
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Figure 3-12: Schematic of the heat exchanger increment sub-network in Flownex.

The different elements are: the steam flow path; the inside convective heat transfer resistance; the inner
fouling layer (IFL) conduction heat transfer resistance and associated thermal inertia; two conduction

heat transfer resistances and thermal inertia components, each associated with half of the pipe wall
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thickness; an outer fouling layer (OFL) conduction heat transfer resistance and associated thermal
inertia; the outside convective heat transfer resistance; the outside gas radiative heat transfer resistance;
and the flue gas flow path. As mentioned in the subsection above, since the Surface Radiation primitive
heat transfer element cannot be connected to a flow element, it is connected to a node. Then the node
is connected to a convection primitive heat transfer element which in turn connects to the flow element
on the flue gas side. The heat transfer coefficient in this “pseudo” convection heat transfer element is
set to be very high, thus it induces negligible thermal resistance on the system. Note that in this
compound component the set direction for heat transfer is from the steam side to the flue gas side. If
the calculated heat transfer direction is in the opposite direction, then the heat transfer value will be a

negative.

It is important to mention that at this stage only the radiation from the surrounding flue gas is
considered, hence referred to as gas radiation. Long range radiation that originates upstream of the heat
exchanger of interest, henceforth referred to as direct radiation is not considered here. However, the
direct radiation will be considered as part of the modelling methodology refinement process in chapter

6 and implemented in chapter 7.

3.3. Summary of the modelling methodology

In this chapter, the network methodology which is encapsulated in the Flownex software was
described. In addition, the Flownex components that are used in the model development were also
highlighted. Furthermore, a model for a generic cross-flow heat exchanger increment was developed
that can be used to construct a heat exchanger of any complex layout. This heat exchanger increment
is a building block which is used in this work to build models of different superheater heat exchangers
of coal-fired boilers. Such heat exchanger increments can be connected one after the other until the
desired geometry and level of discretisation is achieved. It accounts for all the thermal resistances in
the heat exchanger including inner and outer fouling resistances using experimental correlations or
specified constant values. The resulting heat exchanger model can simulate steady state and transient
operations of the heat exchanger. At this stage the heat exchanger increment only falls short of
accounting for direct radiation which originates upstream of the heat exchanger of interest. However,

that phenomenon is addressed in chapter 6 where the modelling methodology is refined.
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4. METHODOLOGY DEMONSTRATION

Disclaimer: A significant amount of the work documented in this chapter was published in the journal Heat Transfer
Engineering in reference [2]. The candidate is indebted to his supervisor together with whom the concept was developed.
In order to demonstrate the modelling methodology described in the previous chapter, it is applied in
this chapter to model the primary superheater of a 50-MW coal fired power plant. This superheater
was also modelled by Taler et al. [4] as well as other researchers in Poland. This superheater was
chosen because most of its geometrical and flow information is available in literature. Some of the

results reported in the literature were also used for model verification.

4.1. Geometrical and flow information

The primary superheater is the heat exchanger on the right in the convective pass of the coal fired
boiler as shown in Figure 4-1. Based on its location in the convective pass, it is a convective

superheater since its most dominant mode of heat transfer is convection.
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Figure 4-1: The convective pass of a 50-MW coal-fired boiler system [48].

Figure 4-2 shows the schematic of a single tubesheet/element of the primary superheater. It is a 12-
pass dual-tube superheater heat exchanger where each pass consists of two tubes in “parallel” with a

complex flow arrangement inside the boiler. These inline tubes are connected on one end to the same
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inlet header and to an outlet header on the other end forming one tubesheet. Therefore, a tubesheet is

made up of 24 tube passes along the length of the flue gas flow path.

Outlet Inlet header
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Figure 4-2: Schematic of the layout of a single tubesheet of the superheater [2].

The inlet header is connected to the 8" pass from the flue gas inlet side and the tubes then follow a
complex arrangement with the outlet header connected to the 1% pass at the flue gas inlet side. The
superheater consists of 74 such tubesheets in parallel across the width of the flue gas duct, and the
width of the flue gas flow path associated with each tubesheet is 104 mm. All the tubes have an outer
diameter of 42 mm and wall thickness of 5 mm. The heights of the first and the last tube with respect
to the flue gas flow are 5.34 m and 4.46 m, respectively. The height of tubes in between is assumed to

be linearly distributed.

At nominal operation the steam capacity of the boiler is 210 x 10* kg/h. The live steam parameters at

the outlet of the final superheater are: pressure of 9.61 MPa and temperature of 532.8 °C [103].

4.2. Superheater Flownex models

A characteristic of the Flownex network modelling approach is that the steam flow model traces the
path travelled by the steam in the superheater. The flue gas flow model also traces the flue gas flow
path. However, in this work the subchannel approach was adopted on the flue gas side. The foundation
of the subchannel approach is that it explicitly assumes that one of the velocity components in a certain

direction is dominant compared to the other velocity components [58, 60]. Thus, only the dominant
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direction is rigorously solved, hence the only computationally expensive direction. In this case, the
flow is predominantly flowing from the inlet of the heat exchanger to the outlet, thus that is the flow
direction adopted. This implies that flow in the vertical direction or across the width of the heat
exchanger is ignored. Thus, mixing and recirculation in certain cases cannot be accounted for. This
assumption can be relaxed at a later stage of the modelling. A more detailed study of the flue gas side

modelling will be presented in Chapter 5 of this work.

The input data required for these models are inlet temperatures, inlet pressures and mass flow rates for

both the steam and flue gas streams. In addition, geometrical information of the superheater is required.

4.2.1. Single tubesheet

Figure 4-3 shows a schematic of the 1-D network model of a single tubesheet. It consists of 24 columns

of tubes along the flue gas flow path.
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Figure 4-3: Schematic of a tubesheet model in Flownex with tubes that are not discretised.

The flow path layout on the steam side reflects the complex layout in accordance to the geometrical
information shown in Figure 4-1 and Figure 4-2. In the tubesheet model shown in Figure 4-3, the tubes
are not discretized along the length in each steam pass. In order to obtain a refined computational
domain, the tubes may be discretised along the length in each pass as shown in Figure 4-4. Figure 4-4
also shows a single tubesheet model but with each of the tubes in each pass discretized into four
increments along the length of the tube. The specific number of four increments is for illustrative

purposes only and therefore in this case a single tubesheet is represented by 96 individual heat
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exchanger increments. A representation of one such a heat exchanger increment was presented in

Figure 3-10.

Steam outlet Steam inlet

T p

Flue gas inlet
Flue gas outlet

Figure 4-4: Schematic of a discretised tubesheet model in Flownex.

From the single node representing the extraction point on the inlet header the flow splits into two
separate nodes, each representing the inlet of one of the two tubes in the 8th pass. Each of the two
tube outlets in the Ist pass is also represented by a node which then converge into a single node
representing the connection to the outlet header. The gauge-like symbols connected to the inlet and
outlet nodes represent boundary values. The boundary values provided at the inlet node are the inlet
steam pressure and temperature. The outlet boundary value is a mass sink. Although the outlets of the
two tubes converge into a single node, there may be different resultant mass flow rates and
temperatures emanating from each. We may therefore detect flow and temperature maldistribution

between the two tubes due to different heat transfer and pressure drop conditions that may occur.

On the flue gas side, the overall inlet and outlet are each also represented by a single node. At the inlet
this implies an assumption of a uniform pressure and temperature distribution over the height of the
flue gas duct. This assumption is temporarily adopted for demonstration purposes but can be relaxed

later. The flue gas inlet pressure and temperature boundary values are specified. From the single inlet
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node, the flow splits to individual inlet nodes to each of the flue gas flow increments along the height
of the flue gas duct. The boundary value specified for the single node at the outlet of the flue gas flow
path is a mass sink. The single node implies that a uniform pressure distribution is assumed, which
may also be relaxed later on. This single overall outlet node is fed by a convergence from the outlet
nodes of each of the flue gas increments. The mass flow rates and temperatures coming from the outlet
of each of the flue gas increments may be different due to different heat transfer and pressure drop
conditions. Possible flow and temperature maldistribution along the height of the flue gas duct may

therefore also be detected.

Note that for now it is also assumed that there is no mixing in the vertical direction between the flue
gas increments as it flows through the heat exchanger. This assumption may also be relaxed by adding
interconnecting flue gas flow path elements in the vertical direction between the various inlet and outlet
nodes of the flue gas increments. In order to obtain the appropriate detailed flue gas flow pattern within
the tubesheet it will require the specification of appropriate representative flow resistances within the
horizontal and vertical flow path elements. However, for the purpose of demonstrating the modelling

methodology the vertical mixing will be assumed to be negligible, as was done by Taler et al. [4].

4.2.2. Multiple tubesheets

In order to investigate the effect of maldistribution of flow and temperature across the width of the
superheater heat exchanger a 3-D model was developed. This showed the extended capabilities of the
1-D network approach. In this 3-D model a tubesheet with one heat exchanger increment (not
discretised) was used, as shown Figure 4-5. A wide selection of input temperature data can be
postulated for both fluid streams. This includes postulating both uniform and non-uniform temperature
profiles for the flue gas inlet temperature. For this model as well, mass flow rate is set up as a sink at
the end of each stream. The mass flow is allowed to redistribute in accordance to the heat transfer and

pressure drops in the system.
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Steam outlet header

+“—— Steam inlet header

Figure 4-5: Demonstration of header flow modelling [2].

4.3. Heat transfer models, pressure drop models and fluid properties

The following subsections will describe the models applied for each of the heat transfer and flow
elements shown in Figure 3-12 in more detail.
4.3.1. Inside convective heat transfer element

The inside convective heat transfer is calculated using the Gnielinski [104] correlation for heat transfer
during turbulent flow of gases and liquids through pipes. The equations are incorporated in the element

using the steam side C# script as illustrated in Figure 3-12. This correlation is given by

(¢/8)RePr

- 1) 4.1
N 1+12.7\/%(Pr2/3—1)[H(d’/l’) } @b

where
&=(1.8log, Re-1.5)" 4.2)
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Here, Nu, Re and Pr are the dimensionless Nusselt, Reynolds and Prandtl numbers, respectively.

d, and /, are the fouled inner tube diameter and the length of the tube respectively. The ranges of

validity for this correlation are as follows

10* <Re<10°
0.1<Pr<1000
d/l <1

4.3.2. Conductive heat transfer elements

The tubes are made of carbon steel with a constant density and specific heat capacity with the value of

7832 kg/ m’ and 0.29 kJ / (kg D(), respectively. The thermal conductivity of the tube material was

given by Taler et al. [4] in W/(m[K) as

k, =35.54+0.004084 [T —2.0891x107° [T? (4.3)

with the temperature 7 given in °C. Unfortunately, no information was provided about the accuracy
of the correlation, but for the temperature range of 300°C < T < 500°C considered in this work, the

thermal conductivity variation is small, namely 32.4 < k <34.9W/(m X ), with an average value of
33.8 W / (m K ) The thermal conductivity of the outer fouling ash layer was also taken from Taler et
al. [4] and its value is k,, =0.07 W/(m[X). The inner scaling thermal conductivity was taken from

Trojan and Taler [105] and its value is k., =0.15 W/(mX).

scale

4.3.3. Outside convective heat transfer element

The outside convective heat transfer is calculated using the Gnielinski [106] correlation for cross-flow

around a single tube incorporated using the flue gas side C# script. This correlation is given by

1,turb

Nu, =03+ [Nu?,, +Nu? (4.4)

Here, Ny, is the overall Nusselt number and Nu,,. and Nu,,, are the resultant laminar and

turbulent Nusselt numbers given by
Nu,,, =0.664[Re, 3/Pr (4.5)

and
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0.037Re Pr
1+2.443Re;"" (Pr**-1)

(4.6)

Y turb

The validity ranges for correlations given by eq. (4.5) and (4.6) are as follows

10<Re, <10°
0.6 <Pr<10’

where the Reynolds number is given by

_p

Re,
U

“4.7)

Here, p, v and p are the fluid’s density, velocity and dynamic viscosity, respectively. [ is the

streamed length given by
1="4 (4.8)
where d is the fouled outer tube diameter.

4.3.4. Outside radiative heat transfer element

The radiation heat transfer from the flue gas to the outer surface of the outer fouling layer is accounted

for using Taler and Taler’s [107] radiation heat transfer coefficient given by

4 4
:JDfeqm'E#[ng—TfOJ
als+e, \T,-T,

4.9)

ro

with O the Stefan-Boltzmann constant given as 5.67%107° W/ (mzK 4). The equivalent emissivity of

the tube, £, is given by

g =25 (4.10)
“ 2-¢ '

with £ the emissivity of the outer tube wall or the outer surface of the OFL. The geometric mean

beam length, S is given by

14
s=C— 4.11
a1 (4.11)
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v is the volume around the tube occupied by the flue gas and A is the outer surface area of the fouled
tube. According to Jones [108] in many practical systems the constant ¢ is taken as 3.6. The

absorptivity coefficient & is given by

a:_ln(l—sg)

S

(4.12)

The total emissivity of the gas ¢ is determined using the weighted sum of gray gases (WSGG) model
from Hottel and Sarofim [109]. The total emissivity for the WSGG model is calculated from
! AT L
£, = Za”(T)[l e } (4.13)
i=0
where a, , denotes the emissivity weighting factor for the i-th gray gas, which is based on the gas

temperature T. The quantity in square brackets in eq. (4.13) is the emissivity of the i-th gray gas, with

absorption coefficient k;, partial pressure p and geometric mean beam length S. The weighting
factor a, , is the fraction of the amount of black-body energy in the spectral regions where a gray gas

of absorption coefficient K, exists. For the “transparent” windows in the spectrum between the spectral

bands of high absorption, the absorption coefficient is assigned a value of zero. This transparent

window is treated as a gas at i =0, with a weighting factor given as

a,=1->a,, (4.14)

This implies that only I values of the weighting factors have to be determined. Smith et al. [110]
presented a polynomial for evaluating the temperature-dependent emissivity weighting factor which is

given by
J .
Qe = b,y T (4.15)
j=1

where b, ; are the emissivity gas temperature polynomial coefficients. For this model the emissivity

coefficients given by Smith et al. [110] for a typical flue gas mixture were used and are given in Table
4-1. These coefficients are applicable for gas and irradiation temperatures within the range 600 to

2400 K and partial pressure-path length product within the range of 0.001 to 10.0 atm-m.
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Table 4-1: Coefficients for emissivity [110].

k. b,, %10 b, ,x10" b, %10’ b, ,x10"
0.4201 6.508 -5.551 3.029 -5.353
6.516 -0.2504 6.112 -3.882 6.528
131.9 2.718 -3.118 1.221 -1.612

The default radiation heat transfer element within Flownex solves the standard radiation heat transfer

equation given by eq. (3.10)
Qrad :AOFJ(TE; _T;;)
with A the outer fouled tube surface area and F the view factor. This standard radiative heat transfer

element is used in the current model but with the view factor given by

ok, @3
F=—% g

o 4.16
als+e, @10

Using this view factor together with eq. (3.10) is equivalent to employing the effective radiation heat
transfer coefficient given by eq. (4.9).

4.3.5. Steam side pressure drop

The pressure drop on the steam side is calculated according to eq. (3.4) which can be further expressed

as
S
Np,, =| L || = 4.17
Do, (diJ(szv ( )
where f is the friction factor and for turbulent flows it is given by Swamee and Jain [111] as
f= 0.25 (4.18)

[1og{(€/(3-7di))+(5-74/Reo'9 )}T

with e the inside wall surface roughness in [ um]. The validity ranges of friction factor correlation

are as follows
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5000<Re<10°

107 s(ijsw‘z
d

4.3.6. Flue gas side pressure drop

The pressure drop on the flue gas side is accounted for using the following equation

2
Ap :Eu'oTVZ 4.19)

where z is the number of tube rows and E « is the Euler number which was experimental determined

and given graphically in Zukauskas and Ulinskas [112].

4.3.7. Steam properties

The model used the built-in steam properties in Flownex. These properties are based on the standards

of the International Association for the Properties of Water and Steam (IAPWS) [113].

4.3.8. Flue gas properties

The flue gas specific heat capacity was calculated from the measured values using the energy balance

between the steam and the flue gas as follows

ep, =— 2 (4.20)

mfg (Tfye - Tfyi)

where @, is the heat transferred to the steam and r, is the mass flow rate of flue gas. T,, and T,, are

the inlet and outlet flue gas temperatures, respectively.

The flue gas density was accounted for using the ideal gas law given as

— P fg
=_"Jf9 4.21
pfg ngng ( )

where p ., T, and R, are the pressure and the specific gas constant, respectively.

f9

The flue gas dynamic viscosity was given as a constant. This constant value was the average of the

inlet and outlet dynamic viscosities each calculated using the following formula

T
= - fg , - (4.22)
1.2635x107 +1.2618x10*T, —1.6929T;

'ufg
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with the temperature given in Kelvin [114].

The flue gas conductivity was specified as a constant. This constant value was the average of the inlet

and outlet conductivities each calculated using the following formula

k,, =9.1446x107° x0.99958" xT;"*° (4.23)

with the temperature given in Kelvin [114].

4.4. Results

This section presents some results to demonstrate the capability of the model. At first a comparison is
made with the results presented by Taler et al. [4]. For reference purposes another comparison is then
made with results from lumped models while assuming different simplified heat exchanger flow
configurations. Following this, results are presented for low load operation and for transients due to
load changes. In these cases, only one tubesheet is modelled with the assumption that the flows and
inlet temperatures are evenly distributed across the superheater. Lastly, results of a study on the

maldistribution of flow and temperature are presented.

4.4.1. Comparison to results of Taler et al. [4]

This section provides a comparison between the results obtained using the new model for the 50 MW
coal-fired boiler’s primary superheater tubesheet with the results provided by Taler et al. [4]. Taler et
al. [4] presented the overall temperatures and flow rates for both the steam and the flue gas side at the
nominal operating conditions. This information is valuable since it is usually very challenging to obtain
suitable measured data from real operating power plants for model validation purposes. However, the
information available has some limitations due to the lack of detail and therefore it is not really possible
to provide a definitive validation based on this. Nevertheless, this comparison does at least provide
qualitative proof of the validity of the new model. The main limitation in the available data is that the
magnitude and distribution of the outer fouling layer thickness are not known. For this reason, Taler
et al. [4] assumed a uniform thickness distribution throughout the whole of the heat exchanger and
then adjusted the outer fouling layer thickness in the model until the computed and known steam
temperature differences over the whole superheater were equal. This same approach was therefore

applied here for the Flownex model.

The input conditions for the Flownex model are the same as those in the model of Taler et al. [4] and

are given in Table 4-2. According to Taler et al. [4] the resulting outlet steam and mean outlet flue gas

University of Cape Town 61



Chapter 4. Methodology demonstration

temperatures were 400.9 °C and 558.6 °C respectively. At a fouling layer thickness of 1.02 mm, the
corresponding outlet steam and outlet flue gas temperatures obtained with the Flownex model are
400.8 °C and 558.2 °C respectively. The detail temperature results obtained with the Flownex model
are compared to that of Taler et al. [4] for the steam, OFL and mean flue gas, as shown in Figure 4-6.
These were obtained using the Flownex model which was not discretised along the tube length as

illustrated in Figure 4-3.

Table 4-2: Input conditions for a complete superheater model with outer fouled tubes at full load.

Details Units Value
Inlet steam temperature °C 337.7
Inlet steam pressure kPa 9600
Steam mass flow rate kg/s 46.2
Inlet flue gas temperature °C 632.6
Inlet flue gas pressure kPa 100
Flue gas mass flow rate kg/s 64.5
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® Outer Fouling layer - TM|

®0 0g0 @ Flue gas - T™M
Oceo oe
o
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o0 Ogo
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Figure 4-6: Comparison of the Flownex model (FM) results with the results of Taler et al. [4] (TM).
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The results shown in Figure 4-6 illustrate that there is good agreement between the steam and flue gas
temperature profiles. Note that in the Taler model the two tubes in each pass were lumped together and
therefore it only provided one averaged value per tube pass. In the Flownex model the two tubes are
modelled separately and therefore it provides two distinct values per tube pass. For the steam side all
the average values of the two distinct temperatures determined with the Flownex model are within

2.7 °C of the averaged values from the Taler model, and also within 2.7 °C for the flue gas side.

However, there is a marked difference between the temperatures obtained for the outer fouling layer
surface. Possible contributing factors to this discrepancy include the following: the difference
introduced by the different convection coefficient correlations used in each model, the difference in
the flue gas properties used in each model, as well as the way in which radiation heat transfer was

accounted for in each model.

Heat is transferred from the flue gas to the OFL outer surface via convection and radiation heat transfer.
The distribution of the different modes of heat transfer through the fouled tubesheet as calculated with
the Flownex model is shown in Figure 4-7. The radiation contribution on the flue gas side is smaller
than that of convection and it also decreases with the flue gas temperature. Since this heat exchanger
is at the back of the boiler convective pass it is not unexpected that convection contributes more than
radiation. The total of the convection and radiation heat transfer is conducted through the outer fouling
layer and tube wall as shown in Figure 4-7. The calculated amount of heat transferred by one tubesheet
of the superheater with all tubes having a constant outer fouling layer of 1.02 mm thickness is
138.7 kW. This duty of the superheater tubesheet presents an error of 0.5% when compared to the
actual duty of 138 kW given by the Taler et al. [4] results.

A sensitivity study was performed to determine the effect of discretization along the length of each
tube by comparing the results obtained with one, two, three, four and five increments respectively. The
results obtained for the 12" pipe situated in the sixth pass of the tubesheet was chosen for comparative
purposes. The results are summarized in Table 4-3. It can be seen that the change in temperatures from
the model with one heat exchanger increment to that with five increments is less than one percent. This
implies that for steady-state scenarios the effect of discretization is negligible if uniform inlet flow
conditions and uniform outer fouling are assumed. For convenience, a one increment heat exchanger

model is used for the rest of the case studies presented here.
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Figure 4-7: A comparison between the different modes of heat transfer.

24

Table 4-3: 1.02 mm OFL thickness - The sensitivity of the outlet steam, tube, outer fouled layer and outlet flue

gas temperature to level of discretisation of the superheater tube sheet with uniform inlet flow conditions.

Number of heat
exchanger increments

Outlet steam

Average temperature for pipe 12 on the 6th pass (°C)

temperature (°C)

Inner pipe surface

Centre of the pipe

Outer pipe surface

Outer fouled pipe surface

Outlet flue gas
temperature (°C)

1 400.79 373.27 374.19 375.00 501.89 558.22
2 400.70 372.78 373.70 374.51 501.25 558.31
3 400.51 372.68 373.59 374.40 500.78 558.50
4 400.76 372.65 373.57 374.37 500.76 558.25
5 400.78 372.49 373.41 374.21 501.09 558.23

4.4.2. Comparison to results from lumped parameter models

As shown in Figure 4-2 the complex flow arrangement of the superheater actually consists of 24 cross-

flow heat exchangers configured in a combination of overall parallel flow and overall counter flow

configurations. This complex arrangement cannot be accounted for in any single one of the

conventional lumped parameter models. It is however instructive to systematically compare the results

of the complex model with that of the lumped models while assuming different simplified heat

exchanger flow configurations. The calculations for the lumped models are based on the Effectiveness
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— Number of Transfer Units (e—NT U ) method. The detail calculations performed here are shown in

Appendix A.

At first, an overall heat transfer coefficient (UA) value of 0.682 k V'%( was calculated from the known

average properties for one tubesheet based on the detail geometry and model described in Section 4.3.
The duties obtained with each of the conventional lumped parameter models while using this calculated
UA value are presented in Table 4-4 together with the resulting errors when compared to the known
duty of one tubesheet, namely 138 kW. The resulting errors vary depending on the configuration
assumed. The lowest error of +4.2% is obtained with the cross flow — both fluids unmixed
configuration and the highest error of +8.9% with the counter flow configuration. This raises the
question of whether these errors are indeed due to the fact that the complex flow configuration is not

taken into account, or whether it is simply due to the averaging of the overall UA value?

Table 4-4: Heat transfers and errors resulting from the different lumped parameter heat exchanger models

based on an overall AU value obtained from the average flow properties for one tubesheet.

Heat exchanger flow configurations Heat transfer (kW) | Error (%)
Counter flow 150.226 8.9
Parallel flow 146.097 5.9
Cross flow - Steam (Cmax) mixed, flue gas (Cmin) unmixed 148.265 7.5
Cross flow - both fluids unmixed 143.753 4.2

In order to evaluate this, another approach was followed. An overall UA value of 0.609 k% was

calculated based on a summation of the incremental UA values calculated in the detail Flownex model.
This re-calculated UA value therefore is not based on averaged properties as before, but rather take
into account the variation in the properties from tube to tube. The duties and errors obtained with the
different lumped parameters models when using this new overall UA value are provided in Table 4-5.
In this instance, the error associated with the pure counter flow model is lowest at only -0.07% while
the cross flow — both fluids unmixed error is highest at -4.2%. This shows that the overall counter flow
model can be used with confidence for steady-state normal operation, provided that an appropriate
value of the overall UA value is known, despite the fact that the complex flow arrangement is not
accounted for. This is not totally unexpected since it is generally accepted that the overall counterflow

approach provides good results for heat exchangers with four or more tube rows in series having a
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“generally” counter flow arrangement. This also provides further qualitative evidence of the validity

of the detailed incremental model when using the same overall UA value.

Table 4-5: Heat transfers and errors resulting from the different lumped parameter heat exchanger models
based on an overall AU value obtained from addition of the incremental UA values calculated in the Flownex

model for one tubesheet.

Heat exchanger flow configurations Heat transfer (KW) | Error (%)
Counter flow 137.864 -0.07
Parallel flow 134.739 2.3
Cross flow - Steam (Cmax) mixed, flue gas (Cmin) unmixed 136.375 -1.1
Cross flow - both fluids unmixed 132.104 -4.2

It can therefore be concluded that the earlier higher errors were mainly due to the inaccurate overall
UA value that was obtained based on the known average properties, rather than being due to the
assumed simplified flow configuration. One could now argue that by employing a simple logarithmic
temperature profile for the steam and flue gas sides (which is consistent with the lumped parameter
model assumptions) it would be possible to accurately calculate the incremental fluid properties.
Therefore, one should be able to obtain a more accurate overall UA value without having to take into
account the complex flow arrangement. However, the simple logarithmic temperature profile will not
properly “track” the changes in the fluid properties since it does not correspond to the actual
temperature profile. This is demonstrated in Figure 4-8. Figure 4-8 also shows that the logarithmic
temperature distribution inherent in the lumped model cannot account for the fact that the steam inlet

header is connected to the 15" and 16™ tubes rather than at the back end of the heat exchanger.
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Figure 4-8: A comparison of steam and flue gas temperatures for the counter-flow lumped model (LM) and

the detail Flownex Model (FM).

The shortcomings of the lumped model can be further demonstrated for more complex situations. One
such case is where there is a partial blockage of the steam flow in one of the two tubes. The results of
this are shown in Figure 4-9. Again, the inherent logarithmic temperature distribution of the lumped
model cannot predict the steam temperature distribution in this case. It seems logical that this limitation
of the lumped approach will also exist in even more complex cases such as non-uniform fouling or
scaling and during transient operations. Therefore, the lumped parameter models have limited scope

in the development of advanced tools for on-line process condition monitoring.
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Figure 4-9: A comparison of steam and flue gas temperatures for the counter-flow lumped model (LM) and

the detail Flownex Model (FM) where one steam tube is experiencing a partial flow blockage.

4.4.3. Load changes and low load operation

Due to the introduction of more intermittent renewable energy plants on the electricity supply grid,
coal-fired plants are forced to change their mode of operation from base load to variable load operation.
This will often involve ramp up or ramp down of the plant as well as running continuously at low load.
However, low load operations of coal-fired plants are not only due to the increase in renewable energy
generation. It is sometimes implemented simply to balance low demand on the grid. In addition, coal

plants sometimes run on low load if a draft group, mill or a feed pump has tripped.

If the steam temperature cannot be controlled adequately during load changes, for example if
attemperator sprays are already 100% open, the tube metal temperatures can increase to values which
might be above the design limit [115]. In certain cases, the firing rate may also not respond quickly
enough during a ramp down. Peet and Leung [116] demonstrated that during a quick load ramp up the

outlet steam temperature sharply increases to values above the design limit.

University of Cape Town 68



Chapter 4. Methodology demonstration

Load changes — from 100% to 50% in 15 minutes

The Flownex tubesheet model with clean tubes was extended to investigate the effect of load changes
on steam and tube metal temperatures. The input conditions given in Table 4-6 were used. In this case
a linear ramp down from 100% to 50% of the steam mass flow in 15 minutes was assumed. When the
outlet steam temperature moved above 421 °C it triggered a linear decrease in the flue gas mass flow

until the outlet steam stabilizes back to a temperature close to the set 421 °C.

Table 4-6: Input conditions for a complete superheater with clean tubes at full load.

Details Units Value
Inlet steam temperature °C 337.7
Inlet steam pressure kPa 9600
Steam mass flow rate kg/s 46.2
Inlet flue gas temperature °C 632.6
Inlet flue gas pressure kPa 100
Flue gas mass flow rate kg/s 40

The response of the outlet steam as well as the first pipe (with respect to flue gas flow path) and second
pipe wall temperatures are shown in Figure 4-10. The increase in outlet steam temperature corresponds
to an increase in the tube metal temperature. This has the potential to result in short term overheating.
The temperature response in Figure 4-10 demonstrates that the model does take into account the

thermal inertia of the system.
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Figure 4-10: The response of the primary superheater to a duty ramp down from 100% to 50% in 15 minutes.

Load changes — from 100% to 50% in 2 minutes

The detail Flownex model can also be employed to investigate the effect of different ramp rates on the
tube metal temperatures. Consider the same case with clean tubes as above and with the same initial
conditions as given in Table 4-6. If the load is now ramped down from 100% load to 50% in 2 minutes
rather than 15 minutes, the resulting temperatures are shown in Figure 4-11. Comparing the
temperature results in Figure 4-11 to that of Figure 4-10 shows that for a quicker ramp rate the tube

temperatures can reach higher values during the load change.
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Figure 4-11: The response of the primary superheater to a duty ramp down from 100% to 50% in 2 minutes.

Figure 4-12 illustrates that different tubes in the heat exchanger experiences different metal
temperature changes due to the heat exchanger duty change. For pipe number 1, which is at the inlet
of the flue gas path and outlet of the steam path, the metal temperature first increases and then reduces
to a value higher than the initial value. This response is different to that of pipe number 16 which is at
the inlet of the steam path and pipe number 24 which is at the outlet of the flue gas path. This figure
illustrates the importance of tube-by-tube discretisation along the flue gas path if detail tube analyses

is of interest. In such scenarios, lumped models will again fall short.
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Figure 4-12: The metal temperature response at the centre of selected pipes in the primary superheater to a

duty ramp down from 100% to 50% in 2 minutes.

4.4.4. Maldistribution of flow and temperature in the superheater header

Due to processes that occur upstream of the primary superheater in the boiler, flow and temperature
maldistribution may be present across the cross section of the duct. This maldistribution in turn leads
to maldistribution of steam flow and temperature going into the outlet header. These upstream
processes may include the firing pattern of burners, slagging on the water walls and the outer fouling
of heat exchangers preceding the primary superheater in the convective pass. The outer fouling pattern
of the primary superheater itself can also influence the flue gas flow and temperature profile across the
width of the heat exchanger. The experimental results shown by Gonzalez et al. [32] demonstrate the
typical temperature maldistribution profile across the width of the heat exchanger in the convective

pass that can be expected during operation.

A Flownex model made up by linking several individual tubesheet models was used to model the effect
of a postulated parabolic flue gas inlet temperature maldistribution. Only half of the boiler width is
modelled while assuming symmetry across the other half of the boiler. The results show the impact on
the flue gas flow distribution across the width of the boiler as well as the steam flow and temperature

distribution going into the outlet header.
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A schematic representation of the superheater model covering more than one tubesheet is shown in
Figure 4-5. Common nodes are created at the outlet of each stream, representing a section of the outlet
header on the steam side and a section of the duct on the flue gas side. Flow is pulled through the heat
exchanger on both the steam and flue gas sides by specifying negative mass sources at the respective
outlets. This allows the flow to redistribute in both streams with respect to the specified flue gas inlet
temperature profile and the interaction between the two streams. The steam inlet header temperature

is assumed to be uniform.

Parabolic inlet flue gas temperature profile — clean tubes

Consider a case of a parabolic flue gas inlet temperature profile as shown in Figure 4-13. It is assumed
that tubesheet 1, which is at the end near the water walls, experiences the lowest flue gas temperature
while the highest temperature is at the centre line of the flue gas duct. This can be expected since the
water walls are significant heat sinks. The boundary conditions are as follows: an average flue gas inlet
temperature of 787.3 °C, a total flue gas mass flow rate of 57.5 kg/s, a uniform steam inlet temperature

of 317.2 °C and a total steam mass flow rate of 49.5 kg/s.
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Figure 4-13: Flue gas flow and temperature distribution from the different tubesheets along the half header
length.
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The outlet flue gas temperature for clean tubes is expected to follow the same parabolic profile as
shown in Figure 4-13. The flue gas mass flow automatically redistributes such that more mass flows
through the region with lower temperature. Corresponding to the higher flue gas temperature towards
the middle of the superheater, the outlet steam temperature is higher in that region compared to the

sides, as demonstrated in Figure 4-14. The steam mass flow is also lower in the region with higher

temperatures.
600 - 0.74
O Steam inlet temperature (°C) 00 Steam outlet temperature (°C)
X Steam mass flow rate (kg/s)
550 - og 0.72
oot O
DDDDDD
o4
X 0O —_
=500 { “x o- - 0.70 2
o X « - 0 O = . _u!o
g X% oU @
O L=
S X6 e
£ 450 - 0% x - 0.68
- O X 3
Q. 0 O X % =
(] O X ]
F 400 ¢ ><><>< - 0.66 S
X X % S
XXX X X % %
350 - - 0.64
O00O0O0O0OOOOOLOLOLOOLOLOOLLOLOLOLOLOOLOLOLOLOOLOLOLOOLOLOOO
300 T T T T T T 0-62

1 6 11 16 21 26 31 36
Tubesheet number

Figure 4-14: Steam flow and temperature distribution from the different tubesheets along the half header

length.

The mass flow maldistribution in both the steam and flue gas passes is due to variations in the heat
transfer rates, which result in variations in the fluid properties such as the density. This in turn
influences the respective velocities and pressure drop characteristics that are taken into account in the

momentum conservation equations.

Figure 4-15 shows the top view map of the associated flue gas temperature distribution across half of

the flue gas duct. It clearly shows the higher temperatures in the middle compared to the sides.
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Figure 4-15: Top view of the flue gas temperature distribution across half of the superheater.
Parabolic inlet flue gas temperature profile — outer fouling around the centre line

During operation outer fouling usually begins to occur on the tubesheets situated closer to the centre
line of the flue gas duct. This is one of the factors that result in the so-called M-type outlet steam
temperature profile, where the outlet steam temperature profile has two or more peaks along the outlet
header. This can be seen in the experimental results of Gonzalez et al. [32]. Such a temperature
maldistribution leads to uneven expansion of the outlet header, thus contributing to thermal fatigue
through the development of bending stresses. This bending can also contribute to tube-to-header weld

cracks.

Using the half superheater model with the inlet conditions the same as those given for the clean case
above, outer fouling of varying thickness was postulated from tubesheet number 31 to 36. Figure 4-16
demonstrates that the outlet steam temperature peaks around tubesheet number 30 and then dips
towards the centre line of the superheater. This corresponds to an increase in flue gas outlet

temperatures for tubesheets numbers 31 to 36 as shown in Figure 4-17.
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Figure 4-16: Steam flow and temperature distribution from the different tubesheets along the half header
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Figure 4-18 shows the top view map of the flue gas temperature distribution across half of the flue gas

duct that resulted due to the outer fouling. Comparing Figure 4-18 to Figure 4-15 it can be seen that

the higher temperatures in the middle region where there is more fouling propagates further

downstream due to less heat transfer in the centre region.
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Figure 4-17: Flue gas flow and temperature distribution from the different tubesheets along the half header

length with outer fouling on the middle tubesheets.
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Figure 4-18: Top view of the flue gas temperature distribution across half the superheater with increased

outer fouling around the centre line.
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4.5. Summary: Application case study

This chapter presented the application of a one-dimensional network approach to model a superheater
heat exchanger with complex geometry. This approach consists of iteratively solving the transient
mass, energy and momentum conservation equations for each of the one-dimensional increments that

are used to construct the complex three-dimensional geometry.

The heat exchanger geometry is discretized along the flue gas flow path as well as along the steam
flow path. Each heat exchanger increment contains the appropriate geometrical information and
thermal resistance characteristics. Empirical correlations were employed to model the inner and outer
convective heat transfer. An effective heat transfer coefficient was used to account for radiation heat

transfer. The emissivity of the gas was calculated using the weighted sum of gray gases model.

The results of a case with outer fouling were compared with results provided by Taler et al. [4]. The
comparison was satisfactory for both the steam and flue gas temperatures. However, discrepances were
observed in the predicted outer fouling layer temperatures. This is most probably due to the differences
in the heat transfer correlations, flue gas properties and thermal radiation models used in the two
different approaches. This comparison provides at least a qualitative validation of the model since a
definitive validation is not possible due to a lack of detail data. This also highlights the need for more

detailed real plant data in order to facilitate model validation studies.

Some of the advantages of the discretized Flownex model were illustrated when compared to lumped
models based on the effectiveness-NTU method. The “black box™ approach inherent in the lumped
models falls short if detailed analysis of the temperature distributions through the heat exchanger is
required. For instance, the lumped models cannot correctly predict the steam temperature distribution
with a steam inlet that is not situated at the end of the heat exchanger, as well as when there is partial
steam flow blockage in one of the tubes. However, for the most simple steady-state normal operation
case the results of the discretized model corresponds very well with the overall counter flow model,
provided that the correct overall UA value is employed in the lumped parameter model. This provides

further confidence in the validity of the discretized model.

The capability of the model to study the effect of different ramp rates during load change on the tube
metal temperatures was demonstrated. The results showed that higher metal temperatures are predicted
on some tubes at increased ramp rates, even during load reduction. In addition, the tube by tube

discretisation makes it possible to observe the different responses of the different tubes during load
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changes. The ability to calculate the flow and temperature maldistribution within a tubesheet as well
as across the width of the superheater was also demonstrated. The results showed the formation of an
M-shaped outlet steam temperature profile due to increased outer fouling near the centre line of the

flue gas duct.

It is envisaged that models based on a transient one-dimensional network approach such as the one
proposed here, can be employed to study complex thermofluid process phenomena that may occur
during intermittent, transient and low load operation of power plants. This could form the basis for
improving operations and for the development of advanced tools for on-line process condition

monitoring in support of preventative maintenance and outage planning.
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5.FLUE GAS SIDE MODELLING

Up to this point in the dissertation, the sub-channel approach has been adopted in conjunction with the
pipe network approach on the flue gas side modelling of the heat exchanger. In essence, only the
dominant flow direction is modelled [58, 60]. This chapter investigates the limitations inherent in the
network approach to fully account for the momentum conservation on the flue gas side within the heat

exchanger, in an effort to illuminate the applicability of the proposed model.

In the convective pass of most boilers, the flue gas flows in a cross-flow configuration over several
tubes and exchanges heat with the steam inside the tubes via convection and radiation. In principle,
these arrays of tubes that are making passes in the boiler are obstructing the flue gas flow. This flue
gas flow is governed by the fundamental physical principles of fluid dynamics. These principles are
mathematically represented by the mass, momentum and energy conservation equations, which govern
computational fluid dynamics (CFD) [117]. It is generally accepted that to obtain realistic flow patterns
on the shell side of heat exchangers the conventional CFD approach with finely discretised meshes on
a body-fitted coordinate system must be used. However, in engineering practise it is particularly time
consuming to employ the conventional CFD approach in modelling the flue gas flow side, thus often
alternative approaches are used. The CFD porous media approach and distributed resistance approach

are some of the alternative approaches used.

It has been established that in literature there is no consensus in the terminology used to refer to these
alternative approaches. Gomez et al. [33] employed the porous media approach yet stated that they
used the conventional CFD technique for their model. Also, there has not been any clear distinction
between the distributed resistance and porous media approaches. In most cases they have been
considered to imply the same concept as in the case of Coelho [25], who used the porous media
approach equations yet stating that he used the distributed resistance concept. The distributed
resistance concept was established in 1974 by Patankar and Spalding [34] to simplify the modelling of
shell and tube heat exchangers. In their initial equations, the effects of viscous diffusion were not
included. However, they stated that the inclusion of such effects in the calculation procedure presented
no difficulty. They also highlighted the possibility of such effects being neglected in densely-filled
spaces such as in shell and tube heat exchangers. Patankar [59] emphasized this where he initially

included the viscous action, effects of body forces and distributed resistance for the momentum
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equations and the effects of diffusion on the energy equation. He then stated that for a densely packed

volume, the diffusion terms can be ignored.

For consistency in this study the CFD porous media approach implies the approach that includes the
viscous diffusion terms, body forces and the distributed resistance term. The distributed resistance
approach implies the approach that eliminates the viscous terms since their effect is assumed to be
negligible compared to that of the distributed resistance coefficients. However, the body forces are still
present. In essence, the assumption that the volume is densely packed is adopted. Thus, the equations
are the same as those in the initial paper by Patankar and Spalding [34]. This brings into focus the

distributed resistance terms.

This chapter systematically investigates the applicability of the network approach in modelling two-
or three-dimensional flows. The adopted systematic approach consists of two parts; the conceptual
development and an illustrative example on a simplified model problem. The conceptual development
begins with the fundamental governing equations for the conventional CFD approach, then the porous
media approach, followed by the distributed resistance approach and then comparing the distributed
resistance approach with the pipe network approach. Therefore, the narrative flows from the general

to the simplified approach.

5.1. The governing equations

For the purposes of this study the flow is assumed to be isothermal and therefore the focus will be on

the mass and momentum conservation equations only.

Continuity equation

@Hﬂ[@spﬁ):o (5.1)

with €, p and u representing the local porosity, density and volume averaged velocity vector,

respectively. Furthermore, ¢ is time and O is the divergence or vector gradient operator.

The local porosity is defined as the ratio of the volume of the void spaces where the fluid flows to the

bulk volume of a porous medium [74] and it is given as

V.
£= Sfluid (52)
Vbulk
where V., is the volume of the fluid and V), is the total volume of the system.
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Momentum equation

a(gfﬁ—) +0goi i) = —e0p + f + el +peRu (5-3)

where y represents the dynamic viscosity and a Newtonian fluid is assumed. f represents the body

forces and R represents the coefficients of the hydraulic distributed resistance due to the matrix

structure obstructing the fluid flow. § and ¢ are constants which are either zero or one depending on

which modelling approach is used.

The generalised mass and momentum conservation equations respectively given by eq. (5.1) and eq.
(5.3) represent the equations for the conventional CFD, porous media and distributed resistance

approaches. The difference for each approach depends on the choice of £, § and ¢ as shown in

Table 5-1. These differences are further discussed below.

Table 5-1: Summary of the corresponding constants to the governing equations.

Details £ ) Y
Conventional CFD approach 1 1 0
Porous media approach 0<e<l1 1 1
Distributed resistance approach 0<e<l 0 1

5.1.1.  Conventional CFD approach

The fundamental governing equations for the conventional CFD approach inherently characterised in
eq. (5.1) and eq. (5.3) in conjunction with Table 5-1 can be applied on an infinitesimally small element
fixed in space, shown in Figure 5-1, represented in a Cartesian coordinate system. Employing these
governing equations in modelling the flue gas in the convective pass of a boiler is always time
consuming in practise. This is due to catering for all the shear terms around the tube bank obstruction,
as illustrated by the velocity gradient shown in Figure 5-2. In addition, this would require a finely
discretised model, similar in topology but much finer than the one shown in Figure 5-3 for flow over
one tube. This makes the model computationally expensive. Thus, for practical engineering cases,

especially for online condition monitoring applications, an alternative approach has to be employed.
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Figure 5-1: Infinitesimal fluid element fixed in space with fluid moving through it.
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Figure 5-2: An example of a fluid flow case that can be modelled with the conventional CFD equations.

[ 7777 ANAVAVATAN
I}I[I///l// \\\\\\\\l\
] |
l\ f
| AR 7171 |
\\\\\\\\\\\ /[//////Il

i
|

Figure 5-3: An example of a particularly course CFD grid for simulating flow over a cylindrical object.
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5.1.2.  Porous media approach

The porous media approach has been widely used as an alternative approach in engineering practise to
model fluid flow in complex equipment with obstructions to flow e.g. shell and tube heat exchangers.
It is based on the continuum approach which regards the volume within the shell of the heat exchanger
as uniformly filled with fluid with distributed resistance to fluid flow [34, 74]. In essence, it models
the fluid as if it is flowing through a porous media where fine scale turbulence effects are modelled

via porous matrix resistance coefficients.

This approach allows for the use of a coarser grid similar to the one shown in Figure 5-4, compared to
that of the conventional CFD approach shown in Figure 5-3, hence improving the economy of
computer solving time and storage [34]. Another advantage of this approach is that it simplifies the
modelling of the fluid flow over the porous structure through incorporating the porosity and a function
that characterises the distributed resistance to flow. Usually, the functions to characterise the

distributed resistance to the flow due to the solid matrix are determined experimentally.
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Figure 5-4: An example of a computational grid for the porous media approach applied in an inline tube

arrangement.

The governing equations for this approach are similar to those of the conventional CFD approach (eq.
(5.1) and eq. (5.3)) with the addition of the porosity and the adjustment to incorporate the distributed

resistance as characterised in Table 5-1.
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5.1.3. Distributed resistance approach

The distributed resistance approach is a simplified version of the porous media approach, with the only
difference being that it does not account for the volume averaged superficial viscous actions as
characterised in eq. (5.3) in conjunction with Table 5-1. However, the underlying continuum approach

to fluid flow is also adopted in this approach.

The equations of the distributed resistance approach are solved on a computational grid similar to that
of the porous media approach illustrated in Figure 5-4. In these equations, the local porosity and the
coefficients of the hydraulic distributed resistance are based on the physical geometry of the enclosure
and obstacles through which the fluid is flowing. These may vary from location to location in the
enclosure depending on the variation in geometry. The determination of these two variables is

explained in detail below.

5.1.4. Porosity

In illustrating the local porosity determination, consider the local flow over a tube as shown in Figure

5-5 in either inline or staggered tube arrangement. Employing eq. (5.2) leads to a porosity of

2
g=1-"P (5.4)
4SLST

This formulation is depended on the chosen control volume.
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Figure 5-5: A control volume around a tube.
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5.1.5. Distributed resistance coefficients

The coefficients of the hydraulic distributed resistance are algebraically related to the solid geometry,
fluid properties and local velocity components. According to Patankar and Spalding [34] these
coefficients can be given by dimensional analyses as follows

L F{p”_f,ﬁ,&j (5.5)
ou U u u

with / representing a local length dimension, e.g. the tube diameter.# , v and W represents the local

velocity components in the x, y and Z directions, respectively in accordance to the Cartesian

coordinates system. From the dimensional analysis groups, a number of empirical correlations have
been developed by researchers in the past. Bell [118], Zukauskas [119], Oka et al [120], and Rhodes

and Carlucci [75] were amongst those researchers who contributed to this study.

Patankar [59] stated that the distributed resistance coefficients Ig for the ; - direction momentum

can be obtained from the friction factor correlations. The correlations are given in the form

__(L)(£er
o)) 5

with v representing the magnitude of the resultant local volume averaged velocity, D, representing
the hydraulic diameter of the interspaces between obstacles and ¢ representing the friction factor. The

latter is usually given by the formula of the form
E=cRe" (5.7
where ¢ and 7 are constants and Re is the local Reynolds number.

A further simplification of the distributed resistance approach would constitute the pipe-network

approach. These are compared next.

5.2. Distributed resistance approach versus the pipe network approach

In this section, the fundamental equations for the distributed resistance approach are compared with

that of the one-dimensional pipe network approach.
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5.2.1. Continuity equation
For the distributed resistance approach, the continuity equation is given by eq. (5.1) as

A(p)

o +0epir) =0 (5.8)

Integrating this equation over the control volume which is fixed in space (an illustration shown in

Figure 5-1) leads to
'm?dv +[[[odeom)av =0 (5.9)

Applying the divergence theorem from vector calculus, the second volume integral on eq. (5.9) can be

expressed as a surface integral as shown below.

JIVoiteomya = [ o) a5

where 5 =7nS . Here, 7 is the outward pointing unit vector to the surface area § of the control

volume.

Since the control volume used in these derivations is fixed in space, the limits of integration are

constant so the time derivative /0¢ can be placed outside the integral as shown below.

1% a7y =5 T ohav

Thus, eq. (5.9) becomes
%J‘_ﬂ(fp)dv +J;j£pﬁ%:0 (5.10)

The first term represents the rate of change of mass within the control volume and the second term

represents the net outflow of mass from the control volume.

Now, the continuity equation given by eq. (5.10) is applied on the network approach. Figure 5-6 shows
a computational domain for the network approach in three-dimensions. Here, the squares represent
nodes and circles represent elements. In the network approach, the conservation of mass is applied on
the nodes. A node represents a volume that contains a fluid with multiple mass flows into and out of
it. After the integration of eq. (5.10) over a 3-D control volume and simplifying the effective volume

represented by £V to v , the continuity equation can be simplified to
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op
ot

) :é(zm, —Zme) (5.11)

since the second term in eq. (5.10) represents the sum of all the mass flows out minus the sum of the

mass flows into the control volume. The continuity equation given by eq. (5.11) is exactly the same as

eq. (3.1) which was introduced in chapter 3 for the network approach.

A
O—IE]

Figure 5-6: A demonstration of the pipe network approach applied on a three-dimensional computational

domain.

The continuity equation for the distributed resistance approach may be solved on a similar staggered
grid as for the pipe network approach. Figure 5-7a) shows a control volume on a two-dimensional grid
on which the continuity equation of the pipe network approach, given by eq. (5.11), is solved. Figure
5-7b) shows a control volume on a two-dimensional grid on which the continuity equation of the
distributed resistance approach, given by eq. (5.1), is solved. Comparing the two diagrams shows that

the continuity equations for the two approaches are exactly equivalent.
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Figure 5-7: A control volume on a two-dimensional grid on which the continuity equation for a) the network

approach and b) the distributed resistance approach is solved.

5.2.2.  Momentum equation

The momentum equation of the distributed resistance approach, given by eq. (5.3) in conjunction with
Table 5-1, can be represented as
0( o)

T+D[@gpgmg):-émp+f+gkﬁ (5.12)

This equation can be transformed into an integral form using the Reynolds Transport Theorem,

resulting in

%I”ﬁspdv+”ﬁ(£pﬁ%)=215 (5.13)

The terms position in the equation corresponds to their positions in the partial differential form of the
equation given by eq. (5.12). The first term on the left represents the rate of change of momentum
within the control volume. The second term on the left represents the net flow rate of momentum out
of the control volume. The term on the right represent the sum of all the forces acting on the fluid

within the control volume, including the distributed resistance terms.

Now, consider a one-dimensional control volume fixed in space as shown in Figure 5-8. Since the
length of this control volume is very small and the change in area is also very small, the volume can

be approximated by Ad/.
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Figure 5-8: An arbitrary infinitely small one-dimensional control volume where A is the effective area [121]

fixed in space.
If it is assumed that the unit vector is always in line with the direction of the flow, eq. (5.13) can be

transformed from a vector equation to a scalar equation as illustrated below. In the 1-D control volume

shown in Figure 5-8, the rate of change of momentum is then transformed as follows
s [[weoav =3
— ||| ugpdV =—|(pv)Ad/
014y P ot (o)

with Ad¢ = edv which is the effective volume and A representing the effective cross-sectional area

in the porous structure.

The net flow rate of momentum out of the control volume is also transformed as follows
o 0
J] u (S,Ou WE) = {pva +&(pV2A) dﬁ} - oV’ A
S
0 >
=—|pvA)dl

o (v4)
The forces acting on the fluid in the control volume can be classified as surface and body forces. The
surface forces are due to physical contact with the fluid in the control volume and include forces due
to static pressure, frictional and secondary losses as well as pressure rises. Body forces are the forces
that do not need direct contact with fluid in the control volume and they include forces due to gravity,
magnetic field and electric field. In the network approach, only the component of the forces that are
aligned with the flow direction are considered. The forces can therefore be represented by scalars rather
than vectors. The net force due to static pressure on the fluid in the control volume is given by

% Ad?
14

The net force due to frictional and secondary losses where there is contact with the wall is given by
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-9 gqy
o

with p, representing the total pressure. The total pressure is the sum of the static pressure and dynamic
. 1, . . .

pressure given by p, = p +5 pv° . The net force due to pressure rise associated with work done on the

control volume is given by

Pow 44y
ol
The component of the body force due to gravity that is aligned with the flow direction is represented
as

0z 1 s

beodV = —-pg —
7Y ,Ogaf

Combining all these terms on a per unit volume basis in accordance with eq. (5.13) leads to

9
or

10 2.\ _ Op apOL apow 0z
+ 29 (2A) =P _%Po  OPow _ 5, 9% 5.14
(o) + 23 (pr4) ar o0 ar P83 (5.14)

valid for both compressible and incompressible flows. With further mathematical manipulations

together with the application of mass conservation, eq. (5.14) reduces to

ov ov  Op _Opy, Opy, 0z
v, v 0p_0Opyw _Opy _, 0z 5.15
P P ot T o Pl (5.1)

For incompressible flows, the second term on the left-hand side which represents the convective

momentum in the direction of the flow can be manipulated as follows

1,
@_a(zmj

PYar ™" o

Now, inserting this into eq. (5.15) and combining it with the static pressure term leads to

i
ov 2 - OPow _ Py, _pg%

=+ 5.16
P ot ol ol ol ol .10
Thus, the equation can be written in terms of total pressure as follows
0v _ 0p, _ Opyy _ 0Py, 0z
—+ = - -pPg— 5.17
or o o ot 8% G4
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Integrating eq. (5.17) over a one-dimensional control volume with a finite length L between the inlet

i and the outlet e with an average cross-sectional area A leads to

0
pLa_‘;-'-(pOe_pOi):pg(zi_ze)+Ap0W_APOL (5.18)

Further manipulations of the transient term results in

i _ A
or L

0
((pOi_pOe)-l-pg(Zi_Ze)-'-ApOW_APOL+Lva_fJ (5.19)

This is the transient one-dimensional form of the momentum equation for incompressible flow that
was introduced in chapter 3, eq. (3.3). This is now a single scalar equation aligned with the flow,

whereas the momentum equation for the distributed resistance approach consist of vector equations.

If a staggered grid approach is adopted in solving the governing equations of the distributed resistance
approach given by eq. (5.1) and eq. (5.3), in two dimensions the grid will be as shown in Figure 5-9.
The continuity equation is solved on the grid points as shown by its control volume while both the x
and y momentum equations are solved in between the grid points. The advantage of the staggered grid
is that it eliminates checkerboard solutions [122]. Furthermore, the pipe network approach adopted in
this work is based on a similar strategy. For the network approach, the governing equations given by
eq. (5.11) and eq. (5.19) are solved on the grid shown in Figure 5-10 which is imposed on the two-
dimensional staggered grid. In Figure 5-10 it is shown that the continuity equation is solved around
the nodes while the momentum equations are solved around the elements. As mentioned above, for the
continuity equation the terms solved for on the distributed resistance approach and the pipe network
approach are identical. However, for the momentum equation it is not the case. The difference is mainly

on the term which represents the net flow rate of momentum out of the control volume.

For further analyses, the term representing the convective momentum (second term in eq. (5.3)) in the
distributed resistance approach is expanded on the Cartesian coordinate system in three-dimensions.

Now, consider the component of this term in the x direction as shown below.

0(&out) N 0(&govu) N o(gowu)
Ox dy 0z

It illustrates that in addition to the momentum convected in the direction of the flow, it includes the

contribution of the momentum from the convected flux in the y and z directions.
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For the one-dimensional pipe network approach, the contribution of the y and z fluxes to the scalar
momentum equation cannot be taken into account at the east and west faces. This is because the term

representing the net flow rate of momentum out of the control volume (second term in eq. (5.14)) is

10
Z&(pva)

which only accounts for the flux in the direction of the flow. This term is then manipulated to represent

the dynamic pressure component and then added to the static pressure to form the total pressure.
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Figure 5-9: A two-dimensional staggered grid in the Cartesian coordinate system.
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Figure 5-10: Imposing the one-dimensional network approach grid on the two-dimensional staggered grid.

University of Cape Town 93



Chapter 5. Flue gas side modelling

According to Greyvenstein [100], solving for total pressure in the network approach presents an
advantage as it retains the convective momentum term in the direction of the flow but eliminates it in
the solution technique. However, accounting for only the convective momentum in the direction of the
flow implies that if the network approach is employed in two- or three-dimensional flows in an open

space might result in some inaccuracies. The magnitude of the error is flow physics dependent.

Furthermore, for the pipe network approach the pressure drop due to the pipe wall friction is given by

(N,
By, —(ZJ D (5.20)

where f is the friction factor and D, is the hydraulic diameter. On the other hand, integrating the x
momentum equation for the distributed resistance approach, which can be derived from eq. (5.3), over
a control volume with finite length L and an average cross sectional area A the resulting distributed
resistance force per unit area from eq. (5.6) is given by

1\éL ,
RuL=|=|>= 521
U (szh,Ou (5.21)

Equating eq. (5.21) to eq. (5.20) shows that
§=f
Thus, the distributed resistance coefficient of the distributed resistance approach can be correlated to

the friction factor in the pipe network approach.

5.3. Case study to systematically illustrate the differences between the approaches

This section aims to further analyse the applicability of the one-dimensional network approach in
modelling two-dimensional or three-dimensional fluid flow. To scrutinize the applicability of the
network approach, it is compared to other modelling approaches. A case specific quantitative analyses

is undergone to magnify the differences.

5.3.1. Model problem

To qualify and quantify the differences, the illustrative example which is schematically shown in
Figure 5-11 is used. This case is isothermal; thus, the energy equation is not solved. This example
involves an incompressible Newtonian fluid flowing through a single 2-D control volume of

dimensions one metre by one metre. The inlet x and y velocities are specified as 5 m/s and 1 m/s,
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respectively. The outlet static pressures are also specified as zero Pascals (this is relative to a reference

pressure as is the custom in CFD codes). The density of the incompressible fluid is given as 1 kg/m?.

In the following subsections, the various modelling approaches are used in a systematic format viz.:
distributed resistance approach, simplified distributed resistance approach, network approach and

custom developed 1-D steady state flow exact solutions.

A
A\ 4

1

1

1

1

1
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1

1
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1

1

Figure 5-11: A single 2-D control volume for an incompressible fluid.

Computational domain discretisation
For comparison purposes, this example is solved on a single 2-D computational cell using a staggered

grid as shown in Figure 5-12 for CFD and pipe network approaches.
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Figure 5-12: Model problem on the two computational domains: a) 2-D CFD staggered grid and b) the pipe

network staggered grid.
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The governing equations

The general governing equations for the distributed resistance approach and simplified distributed
resistance approach are shown below for the 2-D incompressible and isothermal case. The continuity

equation is given by

Ou Ov

—+—=0 5.22
ax  dy (5.22)
and the x and y momentum equations are given by
0(pu)  0(puu) d(pvu) _ _0p
+ + =———+
o T or TWrTgy T Al
(5.23)
d(pv) d(puv)  0(pw) _ _0p
+ + =———+
o TR g Tay T gy AP

with ¢, being a constant which is one for the distributed resistance approach and zero for the

simplified distributed resistance approach.

Distributed Resistance Approach

The distributed resistance (DR) approach is a form of the CFD porous media approach where the non-
matrix diffusion terms are ignored. This is due to the assumption that in a computational domain with
several obstacles most of the resistance to flow will come from the obstacles (porous matrix) rather
than the shear in the fluid. The 2-D equations of this approach are also generically given by eq. (5.22)

and eq. (5.23) with the constant {/,, =1.

Simplified Distributed Resistance Approach

The simplified distributed resistance (SDR) approach is a version of the distributed resistance approach
that only solves for convective momentum in the direction of the flow. Thus, it neglects the cross
convective momentum terms. The equations associated with the simplified distributed resistance

approach are also generically given by eq. (5.22) and eq. (5.23) but with the constant ¢, =0.

1-D Network Approach: Flownex

The network approach used in this work is encapsulated in Flownex. Flownex solves the 1-D
conservation equations of mass and momentum. The governing equations that Flownex solves are as

follows; the continuity at the intersection is given by
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D ity = i =0 (5.24)

and the momentum equation is given by

9,9 - o (5.25)
or o/ o/
where
Ap()L = ﬁpvz
with
f L

2 D,
Some similarities can be drawn from the simplified distributed resistance approach and the network

approach; however, their solution techniques differ, as emphasised below.

The pressure projection methodology is used as a solution technique for the distributed resistance
approach and the simplified distributed resistance approach. In this solution scheme the governing
continuity and momentum equations are solved on a staggered grid computational domain. The
pressure projection methodology is explained below. The network approach employs the Implicit

Pressure Correction Method on a staggered grid computational domain as explained in chapter 3.

Pressure projection methodology
Since the case modelled here is isothermal and incompressible, then the pressure projection

methodology involves three steps:
Step 1: Compute A" from

AW

= =0 Ou”) + uu” 5.26
=0 Da)+ 0 (5.26)
where

i

W_

PV

and

Aw =w'—W'
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with w' an intermediate momentum term.
Step 2: Solve for pressure implicitly from

0=0" 4 0 p™ +1D[QAV_VD) (5.27)

P p

using AW” computed in Step 1.

Step 3: Finally compute the velocity 1" which is at the next time step form

u™ =" A P +1AWF (5.28)
Jo Jo

where the pressures are as computed in Step 2 and Aw” computed in Step 1.

5.3.2. Inviscid flow

Consider an inviscid flow scenario of the model problem shown in Figure 5-11. The analytical solution
is trivial i.e. undisturbed flow. Applying the network approach using the computational grid shown in
Figure 5-12b), the computed velocity and static pressures are presented in Table 5-2. The network
approach predicts that the flow will split equally in both the x and y directions at the outlets. However,
practically this will not be the case, as the flow will continue undisturbed in this case. The full
calculations for this are detailed in Appendix B. Also evident from Table 5-2 are that both the
distributed resistance (DR) and simplified DR approaches achieve the exact solution in terms of
predicted flow velocities. However, only the DR method computes the static pressures correctly. In
the case of the simplified DR approach this is attributed to the cross convective momentum terms being

neglected.

Table 5-2: A comparison of the summary results from the network approach, simplified distributed resistance

approach and distributed resistance approach for a steady state case with inviscid flow.

Details Symbols Units Network | Simplified Distributed
approach DR resistance (DR)

x outlet velocity u, m/s 3 5.000 5.000

y outlet velocity v, m/s 3 1.000 1.000

Internal static pressure Dy Pa -2.22 0.000 2.500

Inlet x static pressure Dw Pa -8 0.000 5.000

Inlet y static pressure Dy Pa 4 0.000 5.000
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To further gain some insights on the modelling approaches, the x and y streams in the model problem
are decoupled as shown in Figure 5-13. These streams are then solved separately using basic fluid
mechanics techniques. This is not far off from what most CFD codes do, but it falls-short because the

two streams are not then linked back to each other. The summary results for this scenario are presented

in Table 5-3.

x direction y direction

[s]
Figure 5-13: Decoupled x and y flow streams.
It is shown on Table 5-3 that the total pressures calculated from each stream are different, i.e.
Do =12.5Pa  p, , =0.5Pa

Thus, if a solution technique which solves for a common total pressure at the common node P were to
be applied, the results from that method will be different from the 1-D exact solution results. Hence,
the difference in the network approach results for such a case presented in Table 5-2, since it solves
for total pressure rather than the static pressure. It is further shown in Table 5-3 that the static pressures

at the common node P are the same and exactly zero. That is
p.p=0Pa p ,=0Pa

This implies that the results from the simplified distributed resistance approach should be the same as
that of the two decoupled streams and that is confirmed by Table 5-2 and Table 5-3. Thus, despite
having the same variation of the governing equations, the solution techniques used for the network
approach and the distributed resistance approach led to different results in this case. This analysis also
brings forth that at the common node P, static pressure is a scalar, yet total pressure is a vector. Thus,

the results of the simplified distributed resistance approach are governed by the directionality of the
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flow which is encapsulated by the physics in the x and y momentum equations. This illustrates the
short-comings of the network approach for such a case, since it solves for total pressure rather than the
static pressure. The inherent 1-D momentum equation of network approach neglects the directionality
of the total pressure at the common node P, thus solving for total pressure as if it is a scalar quantity

at that common node.

Table 5-3: x and y streams results at the nodes from the 1-D exact solutions for inviscid flows.

x direction
Nodes w P E
Details
Units | Symbols | Values | Symbols | Values | Symbols | Values
Velocity m/s Uy, 5 U, 5 Uy, 5
Static pressure Pa Pw 0 DPp 0 PE 0
Total pressure Pa Pow 12.5 Pop 12.5 Por 12.5
y direction
Nodes S | N
Details
Units | Symbols | Values | Symbols | Values | Symbols | Values
Velocity m/s Vg 1 Vp 1 Vy 1
Static pressure Pa Ds 0 Py 0 DPx 0
Total pressure Pa DPos 0.5 Poyp 0.5 Pon 0.5

5.3.3. Resistance to flow

In this dissertation, however flow subjected to resistances due to the presence of tube bundles is
considered. Therefore, now consider the model problem shown in Figure 5-11 which has some
resistance to the flow. For this analysis the Flownex model shown in Figure 5-14 was set-up. Initially,
a small resistance to flow is introduced. The resistance is introduced in Flownex as a friction factor in

all directions and the value is

£=001

Figure 5-14 and the first results column in Table 5-4 show the summary of the results from the Flownex
simulation. Notable in the network approach results column in Table 5-4 is that the x and y outlet

velocities are the same still as was for the inviscid flow case. They just split equally in both directions.
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Pressure 4500 Pa P
)
Static pressure 0.000 Pa
Friction factor 0.010
Pressure 4670 Pa
Static pressure -7.830 Pa Maximum velocity o 3.000 m/s P
£) )
. Maximum velocity 3000ms Y
M I Friction factor 0010 Friction factor 0.010 I
I o " > >
Mass source 5.000 kg/s Total pressure 4545Pa Static pressure 0.000 Pa
Static pressure -2177Pa Pressure 4.500 Pa
. A
Friction factor 0.010
Mass source 1.000 kg/s
—O

Static pressure 4050 Pa
Pressure 4550 Pa

Figure 5-14: Flownex results for the case with low resistance.

Again, for comparison purposes the simplified distributed resistance approach is employed. In this
case, the distributed resistance factors are the same for all the momentum control volumes. In order to
correspond to the friction factor for the network approach, the distributed resistance factor is calculated

to be

B.=B,= 0005 (5.29)
m

Applying the pressure projection methodology, the velocity and static pressure results are summarised
in the middle results column in Table 5-4. Again, the results from the simplified distributed resistance
approach do not correspond to that of the network approach. The results of the simplified distributed
resistance approach demonstrate a flow with directionality whereas the results from the 1-D network
approach does not take directionality into account. For further emphasis on the directionality of the
flow, for the simplified distributed resistance approach the components of directional total pressure in

the common node P are given as
2
+
Poy, =P +%pu2 =0.024 +%x1x(54%j =12.49Pa

2
+
Poy, = p+%pv2 = 0'02“%"1)‘(%) =0.53Pa
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Yet for the network approach, the total pressure is a scalar and as shown in Figure 5-14. It is given as
Po, =4.545Pa

The directionality of the total pressure is due to the dynamic pressure component due to the convective
momentum in the direction of the flow for 1-D cases. Since for the network approach this dynamic
pressure is absorbed in the formulation of the momentum equation using total pressure, it is unable to
manifest the directionality of the flow. Instead, the total pressure at the common node is calculated
using a weighted averaged velocity as illustrated in Appendix B for the inviscid case. For the network
approach, static pressures are obtained through post-processing, hence the reason some negative values

as shown in Figure 5-14.

Again, to illustrate the effect of the additional convective momentum terms on the solution for this
model problem, the distributed resistance approach was employed. The results from the distributed
resistance approach are shown on the far-right column in Table 5-4. As was the case with the inviscid
flow, again for this case only the static pressure values were affected. These static pressures are higher

than those from the simplified distributed resistance approach. The flow patterns are exactly the same.

Table 5-4: A comparison of the summary results from the network approach, simplified distributed resistance

approach and distributed resistance approach for a steady state case with resistance to flow.

Details Symbols | Units Network | Simplified | Distributed
approach DR resistance

x outlet velocity u, m/s 3.00 4.987 4.987

y outlet velocity VY m/s 3.00 1.013 1.013

Internal static pressure Pp Pa 2177 0.025 2.551

Inlet x static pressure Pw Pa -7.830 0.117 5.176

Inlet y static pressure Ds Pa 4.050 0.037 5.056

For further analysis, 1-D exact solutions of the separated x and y streams as shown in Figure 5-13 are
again used to help gain some further insights. From basic fluid mechanics calculations, some results
for the x and y flow streams are shown in Table 5-5 for the nodes. These results are different to those
from the inviscid flow case (Table 5-3) because both the static pressures and the total pressures at the
common node (P) are different. Thus, in this case both techniques that solves for static pressure and

total pressure for the model problem will result in different results to that of the 1-D exact solutions.
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Table 5-5: x and y streams results at the nodes from basic fluid mechanics.

x direction
Nodes w P E
Details
Units | Symbols | Values | Symbols | Values | Symbols | Values
Velocity m/s Uy, 5 U, 5 17 5
Static pressure Pa DPw 2.50 D.p 1.25 Pe 0
Total pressure Pa Pow 15.00 Poxp 13.75 Por 12.5
y direction
Nodes S P N
Details
Units | Symbols | Values | Symbols | Values | Symbols | Values
Velocity m/s Vg 1 Vp 1 Vy 1
Static pressure Pa D 0.10 Py 0.05 Py 0
Total pressure Pa Pos 0.60 Doy 0.55 Pon 0.5

From the results of the 1-D basic solutions on two separate streams, the static pressure values for the

common node P were found not to be the same and they are given as
p.p =1.25Pa Py = 0.05Pa

Thus, this signals that the resulting solutions from the simplified distributed resistance approach will
be different to that of the 1-D separate stream exact solutions. That is evident from the simplified
distributed resistance results shown in Table 5-4 when compared to the results of the 1-D exact
solutions shown in Table 5-5. As practically expected, the simplified distributed resistance approach
shows that some flow from the x direction is diverted to the y direction due to a higher resistance in

the outlet x momentum direction.

The introduction of resistance in the flow paths illustrates some changes to the static pressures at the
common nodes. Thus, stemming from this, next an investigation on the influence of the resistance
factor on the accuracy of the network approach. Flow resistance is added to both directions and by
equal amounts, increasing from zero to five per meter. For this investigation the results of the network
approach were compared to those of the distributed resistance approach. The results are depicted in
Figure 5-15, illustrating that as the distributed resistance factor increases equally in both directions,

the flow distribution percentage error reduces.
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Figure 5-15: The flow distribution percentage error behaviour corresponding to equal variation of the

distributed resistance factor in both directions.

However, the considered heat exchangers in this dissertation do not have equal resistances to flow in
both directions. In this heat exchangers, the highest resistance is due to the tube bundles in the dominant
direction of the flow. Thus, the network approach will present some limitations in modelling the 2-D
or 3-D phenomena of the flue gas side flow of these heat exchangers. Here, the network approach
would depict the flow choosing the path of least resistance which would not conform to real life

observations.

5.4. The applicability of the network approach to heat exchangers in boilers

The insight gained from the study above demonstrated some limitations in using the network approach
for two-dimensional and three-dimensional flows especially those with low resistance to flow. Since,
for this project the network approach is selected as the modelling approach, then going forward the
subchannel approach that was used in chapter 3 will be used to model the radiant final stage superheater
heat exchangers. So, there will be no flue gas side vertical connections to allow for flow mixing. The
flow is assumed to be flowing in channels perpendicular to the tubes. Now, one may ask what are the

practical implications for this?
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In practice the superheater heat exchangers close to the furnace (typically platen types) have large
transversal pitch, which means that the flue gas side looks more like open space with low resistance to
flow. Thus, using the network approach to model the flue gas flow one would not be able to determine
the correct flow patterns. In addition, the flow through these superheaters are typically not
perpendicular to the tubes. So, assuming simple flow perpendicular to the tubes will be erroneous.
However, the correct flow pattern is not very important at all since these heat exchangers are totally
dominated by radiation rather than convection. So correctly accounting for the direct radiation from
the furnace and the gas radiation from the high temperature flue gas in the superheater region is key

for thermal performance determination.

Moving further away from the furnace (towards the final superheater heat exchanger analysed in the
case considered in chapter 6) the transversal pitch becomes smaller and the flow in the convective pass
becomes more perpendicular to the tubes. Therefore, as convection heat transfer becomes more
important and therefore the flow pattern becomes more important, the network approach with flow
specified near perpendicular to the tubes becomes more and more suitable. However, one should
recognise that some adjustment to the convection coefficient correlations may be necessary to slightly
compensate for the flow pattern that might have not been specified correctly (and cannot be predicted
using the network approach). This also brings forth the physical structure factor in the modelling

aspects. Boiler designs and so the heat exchangers in the convective pass vary from station to station.

Therefore, in this work the limitations of network approach encapsulated in Flownex are recognised
based on a fundamental understanding of the simplifications involved. However, from the insight
gained from this study an understanding of the implications of adopting the network approach for the
practical problems modelled here was developed. Thus, it will be used responsibly going forward while

always keeping in mind these limitations.

5.5. Chapter summary

The objective of this chapter was to investigate the applicability of the network approach encapsulated
in the Flownex software in modelling the flue gas side of a superheater heat exchanger in a coal fired
boiler. The study was carried out systematically. It began with a conceptually analyses of most of the
modelling approaches that can be used where the assumptions from one approach to the next were
highlighted. The governing equations of the network approach were compared in-depth to those of the
distributed resistance approach. In the comparison, it was pointed out that the network approach

neglects the convective momentum terms that are not in the direction of the flow since it only focuses
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on one-dimensional flows. Due to the one-dimensional assumption, the convective momentum term is

combined with the static pressure term, allowing for the use of the total pressure in the solution scheme.

To further illuminate some of the limitations of the network approach, a model problem was
considered. The network approach was mainly compared to the distributed resistance approach;
however, some intermediate approaches were used to assist in gaining further insights. For the studied
model problem, the effect of neglecting the convective momentum terms not in the direction of flow
were noticed on the static pressures rather than flow distribution. The major limitation was due to the
solution technique adopted by the network. Due to the inherent assumption of one-dimensional flow
in the network approach, the approach solves for total pressure as a non-directional parameter even at
common nodes in cases of two-dimensional flows. This led to the software being used outside its

inherent computational domain characteristics which is flows in pipes.

After gaining insights on the limitations of the network approach in modelling the open spaces on the
flue gas side of the heat exchangers it was concluded that the philosophy of the subchannel approach
should be used going forward. The subchannel approach philosophy provides better accuracy than
allowing for mixing in the network approach. This implies that flue gas flow computations should be
done in the predominant direction of the flow. Thus, no mixing of the flow in between streams should

be allowed on the network approach model.
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6. REFINED METHODOLOGY: MODEL DESCRIPTION

In Chapters 3 and 4 a network modelling methodology was proposed and demonstrated for convective
superheaters with gas radiation. However, it only accounted for radiation to the heat exchanger tubes
from the gas surrounding it. It did not account for any direct radiation that originated elsewhere (such
as the furnace exit plane or gas contained in a neighbouring heat exchanger) that may pass through the
surrounding gas without being absorbed and then impinge directly on the tubes. This chapter describes
the further refinement of the proposed network methodology to include this direct radiation
phenomenon and presents a practical case study applied to a final stage radiant superheater in a coal-
fired boiler. The information provided include the relevant geometry, heat exchanger arrangements
and the detail description of the Flownex model for the final stage superheater and its process flow

conditions.

High-level models are employed to attain the flue gas conditions and these models are described in
this chapter as well. These high-level models are lumped models or overall performance models of the
systems or subsystems that influence the behaviour of the final stage radiant superheater heat
exchanger. They include the boiler mass and energy (MEB) and parts of the boiler design method

which i1s known as the Gurvich Method.

6.1. Coal-fired boiler geometry and heat exchanger arrangements

The final stage radiant superheater heat exchanger for a 600 MW class coal-fired power plant is
considered. This final superheater is a sub-system in the convective pass of the coal-fired boiler shown
in Figure 6-1. Figure 6-1 shows a schematic of the overall layout of the drum-type two-pass boiler with

the relevant subsystems.

In the arrangement of the heat exchangers in the convective pass, the final superheater of interest in
this work is located in between two heat exchangers as shown in Figure 6-2. Upstream of the final
superheater is the platen superheater and downstream is the secondary/final reheater. The subsystems
upstream and downstream have an influence on the operation of the final superheater. Figure 6-2 also
shows that the flue gas cross flow area reduces as the flow moves towards the back-pass. This cross-

flow area reduction is mostly around the final stage superheater and the final stage reheater.
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Figure 6-1: The overall layout of the boiler including the furnace and the convective pass (Courtesy of

Eskom).
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Figure 6-2: The arrangement of the heat exchangers in the convective pass (Courtesy of Eskom).
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6.1.1. Final superheater heat exchanger geometry

A solid model of the final superheater geometry is shown in Figure 6-3. This superheater consists of
28 tubesheets/elements suspended from the roof of the boiler. Each element has 34 inline U-shaped
tubes connected to the inlet and outlet stub-headers. Each U-shaped tube makes two passes inside the
boiler and consists of three different tube sections in series having the same outer diameter, but with
different wall thicknesses, as highlighted in Figure 6-4 and Table 6-1. The common outer diameter is
44.5 mm while the respective tube wall thicknesses are 6.3, 8.8 and 11 mm. The total height and width

of each element are 15.878 m and 3.921 m, respectively.

Two pipes connect each of the 28-inlet stub-headers and 28-outlet stub-headers to the four inlet and
two outlet main headers, respectively as shown in Figure 6-4. The connecting pipes on the inlet and
the outlet are equally distributed into four legs namely leg A, B, C and D, as illustrated in Figure 6-3.

Pairs of two of these legs connect to each of the outlet main headers.

LegD LegC LegB LegA
| | | |

WAL ERT IR R ERRERAR AR L CLERASE
0000 000 3 b o e o

Left hand side

Right hand side

Rear of boiler

Figure 6-3: The final superheater.
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Figure 6-4: Side elevation (looking from the right-hand side) of the final superheater.

Table 6-1: Geometrical information of each leg in a tube sheet.

Details Units Inlet leg Intermediate leg Outlet leg
(blue) (green) (red)
Outer diameter mm 44.5 44.5 44.5
Thickness mm 6.3 8.8 11
Design temperature °C 527 559 584
Design pressure MPa 19.04 19.04 19.04

6.2. Superheater model development

The model is based on the 1-D network approach introduced in Chapter 3, that involves the solution
of the 1-D mass, energy and momentum conservation equations. In applying this approach, the
physical layout of the tube passes in each element is represented by the layout of the network as shown
in Figure 6-5. To limit the complexity, bundles of tubes may be grouped together and discretised
depending on the level of detail required. The 1-D network approach allows for the mass flow rates
and temperatures to redistribute in the heat exchanger in accordance to the interaction between the

steam flow, flue gas flow and the tube material. In this model, all the relevant modes of heat transfer
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are accounted for. The model is built using specific building blocks which consist of heat exchanger
increments for regions containing tubes, and open flow channels for regions without tubes as shown
in Figure 6-6a and Figure 6-6b, respectively. As concluded in chapter 5, on the flue gas side the model
does not attempt to model the flow patterns due to the limitations of the underlying network approach
itis based on. Thus, the flue gas is modelled as if it is flowing in channels in the predominant direction
of the flow. However, due to the one-directional nature of the steam flow in the tubes, the model

building blocks are connected such that they trace the steam path as is in the real plant.

In this case the 34 parallel tubes per pass in each tubesheet are grouped together in four bundles along
the flue gas flow path. Consider the inlet leg that is modelled using four bundles: the first two of these
bundles each consist of eight tubes and the last two each consist of nine tubes. These tubes are grouped
together with great care, to ensure that tubes with similar inner diameters are bundled together since
this is a requirement for correct calculation of the heat transfer coefficients that depend on the local
flow velocity via the Reynolds number. The total effective tube outer heating surface area for one

tubesheet in the discretized Flownex model is 133.25 m?2.

Note the differences between this final superheater heat exchanger when compared to the primary
superheater heat exchanger that was studied in the previous case study in chapter 4. These differences
allow for the application of a slight variation of the modelling approach when grouping tubes together

and discretization along the height of the final superheater.

In addition to geometrical information, the inputs in this model are inlet temperatures and pressures
for both the flue gas and steam streams, the mass flow rates of these streams, and the direct radiation
heat transfer from the flue gas in the furnace and the platen superheater spaces. Again, in this model,
the mass flow rates were given as negative sources at the outlet, thus effectively drawing the correct
total mass flow through the heat exchanger. This allows the mass flow to automatically distribute along
the flow paths with respect to the thermophysical phenomena occurring in the heat exchanger. Such a
model is capable of analysing maldistribution of mass flow and temperatures within an element as well

as the entire heat exchanger in both steady state and transient operations.
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Figure 6-5: Model layout of one tubesheet/element of the heat exchanger: a) Flownex model imposed on the
physical layout of the final superheater in the plant and b) A stand-alone Flownex model of the superheater to

maximize the details of the model.

As described in Chapter 3 each heat exchanger increment is again made up of a sub-network of nodes
and 1-D flow and heat transfer elements. As shown in Figure 3-9 and Figure 3-12 the different elements
are: the steam flow path; the inside convective heat transfer resistance; the inner scaling layer
conduction heat transfer resistance and associated thermal inertia; two conduction heat transfer
resistances and thermal inertia components that each represent half of the pipe wall thickness; an outer
fouling layer conduction heat transfer resistance and associated thermal inertia; the outside convective
heat transfer resistance; the outside radiative heat transfer resistance; and the flue gas flow path. The
direct radiation into the final superheater is accounted for via a heat addition on the node for the surface

of the outer fouling layer.
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Figure 6-6: The building blocks of the model: a) Heat exchanger increment with two lumped tubes and b)

Flow channel for areas without the presence of tubes.

The modelling methodology also allows for the modelling of the complete heat exchanger by
interconnecting the models of the various elements. The complete heat exchanger model can be
discretised to a level required by the purpose of the model. Here the superheater was discretised into
four sections across the width of the boiler corresponding to the four legs A, B, C and D as explained
in the subsection above. This means that the seven tubesheets contained in each of the four legs are
assumed to be similar and are therefore modelled via one representative tubesheet model that correctly
accounts for the total steam and flue gas flow rates through the whole leg. The schematic representation

of the model is shown in Figure 6-7.

It is highly desirable to demonstrate the applicability of the model based on real plant data. Limited
plant measurements are usually available from the Distributed Control System (DCS) for various
operational conditions. These typically include steam flow rates and temperatures at the inlet and outlet
of each heat exchanger. However, the flue gas mass flow rates and temperatures are rarely measured
directly. An exception is the low temperature economizer heat exchanger at the outlet of the convective
pass where the outlet flue gas temperature is typically measured before it enters the air heater. The
oxygen content is also measured at this point to monitor and control the excess air provided for

combustion.

Since the flue gas flow rate and properties as well as the inlet temperatures and direct radiation are
required to compare the model results with real plant data, separate high-level models must be

employed to obtain such input information. These include developing and verifying a boiler mass and
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energy balance (MEB) that can be used to calculate the flue gas flow rate and properties that are not
directly measured. The mass flow rate of the coal, hot humid air and flue gas calculated from the boiler
MEB are then used in a high-level thermal model of the furnace and the platen superheater. This high-
level thermal model (the Gurvich method) is then used to obtain the inlet flue gas temperature into the
final superheater as well as the lumped direct radiation that impinges onto the final superheater. Finally,
the lumped direct radiation is systematically cascaded over the rows of the heat exchanger. Each row

or group of tubes in rows absorbs some fraction of the direct radiation that impinges on it.
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Figure 6-7: A representation of the Flownex model for a complete superheater modelled with only four

Flownex element models.

6.3. Heat transfer model, pressure drop model and fluid properties

The section presents the various correlations used to describe the flow and heat transfer phenomena in

the model.
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6.3.1. Inside convective heat transfer model

For the inside convection heat transfer coefficient, the Dittus-Boelter’s correlation [13] is used. The

Nusselt number for this correlation is given as

Nu =0.023Re); Pr’* (6.1)

It is valid for fully developed turbulent flow in smooth tubes for fluids [13] with the following range

of conditions, 0.6 < Pr <160, Re,, 210000 and %DZ 10 [123].

6.3.2. Conduction heat transfer

All tubes were assumed to be made of Chromium (low) Steel - (1Cr-0.5Mo). The thermal conductivity

of the outer fouling ash layer was taken from Taler et al. [4] and its value is A, =0.07 W/(m[K).

6.3.3. Outside convective heat transfer model

The outside convection heat transfer is accounted for using the Zukauskas correlation [124] and the

Nusselt number is given as

0.25
Pr
Nu,, = CRe!" Pr" (E} (6.2)

w

where the values of the constants ¢, m and n depend on Reynolds numbers. For an in-line tube

bundle arrangement, 0.7 < Pr <500, 0 <Re,, <2x 10° and for 16 or more tube rows these constants
are shown in Table 6-2. The Prandtl number Pr, is evaluated at the conditions of the tube wall. The

convection heat transfer coefficient is then given by

_ Nuyp A,

- 6.3
2.¢=0 OD ( )
provided the flue gas angle of attack is zero. If the angle of attack is not zero, then the flue gas

convection heat transfer coefficient is given by

h

fep = Egh

=0 6.4)

The correction factor, &, is given by Figure 6-8.
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For the final superheater heat exchanger addressed here, the flue gas flow is assumed to be following

the direction of the mid-plane of the trapezoidal flue gas flow channel. Thus, the flow angle of attack,

¢ 1s determined as illustrated in Figure 6-9. From basic trigonometry

and the angle of attack is given by

tan @ = ™0
X TX
$=90-0

(6.5)

(6.6)

Table 6-2: Constants for the Zukauskus correlations for an inline tube arrangement and 16 or more rows

[124].
Range of the Reynolds number C m n
0-100 0.9 0.4 0.36
100 - 1000 0.52 0.5 0.36
1000 - 2x10° 0.27 0.63 0.36
2x10° - 2x10° 0.033 0.8 0.4
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Figure 6-8: Dependence of heat transfer of banks on the angle of attack [125].

30

10

116



Chapter 6. Refined methodology: Model description

Figure 6-9: An illustration on the characterization of the angle of attack on the flue gas side of the final

superheater heat exchanger.

6.3.4. Radiation heat transfer model

In the model that was presented in chapter 4, only the radiation heat transfer from some gaseous
combustion products were considered, thus simplifying the model. These gaseous products were
carbon dioxide, water vapour and a transparent gas. The transparent gas accounted for the bands where
the two gases were not radiating any energy. However, the radiation from pulverised-coal combustion
products is a more complex heterogeneous system. It involves radiation heat transfer from the gaseous
products and the solid disperse phase. Most of the radiation heat transfer in a boiler takes place in the
furnace with the thermal radiation from the pulverised-coal flame. In the furnace, the radiative transfer
is due to both band emission from CO; and H2O as well as some trace amounts of CO and SO.. This
transfer of energy is coupled with the continuous emission from the different types of particles in the
flame. These particles that are entrained in the combustion gas can be classified as carbonaceous (coal,
char and soot) and inorganic (fly ash). Most of the carbonaceous particles are confined in the firebox
region [126]. The fly ash may be the only particle remaining in the heat absorption region as well as

moving with the gaseous products to the convective pass.

Since this section focuses on modelling radiant heat exchangers in the convective pass of a coal fired
boiler, the radiation of the particle-gas flow mixture only considers the radiation from fly ash particles
in conjunction with the radiation from the gases. This is because fly ash is the dominant type of

particles present in the convective pass, with only traces of the other particles. This work will mainly
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focus on the radiation model used in the Gurvich boiler thermal calculation model [127] which is based
on the former Soviet Union’s standards. However there are other models from literature which include

that of the VDI heat atlas [128, 129] and Blokh [130].

The radiation heat transfer coefficient used in the Gurvich method is given by

o A
h =0, £, (6.7)
2 T,-T,

where 0, T, and T, are the Stefan — Boltzmann constant, average flue gas temperature [K] and

average outer fouled wall temperature [K], respectively. & is the emissivity of the outer fouled wall

and ¢, is the emissivity of the flue gas, which consists of the gases and particles.

The flue gas emissivity is given by

g, =1-exp(-KL,,) (6.8)
where L, and K are the mean beam length and the overall extinction coefficient of the flue gas,
respectively. The mean beam length for the superheater region is given by

Cc

L — rad 69

‘mb 1 1 1 ( )

2l =+ —+—
b h S,

where b, h and S, are the width, height and transverse pitch of the heat exchanger, respectively.
According to Jones [108] in many practical systems the constant C, , is taken as 3.6. The overall
extinction coefficient is given by

K=K +K, (6.10)
where K, is the extinction coefficient of the gas and K, is the extinction coefficient of the fly ash.
The extinction coefficient of the gas is given by

K, =k,r,p (6.11)
where p and r, are the total pressure in the convective pass and volume fraction of the triatomic

gases, respectively. The gas coefficient, k, in 1/ (m[Pa) is given by
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0.78 +1.6r, T
k, =| ———22-0.1{]1-0.37—— [x10”° (6.12)
1000

8
& Lmh

Py
with r, ,, p, and p, the volume fraction of water vapor, the partial pressure of triatomic gas and
the reference pressure, respectively.
The extinction coefficient of the fly ash is given by
K, =k,l,p (6.13)
where k,, and y,, are the fly ash coefficient and concentration, respectively. The fly ash coefficient

is given by
=———£x107 (6.14)

where p, and d,, are the gas density and the diameter of the fly ash particles given in pm,

respectively.
The fly ash concentration is given by

— 0/0 FAxash
Hu =g

8

(6.15)

where % FA is the fraction of the fly ash after combustion, x, is the fuel ash content on an as-

received basis and G, is the flue gas mass excluding the fly ash.

Discretised Flownex model
The default radiation heat transfer element within Flownex solves the standard radiation heat transfer

equation given by
Qi =AF0, (T} ~T}) (6.16)

with A the outer fouled tube surface area and F the view factor. This standard radiative heat transfer

element is used in the current model but with the view factor given by
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F:(‘fw;ljgg 6.17)

Using this view factor together with eq. (6.16) is equivalent to employing the effective radiation heat

transfer coefficient given by eq. (6.7).
Comparison to the VDI heat atlas model

The radiation heat transfer coefficient given by eq. (6.7) as

e +1\ (T'-T!
hr.Gurvitch = 0-0 ( > )gg [ . - J
2 T, -T,

was compared to the radiation heat transfer coefficient given by the VDI heat atlas model [128, 129]

£ T -a.T
By =0, > S 6.18
r.VDI 0[0;, +£W_a c j[ 7—;] _Tw J ( )

g-w

as

where @, is the absorptivity of the flue gas, which consists of the gases and the ash particles.

Using values given in the plant design documents (C-schedule) for a 100% boiler MCR case shown in
Table 6-3, the radiation heat transfer coefficients compared well as shown in Table 6-4. These values
are within 11%. The full calculation for the 100% MCR case from the C-schedule is documented in

Appendix C for the VDI heat atlas model and Appendix E for the Gurvich model.

Table 6-3: Input data from C-schedule for the radiation models at 100% boiler MCR.

Details Units Value
Average flue gas temperature K 1274
Average outer fouled wall temperature K 791
Particle loading kg / m’ 0.046
Outer fouled wall emissivity -- 0.7
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Table 6-4: Comparison between the heat transfer coefficient used in the Gurvich method to that of the VDI

heat atlas for the same temperature values.

Details

Radiation heat transfer coefficient (W/ m'K )

Gurvich method

VDI heat atlas

82.94

91.99

6.3.5. Pressure drop models

The steam side pressure drop was accounted for in the same way as in Section 4.3.5. The pressure drop

on the flue gas side was accounted for using eq. (4.19) which is repeated below for completeness as

2
Ap :Eu'UVTz

where Z is the number of tube rows and Ex is the Euler number [112]. The Euler number can also be

determined using the power law. For inline tube banks with Reynolds numbers ranging between 10*

and 2 x 10° it can be given as

For (b-0.8)/(a-1)21

1.5
Eu :0.52(b 08} Re"
a—1

For (b-0.8)/(a-1)<1

Eu:O.SZ(b _0'8JRer
a—1

where the exponent I is given by

0.5
r:0.12(b_1]
a—1
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In addition, a and » are the pitch-to-diameter ratios or the relative pitches. The relative transverse

pitch a = S%D and the relative longitudinal pitch b= S%D where S, is the transversal pitch and

S, 1s the longitudinal pitch.

6.3.6. Fluid properties

The steam and flue gas properties were accounted for in the same way as in Sections 4.3.7. and 4.3.8.,
respectively in chapter 4. In the calculation of the flue gas density using the ideal gas law, the specific

gas constant was calculated from the universal gas constant (8.314 J/mol [X ) and the molar mass of

the flue gas. This molar mass was calculated based on the flue gas composition sampled from the plant

together with calculations from the boiler Mass and Energy Balance (MEB).

6.4. Mass and Energy Balance (MEB)

In simple terms, the Mass and Energy Balance (MEB) calculates the amount of coal required to
produce the necessary energy to heat water of a specific mass flow rate to the required temperatures at
specific pressures. In general, from the energy balance around the boiler boundary shown in Figure

6-10, the mass flow rate of coal (rhm,) can be calculated as follows

m _ Qinput + Qloss - chedits
coal
Z FRﬂowhﬂaw + CV

coal

(6.22)

Sflows

where, O is the net heat transferred to the Rankine Cycle. This is the total heat transferred into the

input
water/steam circuit via the respective heat exchangers. The steam from this circuit turns the turbines

to produce power. Q, _ is the total boiler heat losses i.e. the radiation losses through the surface. Q

credits
is the combined electrical power into the auxiliaries such as the mills, air heater, primary air fans and
forced draft fan depending on whether they are included in the overall boiler MEB boundary. The

CV,_ , 1s the calorific value of the coal, which is the Higher Heating Value (HHV). FR,  are the mass

coa flow
flow ratios of each specific stream per kg of coal flow. For streams going into the boundary, the mass

flow ratios are positive and negative for streams going out of the boundary. Lastly, 4, represents

flow
the enthalpy of the specific flow streams evaluated at the temperature on the boundary of the overall

MEB [101, 131, 132], as listed on the right in Figure 6-10.
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Figure 6-10: Mass and Energy Balance boundaries.

Implicit in eq. (6.22) is the amount of air required to combust the coal constituents which are usually
referred to as the CHONS. C for carbon, H for hydrogen, O for Oxygen, N for Nitrogen and S for
Sulphur. From stoichiometry, the Theoretical Air Required (TAR) to fully combust the CHONS is

given as

TAR = ﬂZSti L (6.23)
yOZ/air i Mi
Where i =C, H, O, N, S . St, represents the stoichiometric coefficients of O, required to combust
each i — th constituents. M, x and y,,, . represents the molar mass, mass fractions and the mole
fraction of O, in the air, respectively. In order to ensure complete combustion in real boilers an
additional quantity of excess air is added. Thus, the Dry Air Required (DAR) is given as
DAR =TAR(1+ f,) (6.24)

With f,, the percentage of excess air per kg of the TAR. In addition, the air going into the boiler has

to account for the water vapor content in the atmosphere. Thus, the Humid Air Required (HAR) is

given as
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HAR = DAR (1+w) (6.25)

with w the percentage amount of water vapor present in the atmospheric air. The mass flow rate of

the hot humid air into the boiler would therefore be given as

i, = HAR [

coal

(6.26)

If a control volume around the combustion process in the furnace is considered, then from mass balance

the flue gas mass flow ratio (FGR) per mass flow rate of coal is given by

FGR=HAR+1-x_, f;, (6.27)
where x_, is the fraction of ash in the coal and f;, is the percentage of ash that falls to the bottom of
the boiler normally referred to as Bottom Ash. Thus, the flue gas mass flow rate is given by

i, = FGR i (6.28)

coal
6.4.1. Input data to the Mass and Energy Balance

The inputs and assumptions for the overall boiler Mass and Energy Balance are detailed in Table 6-5.
The full names of the abbreviations are as follows: ATT, EC, SH, RH, AH and FW which imply the

attemperator, economizer, superheater, reheater, air heater and feedwater, respectively.

To verify the custom developed boiler MEB, it was applied for a boiler MCR case that is provided in
the plant design C-schedule data. The values of the inputs and assumptions for this boiler MCR case

are shown in Table 6-5.
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Table 6-5: Inputs and assumptions to the MEB for an MCR case given by the C-schedule.

Parameter Units Value
Excess air % 17.36
Ingress air % 1.86
Coal CV (HHV) kJ/kg 16410
Feedwater flow rate kg/s 487.841
Cold reheat flow rate kg/s 467.995
ATT 1 flow rate kg/s 16.826
ATT 2 flow rate kg/s 5.096
ATT 3 flow rate kg/s 10.69
EC water inlet temperature °C 249
SH 3 steam outlet temperature °C 540
RH 1 steam inlet temperature °C 329
RH 2 steam outlet temperature °C 540
ATT 1 spray water temperature °C 249
ATT 2 spray water temperature °C 249
ATT 3 spray water temperature °C 165
EC flue gas outlet temperature °C 334
Ambient air temperature °C 44
AH air outlet temperature °C 288
FW pressure MPa 19.44
RH pressure MPa 4.178
Seal air flow rate kg/s 4.856

6.4.2. Mass and Energy Balance Results

The aim of the overall boiler MEB here is to obtain the mass flow rate of coal, hot humid air and flue
gas. For the boiler MCR case given in the C-schedule, the developed boiler MEB compared well with
the C-schedule data as shown in Table 6-6. This gives confidence in the use of the MEB at different
operating load cases of the real power plant. The full calculation for the MCR case from the C-schedule

is documented in Appendix D.
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Table 6-6: Comparison of the results from the boiler MEB to the C schedule at 100% MCR (Input values were

the same as that from the C schedule).

Details Units C schedule MEB + C-sch. Error (%)
Inputs

Mass flow rate of coal kg/s 98.4 97.6 0.76

Mass flow rate of humid air into the ke/s 637.2 621.9 2 40

furnace

Mass flow rate of flue gas at ke/s 715.8 7284 -1.76

economizer exit

6.5. Thermal Calculations and Energy Balances

To determine the final superheater inlet flue gas temperature and the inlet direct radiation from the
furnace and the platen superheater, an existing boiler design methodology is used. This Gurvich boiler
design method was documented by Zhang et al. [127], Basu et al. [133] and Kakac [134]. It is important
to note that this method was developed to design boilers from design specifications. However, in this
work the method is used to do performance calculations on an existing boiler, thus all the basic sizing
models are not used. In addition, the coal mass flow rate, hot air mass flow rate, flue gas mass flow
rate and the flue gas properties are already available as determined using the MEB that was presented

in the previous section.

6.5.1. Furnace model

The work flow process diagram of the furnace model which is used to calculate the Furnace Exit
Temperature (FET) and the heat absorbed by the furnace is shown in Figure 6-11. The model is centred

around the calculation of the FET, T,, which is determined via trial and error as illustrated in Figure

6-11. It is first guessed and then later calculated using the following equation

Tﬂame
Ty = (6.29)

3 0.6
Oy € Al oo

M 0~ flame™ furnace” ~ furnace

+1

¢ Jfurnace mcoal Cp Jfurnace

where T

flame

is the flame temperature, M is the flame centre modification factor, o is the Stefan-

Boltzmann constant and &

Sfurnace

is the furnace emissivity. A represents the projected area of the

furnace

furnace which absorbs the radiated heat, ¢, is the average thermal efficiency coefficient, ¢, 18

ave
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the heat preservation coefficient, m,,, is the mass flow rate of coal and Cp .. is the mean overall

heat capacity of the combustion products in the furnace.

Guess the Furnace Mass flow rate of Coal CV & Mass flow Geometr
Exit Temperature hot air and enthalpy enthalpy rate of coal Y
b
A 4
> Flame Area of the | Relative height
v < >
; enthalpy furnace of the flame
Furnace exit
enthalpy
A 4 v A 4
Specific heat Flame . = ]
. capacity ) temperature F.ame .centle
2 modification factor
=) <
»| Furnace Exit |«
Temperature B Radiation
properties
No Yes | Furnace exit
enthalpy
FET converged?
Heat absorbed
by the furnace

Figure 6-11: A flow diagram for the calculation of the Furnace Exit Temperature (FET) and the heat
absorbed by the furnace.

The theoretical adiabatic flame temperature, 7,  that results from the combustion of the coal in the

ame

furnace is calculated by performing a Mass and Energy Balance around the combustion process control

volume [101] as shown in Figure 6-12.

n Hame h e

ri.?,m hha — ¢ ?Hcaa." hma.i'

Figure 6-12: Flame energy balance [101].
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From the mass balance around the process control volume, the flame outlet “mass flow” rate can be

given as

i g = 1y + 1t (6.30)

a coal

where m;,, is the mass flow rate of the hot humid air.

The adiabatic flame outlet enthalpy is determined by performing an energy balance around the

combustion process as shown in Figure 6-12. Thus, the flame outlet enthalpy, #,,,, 1s given as
my,hy, +m, (h,, +CV,
hﬂame - ha'"ha c‘o?l ( coal c‘oal) (631)
mﬂame

where £, is the enthalpy of the hot humid air, #_, is the enthalpy of the coal and CV,_, is the calorific

value of the coal. The theoretical adiabatic flame temperature is then determined from the flue gas
properties developed in the boiler MEB.
Using the guessed FET, the furnace exit enthalpy, 4, is also determined. The mean overall heat

capacity of the combustion products, Cp ... 10 per unit mass of coal is obtained by

h -h i1,
Cpﬂtmace = — - ( ’.nfg J (632)
Tﬂame - TFE mcoal

The flame centre modification factor, M which is also referred to as the temperature field coefficient,
accounts for the temperature distribution in the furnace [133]. It is a function of the relative burner

levels as well as the type of fuel burnt. Thus, the flame centre modification factor is given as

M =B-Cx (6.33)

flame

For a suspension-firing boiler burning anthracite and mean pulverized coal, the constants B and C

are 0.56 and 0.5, respectively [127]. The relative flame centre position, x is given by

flame

xﬂame = ‘xh +AX (634)

with Ax an adjustment parameter to account for the burner firing type and the relative burner height,

x, given by

=y
hF

x, (6.35)
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where 5, and h, are the burner height and the height from the dry bottom hopper to the centre of the

furnace exit, respectively.

The radiation heating surface area or projected area of the furnace, A is given by

furnace

Aﬁmzace =X Jurnace (Aﬁmace.toml - Amh) (6.36)

where X is the furnace configuration factor, A represents the entire furnace area as shown

furnace furnace.total

in Figure 6-13 and A, represents the area of doors and holes.

The radiation properties in Figure 6-11 represents the gas inputs needed to account for the radiation in

the furnace. The furnace emissivity, &, 1s given by

€ _ 3 flame
R [ e o

where &, is the furnace flame emissivity. ¢, is the average thermal efficiency coefficient which

is given by
l/Iave = Zslag‘Xfumace (638)

where ¢, ~is the water wall fouling factor.

slag
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Furnace roof

Furnace exit
window

Side wall Front wall Side wall

Back wall

T |~

Dry bottom hopper area

Figure 6-13: Schematic of the projected area of the boiler used for furnace heat absorption calculations.

The furnace flame emissivity is given by

gﬂame = 1 - eXp(_Kﬂame pfurnaceLmb.ﬂtmace) (639)
where p ... 18 the furnace pressure. The furnace mean beam length, L, ... 1S given as
Vfurnace
'mb. furnace = 36 . (640)
furnace.total
with V, .. is the volume of the furnace.
The radiant absorption coefficient of the flame radiation is given as
Kﬂame = kg rg + k_fa/’[fa + kcoxl'XZ (641)

where £, is the radiant absorption coefficient of the gas, r, the volume fraction of triatomic gases, &,
the radiant absorption coefficient of fly ash, 4, the dimensionless fly ash concentration and &, the

radiant absorption coefficient of coke particles. x; and x, are dimensionless numbers determined by

the type of fuel and the firing method, respectively [133].
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In eq. (6.29) the heat preservation coefficient is given by
P jumace =1~ LSS, (6.42)

with Loss. , being the percentage heat loss due to furnace wall radiation and convection.

rad

Once the FET has converged via trial and error, it is then used in conjunction with flue gas properties
to calculate the correct furnace exit enthalpy. Then, the radiative heat absorbed by the furnace in per
unit mass of burnt fuel is obtained through

m

qnfurnace = ¢furnace (hﬂame - hFE )( .

m

coal

fs J (6.43)

The thermal load of radiation heating surface per unit area is then given by

_ mcoalqnﬂtmace
Qr.furnace - (644)

furnace
Figure 6-14 summaries the mass and energy balances around the furnace control volume. Some energy

is lost from the system to the surroundings via the radiation loss, Q, . and the bottom ash, m, h, .

The water walls absorb some of the radiated heat represented by Q

ww *®

’”;g '?IFE Qa‘n'ecr.md

Ql.ossrad _l

bl ha hF: a — ¢ ?Hroaf hcoa."

my,n,,

Figure 6-14: Mass and Energy Balance control volume around the furnace.

Figure 6-14 shows that the heating surfaces after the furnace experiences both the energy of the flue

gas exiting the furnace, m  h,, and the direct radiation, Q, ., ... Which escapes through the furnace

18

exit window.
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6.5.2. Platen superheater model

Effectively the platen superheater receives energy from the combustion via two streams; namely flue
gas energy and direct radiation as illustrated by separating them in Figure 6-15. It is important to
mention that the two streams are separated in this figure for illustration purposes only, in the boiler the
processes are intertwined. Considering the flue gas control volume around the platen superheater, the
flue gas stream exchanges heat with the platen superheater steam, the roof in the platen region and the
water walls in the platen region. These exchanges are predominantly driven by flue gas radiation heat
transfer, however there is also some convection heat transfer taking place. In this model, it is assumed
that all the direct radiation absorbed in the platen superheater regions is absorbed by the platen
superheater steam stream. The unabsorbed direct radiation falls onto the final superheater downstream.
The flue gas stream interacts with the final superheater in two ways. Firstly, since the flue gas in the
platen superheater matrix is at high temperatures, it also radiates heat to the final superheater. Then,

the unabsorbed energy in the flue gas stream also passes through to the final superheater.

Platen superheater flue gas control volume

F. water walls —»| P. water walls >
2
Furnace roof »| Platen roof »| FSH roof
1= = _L _______ I ';
——P: Flue gas stream |
|
e e e e e e e e . .
‘M Final
adiag;, y ; .
Furnace Ion superheater
R - (FSH)
I
——»: Direct radiation e =
N __
v ]
———>|  Platen superheater steam »1 FSH steam
Key > >

Water & steam  Flue gas & Heat absorbed
flow direct radiation  in the platen

Figure 6-15: A process flow diagram for the platen superheater with a control volume around the flue gas

Stream.

The direct radiation absorbed by the platen superheater steam stream is given by

Qr.plat = (q:'.plat.in - q:‘.plat.out ) mcnal (645)
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where ¢, ., and q, ., are the direct radiation from the furnace into the platen superheater and

passing through the platen superheater, respectively. The direct radiation coming into the platen

superheater is given by

D, plat.in.corBrad.plar.

' _ dr.plat.in.cor” “rad.plat.in

r.platin — . (646)
mcoal

with A

vad piatin (D€ 10let radiation area for the platen superheater and ¢, ,,, ., ., the corrected radiative

intensity into the platen superheater. The corrected radiative intensity is given by

q}:platjmcor = ﬂplathAplmAin (647)
where B, is a coefficient accounting for reradiation. ¢, ,,,, is the radiation heat flow into the platen

and it is given by

_ ﬂplatqnfumacemcoal (648)

qr.platjn -

furnace

with 77 ,,, the thermal load distribution coefficient for the platen. ¢, .., and A are the radiative

furnace
heat absorbed by the furnace in per unit mass of burnt fuel and radiation heating surface area of the

furnace, respectively.

In eq. (6.45), the furnace radiation heat leaking out of the platen superheater is given by

I
’ _ qnplal‘in (1 gg‘plat ) ¢pltl[

r.plat.out —
ﬁ plat

the configuration factor from inlet to outlet of the platen. The emissivity of the flue gas in

(6.49)

with ¢

plat

the platen is calculated from

gg.plat = 1 - eXp ( _Kg.platpplathb.plat ) (650)

The radiant absorption coefficient of the flue gas in the platen only accounts for the gas and fly ash as

shown below.

KgAplat = kgj)lat rg + kfap/atﬂfa‘plm (65 1)

and the mean beam length for the platen is given by
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1.8
Lmh.plat = (652)
1 1 1
+ +
bplat hplat ST.plat
with b, h,, and S, the depth, height and average transverse pitch for the platen superheater.

On the flue gas stream, the heat radiated by the gas in the platen superheater matrix onto the final

superheater in per mass of fuel burnt basis is calculated using

4
_ 99 i rsi i T piar . (6.53)

m

coal

is the absorptivity of the gas in the platen,

where 0, is the Stefan-Boltzmann constant, a,

A, ., rsu. 18 the radiation inlet area for the final superheater and ¢, is the correction factor of the fuel.

The average flue gas temperature of the platen is given by

T .. tT
_ T g.plat.in g.plat.out
T(;’.plat.ave - (654)
2
where T, .. is the furnace exit temperature and T, .., 18 the flue gas exit temperature out of the

platen superheater into the final superheater.

The total direct radiation into the final superheater is the sum of eq. (6.49) and eq. (6.53) multiplied by

the mass of fuel burnt resulting in

Qdirect.rad .FSH = (q:' plat.out + qu ) mcnal (65 5 )

In this model, the flue gas exit enthalpy of the platen superheater is calculated via the trial and error

iteration method. The governing equation of the iteration is given by

— mfg hFE - qumcoal - Qc.plat - wa.plat - Qroof.plat
g.plat.out —

(6.56)

mfg

where m  h,, and g,m_, represents the flue gas energy from the furnace into the platen and the heat

fe'°FE

radiated from the flue gas in the platen superheater matrix onto the final superheater, respectively.

Q. ,.; tepresents the amount of heat transferred from the flue gas into the platen superheater steam

stream via convection and gas radiation. Q and Q represent the amount of heat absorbed

ww. plat roof . plat

via convection and gas radiation by the water walls and the roof around the platen superheater,
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respectively. In the model, Q Oy a0d Q. are first guessed, then calculated using

c.plat *
lumped parameter methods in a trial and error iteration scheme together with the flue gas exit

temperature due to the non-linearity of their relationship.

However, if the conditions and mass flow rates of the steam stream in the platen superheater are known,
then the heat transferred from the flue gas stream via convection and gas radiation heat transfer can be

obtained as

chp/at = Qplam‘ut - Q}:plat (657)

where Q and Q, ,, are the heat duty of the platen superheater and the direct radiation absorbed

plat.tot

by the platen, respectively.

If the platen steam stream conditions are unknown, then the lumped parameter Logarithmic Mean
Temperature Difference (LMTD) method can be used to calculate the heat transferred from the flue
gas into the steam stream. The representation of heat transfer calculation using the LMTD method is

given by

Qc.plar = U A AYTLMTDplaU (658)

plat* “g.plat.conv

where U, is the overall heat transfer coefficient of the platen, A is the surface area for heat

g.plat.conv

and AT

transfer consistent with the definition U LMTD. plat

olat is the Logarithmic Mean Temperature

Difference across the platen superheater. The overall heat transfer coefficient is calculated as follows

h

— g.plat
U= 0 : (6.59)
7. plat
1+ 1+ Eash‘plat + h h’g‘plat
c.plat 's. plat
where &, ... h, ,, and E, . representing the flue gas heat transfer coefficient, steam heat transfer

coefficient and ash deposition coefficient, respectively. The term Q, /0. ,, accounts for the

increase in temperature of the ash deposit layer due to the radiant flux from the furnace which in turn

reduces the amount of heat transferred by both convection and radiation on the flue gas side.
The heat transfer coefficient of the steam is given by

Nu? al‘/]SA at
- =—‘I;’) = (6.60)

plat
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with ID , representing the average inner diameter of the tubes in the platen superheater and A,

plat s.plat

the thermal conductivity of the steam. The steam side Nusselt number is

PrO .4
d. plat s. plat s.plat

Nu =0.023C (6.61)

s.plat

and Pr

s.plat

where C Re represents the correction coefficient of tube diameter, the Reynolds

d.plat s.plat

number and the Prandtl number on the steam side.

The flue gas heat transfer coefficient, A in eq. (6.59) is a combination of the convection heat

g.plat

transfer coefficient, and the radiation heat transfer coefficient, £

g.conv.plat

¢.rad pla A0 18 given by

oD,
By o = e, +h

'g. plat g.plat 9 S 'g.conv. plat g.rad. plat
L.plat

(6.62)

{epar» OD,,, and S represents the utilization coefficient of the platen superheater, outer

L.plat
diameter of the platen tubes and average longitudinal pitch of the platen, respectively. In eq. (6.62),

the term in front of the convection heat transfer coefficient, 70D, /25, ,,, accounts for different heat

transfer surface areas used in the correlations.

The flue gas convective heat transfer coefficient is given by

Nl/l . lat/] .plat
thconv.plat = gopl) = (663)
plat
with OD , representing the outer diameter of the tubes in the platen superheater and A the
plat p g p p g.plat

thermal conductivity of the flue gas. The flue gas side Nusselt number is
NugAplat = 0‘2CzAplatCsAplatCwAplatR;?:?)Slat Pr}?j):l;at (664)
where Re, ,, and Pr, , -~ are the Reynolds number and Prandtl number for the flue gas. C_ .
C, . and C, , represents the correction factor for the tube rows, geometric arrangement, and flue

gas composition and temperature, respectively.

In eq. (6.62), the radiation heat transfer coefficient is

]
£ +1 T lat.ave
hg.rad.plat = 0-0 ( = j(gg.platT;.plar.ave = (665)
2 (T, pV
g.plat.ave
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with 0,, € and & representing the Stefan-Boltzmann constant, tube wall fouling emissivity and

g.plat

flue gas emissivity in the platen, respectively. T, is the average flue gas temperature in the platen

.plat.ave

calculated in eq. (6.54). The temperature of the ash deposit layer on the tubes is given by

1 Qr. plat + Qc. plat
A

g.plat.conv

T

w. plat

=T

s.plat

+ E

ash. plat h
s. plat

(6.66)

where T

s.plat

being the average steam temperature in the platen. The term

1 Qr‘ plat + Qc‘ plat
A

g.plat.conv

E

ash. plat h
's. plat

accounts for the increase in temperature of the ash deposit layer.

The heat absorbed by the water walls and the roof in the platen superheater region is calculated using

the overall heat transfer coefficient given by eq. (6.59). Thus, the heat absorbed by the water walls is

given by
wahplar = UplarAwwnplarATww.plat (667)
where A, ., 1s the heat transfer surface area for the water walls around the platen region. The average

temperature difference for heat transfer is simplified to

AT

ww. plat

=T

g.plat.ave

-T

s.ww. plat

(6.68)

with T

s.ww. plat

representing the water temperature of the water wall around the platen superheater region.

The heat absorbed by the roof around the platen region is given by

Qr{)()f = Uplat Arw;f‘platA Tr(mfplat (6 69)

with A representing the heat transfer surface area for the roof in the platen superheater region.

roof . plat

The average temperature difference for heat transfer is given by

AT =T -T (6.70)

roof . plat g.plat.ave s.roof . plat.ave

where T

s.roof . plat.ave

is the average steam temperature within the roof of the platen superheater.
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6.5.3. Final superheater model

The final superheater lumped thermal calculation model is effectively similar to that of the platen
superheater, as schematically shown in Figure 6-16. However, it is important to note that almost all
the direct heat transfer leaking from the platen is absorbed by the final superheater panels. This includes
both the direct radiation leaking out from the furnace and the direct radiation from the gas in the platen

superheater matrix.

On the other hand, if the gas temperature inside the final superheater is high enough, there could also
be direct radiation leaking out at the back of the final superheater. This will be the heat radiated by the
gas in the final superheater matrix onto the final reheater downstream on the flue gas side. However,
in this work this radiation will be neglected since there are no obvious models available of how to
distribute the radiation between the different heat exchanger increments. Only the effect of this

radiation on the outlet flue gas temperature will be addressed at the end.

In calculation of the area around the water walls, there is an additional area due to the water walls

below the final superheater.

Final superheater flue gas control volume

»{ FSH water walls f——>

a2
Platen roof »| FSH roof »| FRH roof
1= = _L ________ - ';
——>: Flue gas stream —
e e ] I
Platen Final reheater
superheater (FRH)
|== = —————— =
|
——»: Direct radiation |
R __
b, f 1
Platen steam > Final superheater steam =

Ke > > >
Y Water & steam  Flue gas & Heat absorbed

flow direct radiation in the FSH

Figure 6-16: A process flow diagram for the final superheater with a control volume around the flue gas

Stream.
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6.5.4. Results from the thermal calculations

The main aim of the high-level thermal calculations of the furnace and platen model was to obtain
inputs to the detailed discretized final superheater model. These inputs are the inlet gas temperature
and direct radiation falling onto the final superheater. For the MCR case given in the C-schedule, the
results are shown Table 6-7. The outlet flue gas temperature was also calculated so that it could be
used to compare with the results from the discretized Flownex model. The full calculation for the MCR

case from the C-schedule is documented in Appendix E.

Table 6-7: Inputs to the final superheater calculated from the Gurvich Method for the MCR case.

Details Units Gurvich Method values
Inlet direct radiation MW 27.078
Gas radiation view factor -- 0.315
Inlet flue gas temperature °C 1059.1
Outlet flue gas temperature °C 942.6

6.6. Direct radiation onto the Final Superheater

The direct radiation that falls onto the tubes of the final superheater is gradually absorbed from the
front tube rows until the back rows. The lumped amount of direct radiation calculated in the subsection
above is therefore cascaded over the tube rows. According to Ganapathy [135], the fraction of energy

absorbed by the first row is given by

2
a (O) =3.14 ODyrgy _ ODpyy sin”! [ODFSH j + [ Sr.rsn j -1 |- Sr.rsn 6.71)
FSH . :
Sresu Srrsn Sr.rsn ODr, ODr,

and the fraction of the energy absorbed by the subsequent rows is given by

()Hz()ﬂ () 61

6.6.1. Direct radiation inputs to the final superheater

As shown in Figure 6-5 the Flownex model of each element/tubesheet in the final superheater was

discretized into four regions per pass. Applying the direct radiation cascading method given by eq.
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(6.71) and eq. (6.72) on the direct radiation given in Table 6-7 for the boiler MCR case, results in the

distribution shown in Table 6-8.

Table 6-8: Amount of direct radiation absorbed by the different rows in the final superheater Flownex model

for the MCR case.

Row numbers Units Amount of radiation
Rows 1 -8 kW 13 200.81
Rows 9 - 16 kW 6 765.18
Rows 17 - 25 kW 3759.21

Rows 26 - 34 kW 1772.09

Rows 35 - 43 kW 835.36

Rows 44 - 52 kW 393.79

Rows 53 - 60 kW 171.2

Rows 61 - 68 kW 87.74

Since the discretized model groups the tube rows in a bundle of eight or nine tubes, then the highest
direct radiation is experienced by the first bundle. From there it reduces, as it cascades toward the back
of the heat exchanger. In a bundle, the direct radiation is assumed to be evenly distributed along the
height of the bundle. Thus, in the Flownex model shown in Figure 6-5 the input direct radiation in each
heat exchanger building block is correlated to the area it represents with respect to the total area along
the height of the bundle. This is dependent on the discretisation employed. As per the description in
chapter 3, each heat exchanger building block is a compound component that consists of a network of
1-D elements as shown Figure 6-17. In this heat exchanger building block, the direct radiation is
introduced as a lumped heat input on the node at the outer fouling layer surface. This direct radiation
increases the temperature of the outer surface of the outer fouling layer. This in turn has an influence
on the flue gas radiation and convection heat transfer. The full calculation of the direction radiation for

the MCR case from the C-schedule are documented in Appendix E.
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Figure 6-17: Schematic of the heat exchanger increment
sub-network in Flownex as shown in Figure 3-12 but with an illustration of the node on which the direct

radiation is introduced.

6.7. Model validation

Disclaimer: The conceptual idea of the model validation methodology was introduced in reference [3] which was published
as part of the proceedings of the 16™ International Heat Transfer Conference. The candidate is indebted to his supervisor
together with whom the concept was developed.

In this chapter, a refined modelling methodology was developed that may help to better understand the
relationship between the operational conditions and metal temperatures encountered in boiler heat
exchangers. To further establish confidence in the results produced with this methodology it should be
validated. However, detailed validation of heat exchanger process models in a real power plant poses
significant challenges. These include the sheer size and complex geometry of the equipment, and the
fact that the relevant steam and flue gas flow rates and temperatures are not always directly measured.
It is also not possible in an industrial setting to simply install additional measurement equipment.

Therefore, the methodologies and results of such validation studies are rarely found in open literature.

Here, a validation methodology is proposed and applied to the developed radiant final superheater

process model. The validation methodology employs a combination of the plant design data (C
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schedules), the boiler Mass and Energy Balance (MEB) and the Gurvich boiler design method to obtain
the various inputs as well as the outputs for comparison purposes. The steam temperatures and
pressures are readily available from the C-schedules. However, some flue gas values are determined

using a combination of the MEB and the Gurvich method.

6.7.1. Model validation methodology

The systematic validation methodology proposed here is based on a combination of C-schedule data,
a boiler MEB and the Gurvich method which are used to obtain the necessary data. This methodology

1s as follows;

Step 1: Run the model in steady state for the 100% MCR (Maximum Continuous Rating) case while
prescribing the same boundary values for the total steam mass flow rate, inlet temperature and pressure
as well as the same flue gas properties, total flue gas mass flow rate and inlet temperature and pressure.
Here, it is assumed that the steam and flue gas temperatures and pressures are uniformly distributed at
the inlets and that the tube surfaces are perfectly clean with no outer fouling, inner scaling or flow

blockages. Compare the calculated total heat exchanger duty with that of the C-schedule.

Since the C-schedule data is usually based on a typical operational scenario where there is already
some degree of plant degradation (typically fouling on the outside of the tubes, etc), it can be expected

that the calculated duty for the clean heat exchanger will be greater than that specified in the C-schedule.

Step 2: Therefore, the model is now calibrated by adjusting the overall heat transfer coefficient (UA)
across the whole heat exchanger until the calculated duty matches that of the C-schedules. The
adjustment of the UA value involves a systematic and careful adjustment of the thickness of the outer

fouling layer.

The rationale behind adjusting the outer fouling layer is that during operation there will be build-up of
ash particles on the outer surface of the tubes. However outer fouling might not be the only effect
which can influences the change in the UA value. During normal operation, the flue gas flow is not
uniform across the inlet cross sectional area of the heat exchangers. Thus, these flow patterns have an
influence on the convection heat transfer process. On the other hand, the heat exchanger of interest is
a radiant superheater, thus an adjustment to the convective heat transfer coefficient might not
necessarily result in a significant change on the UA value. Hence, this effect is assumed to be negligible

in this case.
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Step 3: Run the calibrated model for the other steady state off-design load cases provided in the
C-schedule. Compare the model results to the C-schedule data. A successful outcome in Step 3 without
any re-calibration of the model implies that the calibration that was performed in Step 2 is not simply
a case-by-case “tuning” of the model. Rather, once it was calibrated for 100% MCR the model is
applicable to a broad range of well-defined off-design operational conditions at a given state of plant
degradation that was assumed by the Original Equipment Manufacturer. Therefore, good comparison

for all off-design cases at this stage provides suitable evidence of the validity of the model.

6.7.2. Application of the model validation methodology

This section presents the results of the validation study conducted for the Flownex model of the final stage
superheater heat exchanger of the 600 MW class coal-fired power plant. The single element Flownex
model present in Figure 6-5 was employed. Based on the data provided by the C-schedule, the flue gas
input data including direct radiation was calculated using the boiler MEB and the Gurvich boiler design
method. In this case the heat absorbed by the side and bottom water walls and roof around the

superheater is assumed to be negligible.

Step 1: The model with clean tubes was used to predict the duty of the superheater for the 100% MCR
case under steady state operating conditions. As expected, the model overestimated the duty by 68.2%

when compared to the C-schedule data as shown in the middle column Table 6-9.

Step 2: Upon calibration via the introduction of an outer fouling layer (OFL) of 0.864 mm in thickness,
the predicted duty matched that of the C-schedule as shown in the right-hand column in Table 6-9. It is
important to note that there is some discrepancy in the flue gas outlet temperature which might be due

to flue gas properties used or the zonal influence of the imposed direct radiation.

Step 3: The calibrated model was then used to predict the heat exchanger duty for the off-design 68.6%
MCR load case as shown in Table 6-10. The predicted duty was within 2.1%. This good comparison
provides suitable evidence of the validity of the model results. The resulting outlet steam temperature

was also slightly higher.
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Table 6-9: Comparison of the Flownex (FNX) model with clean and fouled tubes to the data derived from C-

schedule information at MCR.

Details Units C-schedule FNX model - FNX model -
= Clean Fouled
S | Boiler load MCR 100%

& | Outer fouling layer mm - 0 0.864
Heat Exchanger Duty | MW 107 180 107
Mass flow rate kg/s 509.8 509.8 509.8

§ Inlet pressure MPa 17.88 17.88 17.88

S | Inlet temperature °C 473.6 473.6 473.6
Outlet temperature °C 540 591.9 540.1

. | Mass flow rate kg/s 728.4 728.4 728.4

| Inlet pressure kPa 83.5 83.5 83.5

E Inlet temperature °C 1059.1 1059.1 1059.1

= Outlet temperature °C 942.5 828.8 939.5

Table 6-10: Comparison of the Flownex (FNX) model with fouled tubes to the C-schedule data at low load

cases.

_ Details Units C-schedule FNX model

£ | Boiler load MCR 68.6%

g Outer fouling layer mm -- 0.864
Heat Exchanger Duty MW 72.5 74
Mass flow rate kg/s 349.7 349.7

£ | Inlet pressure MPa 16.84 16.84

% Inlet temperature °C 472.1 472.1
Outlet temperature °C 540 542

. | Mass flow rate kg/s 538.8 538.8

| Inlet pressure kPa 83.5 83.5

é Inlet temperature °C 937.0 937.0
Outlet temperature °C 829.0 825.6

6.8. Direct radiation downstream of the final superheater

As mentioned in the subsection above, if the flue gas temperature within the final superheater is high
enough, it could also emit direct radiation towards the downstream final reheater heat exchanger. This
heat rejection via direct radiation will influence the local flue gas temperature and therefore also the

local gas radiation and convective heat transfer within the superheater. In addition, it influences the
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outlet flue gas temperature of the final superheater which in turn becomes the inlet flue gas temperature
for the downstream reheater heat exchanger. In this work, detailed accounting of this direct radiation

is not done. However, its potential influence is considered in the following paragraphs.

Consider the 100% MCR case from the C-schedule data that was discussed above without taking into
account the direct radiation from the final superheater onto the final reheater. The resulting outlet flue
gas temperature was 939.5°C as shown in Table 6-9. If the direct radiation onto the final reheater is
calculated using the same lumped analysis that was employed in the platen superheater, it works out
to be 9.492 MW as presented in Table 6-11. If this direct radiation is also extracted from the calculated
final flue gas temperature from the discretized model, the outlet flue gas temperature reduces to

925.3°C as shown in Table 6-11.

Table 6-11: Summary of the study on the influence the direct radiation from the final superheater onto the
downstream final reheater has on the outlet flue gas temperature of the final superheater for the 100%MCR

C-schedule case on the validated final superheater Flownex model.

Details Units Value
Direct radiation onto the final reheater MW 9.492
Outlet flue gas temperature of the FSH without accounting °C 939.5

for the direct radiation onto the final reheater

Outlet flue gas temperature of the FSH with the direct °C 925.3
radiation onto the final reheater accounted for

The results show that although the potential direct radiation is roughly 9% of the total heat exchanger
duty, the impact of that on the outlet gas temperature is relatively small. However, the convective heat
transfer is driven by the temperature difference between the fluid streams and the gas radiation is driven
by the difference in the temperatures each to the power four. The results shown in Table 6-11 imply
that the convection heat transfer might be overestimated by roughly 1.4% and the gas radiation heat
transfer by 2.5%. Therefore, the impact of neglecting the potential direct radiation at the outlet of the

final superheater should be relatively small.

In the Gurvich model, this direct radiation is calculated as a function of the average flue gas
temperature. Therefore, it does not provide a mechanism for distributing the direct radiation between

the different tube rows as was the case for the incident direct radiation. Thus, a recommendation for
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future work is to apply more refined methods such as the zonal model inside the areas in between the
tubesheets of the platen and final superheaters to account for both the incoming direct radiation and
the outgoing direct radiation. These models should be discretised and linked up to the Flownex heat
exchanger building blocks, again at the outer surface temperature node as was indicated in Figure 6-17.

However, the platform solution speed will be affected in the process.

6.9. Summary of the chapter

This chapter presented a refinement to the developed heat exchanger process modelling methodology.
It systematically provided the information and models necessary in the application of the refined
process modelling methodology on a practical case. This practical case is a final stage radiant
superheater in a coal-fired boiler. The required information included the relevant geometry, heat
exchanger arrangements, Flownex model description and process flow inputs. The inputs on the flue
gas side are not measured in the plant, thus high-level models for attaining the input data into the final
superheater model had to be employed. These high-level models included an overall boiler Mass and
Energy Balance as well as the Gurvich thermal calculation method. The Gurvich method brought to
light the importance of the direct radiation from the furnace and the platen superheater upstream on
the final stage radiant superheater. The refined process modelling methodology is capable of
accounting for this direct radiation. It accounts for it as a heat source on the node of the outer fouling
layer’s outer surface, thus heating the outer fouling layer as it would be the case practically. Hence

influencing the local gas radiation and convection heat transfer on a discretised model.

These high-level models were applied for a boiler 100% MCR case and a 68.6% MCR case provided
in the plant design C-schedule data. For completeness, inputs and results for the 100% MCR case were
shown. The developed boiler MEB was successfully verified against the C-schedule data, thus creating
confidence in its further use on other cases. The developed network model of the final stage radiant
superheater model was also calibrated and validated against the heat transfer duty provided by the C-

schedule. The successful validation of the model created confidence in the use of the model.

Further insight was gained when analysing the impact of accounting for the direct radiation originating
from the flue gas in the final superheater region towards the downstream final reheater heat exchanger.
This analysis showed that the impact will be small and also illuminated that for better accounting of

this direct radiation, refined models such as zonal model must be used.
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/. REFINED METHODOLOGY: MODEL APPLICATION

This chapter presents the application of the refined modelling methodology to the cross-flow radiant
superheater heat exchanger that was described in chapter 6. The emphasis of this chapter is on the
application of the superheater model on a real plant case supported by plant data. From these

applications, the value added by such a model is magnified.

Limited plant measurements are usually available from the Distributed Control System (DCS) for
various operational conditions. These typically include steam flow rates and temperatures at the inlet
and outlet of each heat exchanger. However, the flue gas mass flow rates and temperatures are rarely
measured directly. Since the flue gas flow rate and properties are required inputs for the model, the
boiler MEB is employed to obtain them. To obtain the inlet flue gas temperature and the direct radiation
into the final superheater, the Gurvich method is employed. In this chapter, only the direct radiation

from upstream that impinges on and is absorbed by the final superheater is accounted for.

In addition, for this radiant final stage superheater heat exchanger additional measurement equipment
was installed on the plant. This involved the installation of thermocouples both at the inlet and outlet
of the heat exchanger as detailed in the section below. Some details of the installed thermocouples

were presented by Rousseau et al. [136].

7.1. Description of plant measurements: installed thermocouples

In the plant, thermocouples were installed on the manifold headers and on the stub headers of one of
the elements that is known to experience the highest temperatures. The positions of the installed
thermocouples on the manifold inlet and outlet headers of the final superheater are shown in Figure
7-1. Figure 7-1 clearly illustrates that there are four manifold inlet headers and two manifold outlet
headers. Each manifold inlet header corresponds to one leg of the overall heat exchanger. Each leg
consists of seven elements. So, for each manifold inlet header, one thermocouple was installed on the
connecting pipes to determine the inlet steam temperature. The assumption was that within the header
the steam temperature will be relatively uniform [136]. To be more specific, the thermocouples were
installed on one connecting pipe for elements 6, 11, 18 and 23. On the outlet, the outer legs (Leg A
and D) share a manifold header and the inner legs (Leg B and C) share the other manifold header. For
each of the outlet manifold headers, measurements are taken on every second connecting steam pipe.

This implies that measurements are taken for each of the 28 elements, since each element has two
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connecting pipes which are connected side by side on the manifold headers. It is assumed that each

measurement represents the average outlet steam temperature for each of the 28 elements [136].

Outlet manifolds

Inlet manifolds

Figure 7-1: Positions of the steam pipes coupled to the inlet and outlet manifolds on

which thermocouples have been installed [136].

Due to element 23 usually experiencing the highest temperatures more detailed measurements were

installed on this element, shown in Figure 7-2. Thus, each of the 34 tubes had a thermocouple installed

fiiss

at the outlet near the outlet stub-header as shown in Figure 7-3.

ELEMENT-23
OUTLET STUB-BOX

Figure 7-2: Thermocouples installed on the tubes just before the outlet stub-box of element 23 (Courtesy of
Eskom).
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Figure 7-3: Thermocouples installed on every tube on element 23 just before the outlet stub-box (Courtesy of

Jean-Piere du Preez & Professor Pieter Rousseau,).

Disclaimer: The installation of thermocouples on the final stage superheater heat exchanger was part of Jean-Pierre du
Preez’s Master of Science in Mechanical Engineering Project registered at the University of Cape Town under the Eskom
Specialisation Centre in Energy Efficiency [137]. The installation was funded by Eskom Research, Testing and
Development (RT&D) under the leadership of Dr Mark Newby. The acquisition of data both from the DCS and the installed
thermocouples was part of Lethukuthula Vilakazi’s Master of Technology in Mechanical Engineering Project registered

at the Vaal University of Technology under the Eskom Specialisation Centre in Energy Efficiency [138].

7.2. Input data

In addition to the installed thermocouple data, DCS data had to be obtained for the final superheater
heat exchanger and the high-level models (the boiler MEB and the Gurvich method model) as

discussed in the sections above. This DCS data is presented in Appendix F.
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7.3. Model application: Steady state

In this section, the discretized superheater model developed and validated in the previous chapter is
applied to real plant scenarios for steady state operations. Cases for both uniform and non-uniform flue

gas and steam inlet conditions across the width of the superheater are analysed.

Furthermore, results of a case study at the plant full load are presented. In the acquisition of steady
state real plant operation data, it is important to identify a time period during which the plant was
operating at near steady state, as highlighted by Rousseau et al. [136]. In most instances, the plant
hardly operates at true steady state, but achieve quasi-steady state. In the acquired plant data, a quasi-
steady state period of full load plant operation was observed between 17:30 and 22:40 for the selected
day, as illustrated by Figure 7-4 for the generator load. Therefore, all inputs used for the full load

steady state case are based on the mean values of the respective inputs obtained within this period.
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Figure 7-4: Recorded generator load data from the plant DCS, showing a full load quasi-steady state from
around 17:30 until 22:40 [136].

For the discretized model, inputs for both the steam and flue gas sides are required. For the case
modelled here, the input data is shown in Table 7-1. Inputs for the steam side are obtained from the
recorded DCS real plant data. For the flue gas side, the high-level models, i.e. the boiler MEB and the
Gurvich method, were employed to obtain the inputs. These are lumped input values, thus relevant

methods of distributing them into the discretized Flownex superheater model are employed. As
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illustrated in the previous chapter, the direct radiation is cascaded over the heat exchanger model using
eq. (6.71) and (6.72). For each lumped bundle of tubes in the Flownex model, the direct radiation is
assumed to be uniformly distributed along the length of the tubes. The input to the high-level models
are presented in Appendix F. These inputs were used in the high-level models similar to those presented

in Appendix D and E for the 100% MCR C-schedule case.

Table 7-1: Inputs to the final superheater from the DCS and resulting from the high level MEB and Gurvich

model calculations for the full load real plant case.

Details Units Value
Mass flow rate of steam into the final superheater kg/s 470.6
Inlet steam temperature °C 485.9
Mass flow rate of flue gas out of the economizer kg/s 839.9
Inlet flue gas temperature °C 1030.9
Direct radiation into the final superheater MW 22.297
Gas radiation view factor for the final superheater -- 0.259

7.3.1. Uniform inlet conditions across the superheater width

If it is assumed that all 28 elements of the superheater heat exchanger experience the same steam and
flue gas inlet conditions as well as the same fouling conditions, then the one element model for the
superheater shown in Figure 6-5 can be used as a representative model for the whole heat exchanger.
This is the model that was calibrated and validated against plant design C-schedule data in the previous

chapter.

Firstly, the model had to be run in steady state for the full load case while prescribing the same
boundary values for the total steam mass flow rate, inlet temperature and pressure as well as the same
flue gas properties, total flue gas mass flow rate and inlet temperature and pressure as those obtained
from the DCS data together with the data generated via the boiler MEB and Gurvich method. The
calculated total heat exchanger duty was then compared with that of the real plant data. Since the real
plant is likely to be operating with a different degree of fouling than the C-schedule for which the
model was calibrated earlier, it can be expected that the calculated duty will be different from the
measured value. The results presented in Table 7-2 confirm this. Therefore, the model was then re-
calibrated by further adjusting the outer fouling layer thickness uniformly across the whole heat

exchanger until the calculated duty matched that of the real plant data as illustrated in Table 7-2. This
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was achieved when the outer fouling layer thickness was 1.65mm. This adjustment is based on the
rationale discussed in the previous chapter. In addition, there was a satisfactory comparison of the
outlet lumped steam and lumped flue gas temperatures from the discretised model to those from the

DCS, MEB and Gurvich method.

Table 7-2: Comparison of the results from the discretized Flownex (FNX) model which has been calibrated
against the MCR of the C-schedule (left) and the re-calibrated model against the plant full load at the date

and time selected in this calculation.

Details Units | DCS, MEB & FNX model - FNX model — Re-
Gurvich Calibrated Calibrated for real
T;E against C- plant data
2 schedule
¢ | Plant load Full 100%
Outer fouling layer mm -- 0.864 1.65
Heat Exchanger Duty | MW 67.5 95.7 67.4
Mass flow rate kg/s 470.6 470.6 470.6
§ Inlet pressure MPa 16.3 16.3 16.3
g}; Inlet temperature °C 485.9 485.9 485.9
Outlet temperature °C 533.5
. | Mass flow rate kg/s 839.9 839.9 839.9
£ | Inlet pressure kPa 83.5 83.5 83.5
E Inlet temperature °C 1030.9 1030.9 1030.9
= Outlet temperature °C 975.9

The results of the calibrated discretized model of the superheater can then be used to study the
temperatures and heat transfer trends in locations of the heat exchanger where a measuring instrument
cannot be easily installed. Figure 7-5 shows some 2-D temperature profiles on a section of heat
exchanger just above the U-shape. This section of interest is illustrated by the rectangular shape
imposed on the superheater model on Figure 7-5a). Figure 7-5b) shows a relatively uniform variation
of the flue gas temperature across the depth of the heat exchanger section. This is because the inlet flue
gas temperature is assumed to be uniform across the cross-section of the inlet to the superheater. Figure
7-5c) presents the temperature profile of the outer surface metal temperatures. As expected, the
temperatures are lower at the steam inlet side than the steam outlet side. Also, on the steam outlet there
18 a variation since the inner tubes are shorter than the outer tubes. Thus, the steam in the inner tubes
is subject to less heat transfer than that on the outer tubes because of the difference in the heat transfer

area.
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Figure 7-5: Temperature results along the depth of a section of the superheater element; a) the 2D section of

interest, b) flue gas temperature profile and c) outer surface metal temperature profile.

A more detailed and comparative analyses of the temperature and heat transfer distribution is presented
in Figure 7-6. The focus is on a cut away section corresponding to the discretisation of the model as
illustrated in Figure 7-6a). This is an interesting section because it consists of both the inlet and outlet
steam sides. Figure 7-6b) and Figure 7-6¢) respectively present the temperature and heat transfer
distribution for this discretised section. For each of the heat exchanger building blocks, the recorded

steam and flue gas temperature values represent an average between the inlet and the outlet.

In Figure 7-6b), the inlet steam temperature is uniform since it was specified in the model and the
outlet steam temperatures are slightly higher on the flue gas outlet side. The trend of the outlet steam
temperature conforms to the expectation as per the geometry of the heat exchanger. The outer tubes
are longer, thus have a higher heat transfer area than the inner tubes. Therefore, these outer tubes
absorb more heat resulting in a higher outlet steam temperature. Due to the high thermal conductivity
of the tubes, the tube outer surface temperatures are close to the corresponding steam temperatures. As
explained in chapter 4, the outer fouling layer temperature is higher than that of the tube surface due
to its low thermal conductivity. However, the relationship between the temperatures of the flue gas

and outer fouling layer for this heat exchanger differs to that analysed in chapter 4. This is because the

University of Cape Town 153



Chapter 7. Refined methodology: Model application

final superheater analysed here is a radiant heat exchanger, while in chapter 4 a primary superheater

which is a convective heat exchanger was analysed.
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Figure 7-6: A detailed analysis of the inlet and outlet of the heat exchanger with; a) the location of the region

of focus on the heat exchanger, b) the temperature distribution and c) the heat transfer distribution.

In addition to the gas radiation, direct radiation must also be accounted for in a radiant superheater as
shown in Figure 7-6¢). Due to the location of such heat exchangers, the luminous radiation from the
furnace also leaks out and reaches these heat exchangers. Also, the long-range direct radiation from
the flue gas in the platen superheater region upstream does reach the final superheater. As a result of
the direct radiation heat impinging on the first bundle close to the furnace, the temperature of the outer
fouling layer is almost the same as that of the flue gas, as shown in Figure 7-6b). This phenomenon
corresponds to a relatively small amount of heat being convected and radiated via gas radiation from
the flue gas stream onto the outer fouling layer as illustrated in Figure 7-6¢). Despite this phenomenon
occurring, there is still a significant amount of heat conducted into the steam flow, as indicated by the
data points named “OFL Conduction”. Thus, for the first bundle almost all the heat transferred to the
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steam is a result of the direct radiation. As the direct radiation gets absorbed by the tubes, its
contribution decreases as the flow cascades towards the back of the superheater. This corresponds to
an increase in the contribution of gas radiation and convection heat transfer as shown in Figure 7-6¢),

supported by the flue gas and outer fouling temperature trends shown in Figure 7-6b).

Furthermore, the direct radiation heat impinging on the first bundle close to the furnace can result in a
case where the temperature of the outer fouling layer is higher than that of the flue gas. This
phenomenon could then correspond to heat being convected and radiated via gas radiation away from
the outer fouling layer into the flue gas stream. However, despite such a phenomenon occurring, the
net heat would still be conducted into the steam flow. Thus, for the first bundle all the heat transferred

to the steam would be a result of the direct radiation.

7.3.2. Non-uniform inlet conditions across the superheater width

Here, the focus is on modelling the complete superheater heat exchanger to allow for the variation of
parameters and properties in three-dimensions i.e. width, depth and height. The discretized model
shown in Figure 6-7 is used. This model bundles together the seven elements in each leg. In this model,

the energy absorbed by the side water walls, bottom water walls and roof is assumed to be negligible.

As alluded to in the sections above, the input from the high-level MEB and Gurvich model lumps the
flue gas energy at the inlet face of the superheater. However, it is well accepted that in reality the input
energy at the inlet face of the superheater is not uniformly distributed, as illustrated by Babcock &
Wilcox [12]. For the boiler considered in this, a CFD model which included the furnace, platen
superheater and final superheater was developed separately by Dr Ryno Laubscher of the University
of Cape Town. The CFD model focused on the flue gas side without solving the detailed steam flows
inside the tubes. The results from this CFD model of the boiler confirmed that the inlet flue gas
temperature to the final superheater varies across the face of the inlet cross-section. When the CFD
model was run with inputs corresponding to the 100% boiler MCR case C-schedule data, it resulted in
the normalized inlet flue gas temperature profile presented in Figure 7-7. In the CFD model, the boiler
was assumed to be symmetric, hence half of the boiler was modelled. Notable is that the temperature
is highest at the centre location across the width but below the midpoint along the height. In addition,
the temperature reduces closer to the side water walls and the roof since some energy is absorbed there.
Furthermore, due to the conservation of momentum the flow profile has an impact on where the hot
spots are in the boiler. In this work the normalized inlet flue gas temperature profile shown Figure 7-7

is used to proportion the lumped input energy from the high-level Gurvich model. The resulting
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temperature distribution is then used as an input to the discretized superheater model. This energy

distribution is only done for the inlet flue gas temperature, not for the direct radiation.

Disclaimer: The CFD model for the boiler analysed in this work from which the inlet flue gas profile was extracted was
developed by Dr Ryno Laubscher based on geometrical information and plant design C-schedule data. This is part of the
wider research work done in the Eskom Specialisation Centre in Energy Efficiency under the Department of Mechanical

Engineering at the University of Cape Town.
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Figure 7-7: Normalized flue gas temperature plot at the inlet face of the final superheater from a CFD model

with a symmetry assumption at the mid-plane of the heat exchanger (Courtesy of Dr Ryno Laubscher).

Applying the steam mass flow rate and averaged flue gas side input values presented in Table 7-1, but
while accounting for the variation of the inlet steam temperature on each leg, the outlet steam
temperatures shown in Figure 7-8 were obtained from the discretised model. The model results are
compared to the results from the detailed thermocouple measurements shown in Figure 7-1. For this
analysis, the inlet steam temperatures are assumed to correspond with the measurements from the
thermocouples on the inlet headers. The rationale behind the assumption is that the thermal
conductivity of the metal is high, thus the difference between the steam and metal temperatures should
be negligible. In addition, it was assumed that the measured inlet temperatures are a representation of

the steam temperatures in the inlet headers for each leg.

At first it was assumed that the outer fouling layer was uniform across the heat exchanger as was the

case when uniform inlet conditions were imposed. Since now the inlet thermocouples temperatures
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were to be used as inputs, the model used for the case with uniform inlet conditions was recalibrated.
Using the average inlet temperature of 478.9°C, calibrating against an average outlet temperature of
536.9°C, a satisfactory comparison was achieved when an OFL of thickness 1.2mm was imposed. The

corresponding outlet steam temperature from the discretized model was 536.7°C.
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Figure 7-8: Comparison of the outlet steam temperature from the discretized model to the detailed
thermocouple data with the inlet tube metal temperatures from the thermocouples used as inlet steam
temperature inputs for the discretized model. For this discretized model all the legs were made up of the

recalibrated model from the uniform inlet condition section above.

Hence the assumption that each leg experiences an OFL thickness of 1.2mm was employed. The
resulting outlet steam temperatures from the discretised model are shown in Figure 7-8. The
temperature results from the discretized model for Legs A, B and D do not correspond to those from
the averages per leg from the thermocouples for those legs. With those from Legs A and B slightly
lower and that from Leg D slightly higher. However, the results for Leg C are almost the same. The
average of all the measured temperatures from the thermocouples at the outlet is 536.9°C, while that
from the discretized model is 531.3°C. Thus, despite imposing the same OFL across all the legs on the
discretized model the average outlet temperature is relatively close to the average from measurements.
The slight discrepancy of about 1% can be attributed to the imposed profile on the flue gas side. This
implies that the inlet energy that was provided by the uniform profile was slightly higher than the
energy imposed in this case study where a non-uniform profile is imposed. Here, the inlet flue gas

temperature corresponds to the CFD normalized profile which conforms to reality. Even though the
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average outlet steam temperature is slightly lower, the trend of the temperatures still depicted a profile

which is related to the inlet steam temperature profile.

Although the calculated outlet steam temperatures from the discretised model compared reasonably
well with the measurements in the analysis above, it must be noted that an OFL of uniform thickness
was still assumed for all the tubes in the representative elements for all the legs. In practice however,
different OFL thicknesses will occur and therefore a separate recalibration of each leg is required to
fine-tune the discretised model. After recalibration of each leg, the steam temperature results from the
discretised model compared well to the average temperature results from the thermocouple
measurements for each leg as shown in Figure 7-9. Such results were achieved with the following OFL
thicknesses: Leg A — 0.85mm, Leg B — 0.65mm, Leg C — 1.2mm and Leg D — 1.55mm. With such
satisfactory results, the respective temperatures of the lumped elements per leg can be analysed. This

analysis can be done in areas of the heat exchanger where measuring equipment cannot easily be

installed.
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Figure 7-9: A comparison of the results from the discretized model to the detail measurements from the

custom installed thermocouples after recalibration of the model for each leg.

However, note that due to the discretisation of four lumped legs across the width of the boiler, it is not
possible to fully capture the “M-type” profile shown by the detailed outlet metal temperature
measurements. The modelling methodology does allow for a more finely discretised complete

superheater model to simulate the “M-type” outlet steam temperature profile as illustrated in chapter
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4. It should be mentioned that with the level of discretisation along the height of an element, such a
model would not be very quick to run. Thus, somehow eliminating some of the benefits of the 1-D
network approach modelling. The building of such a finely discretised model using the developed
element models of the final superheater can be done as future work for illustration purposes, thus it is

deemed to be outside the scope of this work.

7.3.3. Element 23 — Non-uniform inlet conditions

For further validation of the discretised superheater model, element 23 is analysed. This element had
thermocouples installed at the inlet and outlet stub headers as detailed in the sections above and

illustrated in Figure 7-3. Element 23 is part of leg D in the complete heat exchanger arrangement.

As alluded above, the detailed CFD model calculations showed that the flue gas temperature is not
uniform at the inlet cross-section of the final superheater but has a non-uniform profile as illustrated
in Figure 7-7. Thus, for the model of element 23 as well, an inlet flue gas temperature profile has to be
imposed at the inlet boundary. This inlet profile has to account for the variation of temperature along
the height of the element at the inlet as illustrated in Figure 7-10a). Figure 7-10a) represents the
normalised inlet flue gas temperature along the height of the inlet cross section as extracted from the
CFD model results around the location of element 23. The profile illustrates that closer to the roof
there are lower temperatures because some of the energy there is absorbed by the roof. There is some
significant variation along the height of the element which corresponds with the flue gas flow profile
as determine by the momentum conservation. The weighted normalized inlet flue gas temperature
profile that corresponds to the discretisation of the Flownex model is shown in Figure 7-10b). The inlet
flue gas averaged temperature of 1030.9°C calculated via the Gurvich method and presented in Table
7-1 is imposed together with the normalized inlet temperature profile shown in Figure 7-10b) to obtain
the inlet flue gas temperature profile for the discretized model of element 23. This is done such that
the resulting average inlet flue gas temperature is the same as the lumped temperature from the Gurvich

method.
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Figure 7-10: The normalized inlet flue gas temperature distribution of the final superheater along the height

weighted with respect to the flow across the width of the inlet ducting before the final superheater based on

detailed CFD results; a) shows the profile as extracted from the CFD and b) shows the profile that

corresponds to the discretisation of the discretized Flownex model (Courtesy of Dr Ryno Laubscher).

On the steam side, input data was obtained from the DCS for leg D of the final superheater. The inlet

steam temperature is presented in Table 7-3, together with the outlet steam temperature which was

extracted to compare with the model results.

Table 7-3: Inlet and outlet steam temperatures from the DCS for leg D and the weighted averaged outlet

steam temperature from the Flownex model of element 23.

Details Value [°C]
Inlet steam temperature - DCS 494.9
Outlet steam temperature - DCS 535.7
Outlet steam temperature — Flownex model 535.5

The relevant input data from Table 7-1 and Table 7-3 was applied on the calibrated model of the final

superheater heat exchanger element. After calibrating the outer fouling layer thickness to 2.05mm, the

resulting steam temperature corresponded to the DCS data as illustrated in Table 7-3. The outlet metal
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temperature results of the calibrated model were further compared to the values obtained from the
custom installed thermocouples as illustrated in Figure 7-11. The overall trend of the tube outer surface
temperature predicted by the discretized model corresponds to the trend of the readings from the
thermocouples. The temperatures are slightly lower on the inner side compared to the outer side. In

addition, the comparison of the discretized model results to the average values from the thermocouple

recordings was satisfactory as well.

Again, 2-D profiles of the flue gas and outer tube surface temperatures for a section of element 23
above the U-shape area were plotted as shown in Figure 7-12. Figure 7-12b) illustrates the effects of
the imposed inlet flue gas temperature profile. Figure 7-12c) shows the metal temperatures at the outer
surface of the tubes. Despite having a flue gas profile imposed on it, the metal temperatures conform
to the expected profile of lower temperatures at the inlet and higher temperatures at the steam outlet

corresponding to the steam temperatures.
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Figure 7-11: Comparison of the outlet metal temperature of the Flownex model results to the average

thermocouple measurements from the plant for element 23 at plant full load.
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Figure 7-12: Temperature results along the depth of a section of element 23; a) the 2D section of interest, b)

flue gas temperature profile and c) outer surface metal temperature profile.

7.4. Model application: Transient

This section illustrates the applicability of the refined heat exchanger modelling methodology to
simulate the behaviour of a heat exchanger in response to an overall transient event of the plant. This
is built on the transient scenarios that were analysed in chapter 4 where the ability of the models to
simulate transients was introduced. However, in chapter 4, the focus was on behaviour analysis of the
model in response to case specific disturbances introduced for sensitivity analysis purposes. Here, the
focus is on the behaviour of the refined model of the final stage radiant superheater in simulating a
transient event that took place in the real plant, in line with the purpose of having a model suitable for
online condition monitoring. The selected real plant transient event was due to a reduction in plant
load from full load at 22:40 to about 65% full load at 23:30 on the selected day as illustrated in Figure

7-13. So, the plant undergone this dynamic operation for a period of 50 minutes.
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Figure 7-13: Generator load illustrating a transient event where the load drop from full load to about 65%

full load between 22:40 and 23:30 on the selected days.

The transient event was sandwiched between two quasi-steady state operations; at full load from 17:30
to 22:40 (case that was analysed under the steady state section above) and at 65% full load from 23:30
to 3:10 the next day. So, the questions addressed in this section are: How did the final superheater
behave during this 50-minute transient period, and was it different from the steady state operations and

can the final superheater discretized model capture this behaviour?

As mentioned above, in the plant it is very rare to measure temperatures and mass flow rates on the
flue gas side, hence the development of the high-level boiler MEB and Gurvich method for obtaining
the flue gas side input data. However, these models are limited to quasi-steady state cases due to the
cumbersome nature of the calculations and the modelling platforms on which they were developed.
Thus, for the selected transient event the high-level models were employed for the two steady state
cases that sandwiches it and the resulting values were presented on Table 7-4. For each steady state
case, mean values over the duration of that particular operation state were used as inputs in the high-
level models. The input data used is presented in Appendix F. In the final superheater discretized model
for this transient case, the direct radiation and the view factor were assumed to be average values
covering both the steady state cases and during the transient event. This assumption can be adapted
later to account for the variation in the model if need be. For the flue gas mass flow and inlet

temperature, it was assumed that they vary linearly between the two steady state cases.
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Details Units Value Value
Generator load Full load 100% 65%
Mass flow rate of steam into the final ke/s 470.6 3023
superheater

Inlet steam temperature °C 485.9 461.5
Mass ﬂqw rate of flue gas out of the kels 839 9 5317
economizer

Inlet flue gas temperature °C 1030.9 919.9
Direct radiation into the final MW 22997 16.503
superheater

Radiation view factor for the final _ 0259 0.971

superheater

Corresponding to the reduction in the generator load for the transient event of interest as recorded by

the DCS, the steam mass flow rate out of the final superheater to the high-speed turbine also reduced

as illustrated in Figure 7-14. This data was obtained from the DCS on a ten-minute interval basis. As

per the assumption adopted for this case, the flue gas mass flow rate for the transient event varied

linearly between the two steady state events as shown in Figure 7-14. These mass flow rates were used

as inputs in the final superheater discretized model.
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Figure 7-14: Flue gas and steam mass flow rates which corresponds to the generator load between 17:30 and

3:10 on the selected days.

Again, for simplicity it is assumed that all 28 elements of the superheater heat exchanger experience
the same steam and flue gas inlet conditions across the width of the heat exchanger as well as the same
fouling conditions. Therefore, the one element model for the superheater shown in Figure 6-5 can be
used as a representative model for the whole heat exchanger. The input steam temperature is an average
of the values from the four legs. The inlet flue gas temperature is assumed to be uniform across the
width of the inlet cross-section of the heat exchanger but not along the height. This allows for the
adaption of the flue gas temperature profile as illustrated in the modelling of the steady state case for

element 23 in the section above.

By again extracting data from the detailed CFD model results, the inlet flue gas temperature profile
illustrated in Figure 7-15a) was determined. Figure 7-15a) represents the normalised weighted average
inlet flue gas temperature along the height of the inlet cross section. This profile is a weighted average
with respect to the flow distribution across the width of the duct at every elevation. The normalised
weighted inlet flue gas temperature profile that corresponds to the discretisation of the Flownex model
is shown in Figure 7-15b). As alluded in the previous sections, the input inlet flue gas temperatures,
shown in Table 7-4, obtained from the high-level Gurvich method are lumped values across the inlet
cross section of the superheater. These lumped values are then imposed on the profile for the

discretized model for each of the steady state cases (100% and 65% full load) as was the case for the

University of Cape Town 165



Chapter 7. Refined methodology: Model application

steady state model for element 23. For each of the inlet flue gas boundary condition, a linear

distribution of the inlet flue gas temperature assumption is imposed.
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Figure 7-15: The normalized inlet flue gas temperature distribution of the final superheater along the height
weighted with respect to the flow across the width of the inlet ducting before the final superheater which is
based on detailed CFD results; a) shows the profile as extracted from the CFD and b) shows the profile that

corresponds to the discretisation of the Flownex model (Courtesy of Dr Ryno Laubscher).

To illustrate the applicability of the final superheater discretized model for online condition
monitoring, it was run for the sets of input values from 17:30 to 3:10 the next day. The model was first
calibrated using the full load steady state data to account for the difference in the inlet flue gas
temperature profiles and radiation characteristics when compared to the homogenous case that was

modelled above.

However, before embarking on detailed analysis of the transient event it is prudent to ensure that a
time-step independent solution is obtained. It is therefore necessary to start off with a study to
investigate this over the computational time domain. From the DCS, the input data is reported in 10
minutes intervals, thus the maximum allowable time-step size in order to avoid missing important
details is 10 minutes. For ease of reporting and comparison, time-step sizes which are mathematical
factors of 10 minutes are considered. A C sharp script that was developed by Willie Le Grange [139]

was used to import the DCS 10 minutes intervallic data in to the boundary conditions of the Flownex
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models. This script allows for linear interpolation on the data points to get values at a desired point in

time.

Figure 7-16 demonstrates the time-step size dependence of the results obtained from the model. It is
presented in terms of the normalised error squared results for various time-step sizes. The error squared
for each time-step size was determined by summing the square of the difference between the model
results and measured outlet steam temperatures at 10 minutes intervals. The resulting errors squared
were then normalized using the maximum error squared. The nearly corresponding values for the
smaller time steps of less than 5 minutes, indicate that a further reduction in the time-step size will not
results in a variation in the results, and therefore time-step independence is obtained. For further

analyses in this work, the one-minute time step size was selected.
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Figure 7-16:Normalized sum of error squared results from the transient model time step dependency analysis

over the computational time domain.

Figure 7-17 presents steam temperature trends from the DCS as well as the discretized model. The
inlet steam temperature trend illustrates a noticeable variation during the transient event between 22:40
and 23:30 and this was provided as an input to the model. A satisfactory comparison was achieved
between the resultant outlet steam temperature from the model and the outlet steam temperature
extracted from the DCS data. Overall the outlet steam temperature remains relatively constant
regardless of the transient event. This conforms to the requirements of the operations as controlled by

the operators using attemperators and other techniques during operations. The discretized model
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predicted satisfactory results without any physical adjustments during the simulation of the transient
event on the selected days. The results show that the model analysis provides an acceptable
representation of the real-plant behaviour during transient events where there are complex interactions
between various thermofluid processes while multiple parameters are changing at the same time. These
parameters are the steam and flue gas flow rates, as well as the inlet steam and flue gas temperatures.
The discretized model can therefore be used (after some calibration to account for the level of fouling)

to study the impact of real-world operational scenarios in support of power plant flexibility and low

load operation.
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Figure 7-17: Inlet and outlet steam temperatures from both measurements and results from the Flownex

model simulations for the selected days.

Figure 7-18 illustrates some of the potential advantages of having a discretised model since it shows
some detail results of the metal temperatures on the outlet steam side as per the discretisation. It shows
the temperature variation along the depth of the superheater highlighting the hottest building block
rather than an average temperature as it would be in a lumped model. The variation of the metal
temperatures at the outlet shows a difference of about 25°C between the hottest and the coldest bundle.
This provides a better understanding of the phenomena taking place in the heat exchanger.
Measurements to such details are usually not available in most coal-fired boilers, thus illustrating a

potential benefit of the discretized model.
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The insights from Figure 7-17 and Figure 7-18 illustrated that the model can be used to establish a
better understanding of the relationship between the operational conditions and the metal temperatures
in superheaters. In the case analysis, the model simulated the thermofluid processes of a final stage
radiant superheater during steady state operations and a transient event. The detailed nature of the
temperature results from the model can provide temperature inputs to FEM analyses to study the
resulting thermal stresses during the operation. Thus, the model can used as a first step towards
establishing a better understanding, predicting and managing how the operational conditions affect the
life span of components such as the tubes, stub headers and main headers. Such integrated process
models can therefore be employed to study complex thermofluid process phenomena that may occur
during intermittent, transient and low load operation of power plants. In addition, such models could

be useful for predictive and preventative maintenance as well as online condition monitoring.
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Figure 7-18: Outlet metal temperatures corresponding to the discretization of the Flownex model at the steam

outlet side from the simulations of the events of the selected days.

It is important to note that in the transient analysis, the effect of different discretization of the heat
exchangers were not looked at. The transient analysis was based on the least possible discretization
that still captured the geometry effects correctly. However, increasing the number of building blocks,

will increase the solvers’ time to reach a solution.

University of Cape Town 169



Chapter 7. Refined methodology: Model application

7.5. Summary of the chapter

This chapter presented the application of the refined modelling methodology on a final stage radiant
superheater heat exchanger in both steady state and transient regime. This refined modelling
methodology and the model development of the final superheater were presented in the previous
chapter. Since a radiant superheater was modelled, it meant that the direct radiation which is leaking
both from the furnace and from the high temperature flue gas in the platen superheater region upstream
had to be accounted for. Real plant data from the DCS was used as inputs to the steam side of the
developed heat exchanger model as well as on the high-level models (boiler MEB and the Gurvich

method) used to generate inputs on the flue gas side.

The model that was calibrated using the C-schedule data in the previous chapter was re-calibrated first
before being used in the analyses of the behaviour of the heat exchanger. The rationale behind the re-
calibration was that it was plausible that the UA value for the C-schedule cases and real plant cases
would be different. This is because the operation environment and heat exchanger conditions for both
cases were likely to be different. The heat exchanger would most likely have degraded from when the

plant was commissioned.

The calibrated heat exchanger was then applied in steady state for two cases. In the first case, it was
assumed that the inlet steam and flue gas conditions were uniform. From this study some insights were
gained in cases where it would be difficult to install measuring instruments. 2-D temperature plots of
the flue gas and the metal outer surface on a selected section of the heat exchanger were presented.
Most important from the profiles was the illustration of the variation of the metal temperatures from
the inlet to the outlet on the steam side. In addition, a variation was observed at the outlet where the
inner tubes exhibited lower temperatures than the outer tubes which are slightly longer. A more
detailed analyses of the temperature distribution in comparison with the modes of heat transfer present
was also done corresponding to one discretisation channel on the flue gas side. It was insightful to note
that due to the presence of the direct radiation that heats up the outer surface of the outer fouling layer,
the temperature of this layer can be higher than that of the flue gas in the vicinity. Thus, this may result
in gas radiation and convection to the flue gas in the region than into the steam from the flue gas.
However, it is important to note that regardless of the possible reverse gas radiation and convection,
the resultant heat transfer is conducted to the steam flow and this conducted heat is entirely due to the

direct radiation.
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The second case was slightly more complex with the acknowledgement that the input conditions are
non-uniform as illustrated by plant measurements as well as CFD results. After adjusting for such
conditions in the model and the UA value re-calibration across the width of the heat exchanger, the
results were analysed. Notable the resulting temperatures compared well to the detailed temperature
measurements from the custom installed thermocouples. Element 23 was also studied in detail,
yielding successful and insightful results when compared to the detail plant data from measurements.
Note that all along the discretized model allowed for the steam flows within the tubes to redistribute
automatically based on the solution of the momentum conservation equations. This could unfortunately

not be validated in detail since it is not practical to conduct measurements regarding this.

The model built using the refined modelling methodology was further employed to model a transient
event that occurred during the operation of the plant. This was also aligned with illustrating the
potential applicability of the model for online condition monitoring purposes. The transient event was
sandwiched between two steady state cases: a full load and 65% full load. The developed high-level
models were employed to obtain the flue gas side data for both these steady state cases, with linear
distribution or averaging used to proportion the values during the transient event. The comparison of
the resulting outlet steam temperature from the discretized model to that from the DCS was satisfactory
and it conformed to expected operation norms. The discretised nature of the model also illuminated
the capability of the model to determine more detailed results in the heat exchanger than measurements

for most plants.

The detailed nature of the steam and metal temperature results that can be extracted from the
discretized model illustrates the usefulness of such a model in providing boundary conditions for
models that can be used to study the thermal stresses in tubes and header walls. The discretisation of
the heat exchanger building block demonstrated the ability of the model to calculate metal temperatures
radially along the thickness of the tube. The discretization of an element along the depth illustrated
that detailed steam and metal temperatures can be obtained at the inlet and outlet. Also, the
discretization of the element along the height provides access to tube metal temperatures that would
not be easily accessible using conventional measuring techniques. In addition, the ability to discretise
the heat exchanger across the width allows the model to be able to provide detailed outlet steam and
metal temperatures along the main headers. The steady state and transient temperature results can
provide useful input information for FEM analyses of the resultant thermal stresses in the tubes, stub

headers and the main headers. This will in turn serve to better understand, predict and manage how
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steady state and transient operational conditions affect the life span of these superheater heat exchanger

components.
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8. CONCLUSIONS AND FUTURE WORK

This chapter provides some conclusions based on the work presented in this report and highlights the

need for further work to be done in future projects.

8.1. Conclusions

Failures of tubes and outlet headers in heat exchangers of coal-fired boilers are major contributors to
unplanned capacity losses in the South African power generation sector. These failures are more
prevalent in the critical components which operate at high pressures and temperatures such as final
superheaters and reheaters. Various operational anomalies and operational procedures can contribute
to these failures. The continued introduction of intermittent renewable energy power plants on the
national electricity grid aggravates this situation. Coal-fired plants are now required to operate under
varying and low load conditions rather than at the base load conditions for which they were originally
designed. In order to mitigate the damage to critical components that may result from these changes in
operational conditions, a better understanding of the relationship between the operational conditions
and metal temperatures has to be established. Mathematical process models are invaluable to create

such understanding

Hence, a methodology to model the thermofluid processes of a heat exchanger in the convective pass
of a coal-fired boiler was developed. This methodology is based on the 1-D network approach
encapsulated in the Flownex software. It entails the simultaneous solution of the transient 1-D forms
of the conservation equations for mass, energy and momentum, combined with the applicable closure
relations, boundary values and initial values. The modelling methodology was then applied to model
two different superheater heat exchangers with complex geometries. On the tube side, the model traced
the complex steam path layout as in the real plant, which becomes advantageous if detailed tube
analysis is of importance. The sub-channel method was employed for the flue gas side. The heat
exchanger geometry is discretized along the flue gas flow path as well as along the steam flow path.
In order to limit the complexity, bundles of tubes can be grouped together and discretised along the
height depending on the level of detail required. Each heat exchanger increment contains the
appropriate geometrical information and thermal resistance characteristics. Empirical correlations
were employed to model the inner and outer convective heat transfer. An effective heat transfer
coefficient was used to account for radiation heat transfer on the flue gas side, accounting for radiation

from both the gas and the ash particles. For the radiant superheater, the direct radiation was accounted
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for as well. The methodology was applied on one tubesheet as well as connecting several tubesheets

in parallel in order to capture flow maldistribution across the width of the heat exchanger.

The developed models struck an appropriate balance between simplicity and accuracy such that they
were easily applied in a real-life cases to gain an understanding of operational flow and temperature
distributions in the heat exchanger. The detailed nature of the steam and metal temperature results that
can be extracted from the discretized model illustrated the usefulness of such a model in providing
boundary conditions for models that can be used to study the resultant thermal stresses in tubes and
header walls. The discretisation of the heat exchanger building block demonstrated the ability of the
model to calculate metal temperatures radially along the thickness of the tube. The discretization of an
element along the depth illustrated that detailed steam flow and temperatures as well as metal
temperatures can be obtained at the inlet and outlet. Also, the discretization of the element along the
height provides access to tube metal temperatures that would not be easily accessible using
conventional measuring techniques. In addition, the ability to discretise the heat exchanger across the
width allowed the model to be able to provide detailed outlet steam and metal temperatures as well as
steam flows along the main headers. The steam flow results can be useful input values for CFD models
for flow in the stub headers as well as the main headers. The steady state and transient temperature
results can provide useful input information to sufficient details for FEM analyses of the resultant

thermal stresses in the tubes, stub headers and the main headers.

Thus, the research has shown that a network approach can be employed to construct integrated
thermofluid process models of radiant superheater heat exchangers that provide results of outlet steam
flows and temperatures that can be used as boundary values for CFD and FEM analyses to determine
the resultant thermal stresses in headers. This will in turn serve to better understand, predict and
manage how steady state and transient operational conditions affect the life span of these components.
Such integrated thermofluid process models could be used to gain insights on typical problems and to
investigate the risks and impacts associated with intermittent operation, load following and variations
in coal quality. This, together with appropriate and sufficiently accurate measurements of key
parameters as well as the high-level models, could inform decision making for operating and
maintenance strategies to improve component availability and reliability, and to reduce the cost of

power produced.

Further insight that was gained is that the modelling methodology allows for a range of model

discretisation. This is due to the ease of set-up a model provided by the developed heat exchanger
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building blocks. Thus, the adopted level of discretisation depends on the complexity of the heat
exchanger as well as the purpose of the developed model. It is important to mention that the level of
discretization adopted in a specific model has an impact on the speed of developing the model
especially the opening of the inputs and results grid. However, this might be dependent on the computer
used. Thus, the use of a minimum number of heat exchanger building blocks for a particular purpose
is advisable. The level of discretization along the width also influences how the calibration of the
model is done. This is because different OFL thicknesses maybe employed at different width positions.
This is case dependant as the operational conditions differ with time. In the model calibration, it is

always important to observe if the inferred fouling thicknesses are reasonable.

8.2. Future work

Some insights were gained in the development and employment of the modelling methodology used
in this work. Some of these insights shone some light on the limitations of the modelling approach
with which further improvement could be done in order to fine tune the accuracy of the modelling and

cater for more thermofluid phenomena taking place in the heat exchangers.

One particular limitation of the modelling approach was illuminated in chapter 5, where the flue gas
side flow modelling was analysed. After the extensive analysis it was concluded that for the adopted
1-D network approach, the flue gas flow should be modelled using the subchannel approach. This
basically meant that the flow was restricted to flow in channels without any mixing in between, which
is not the case in a real plant. The major limitation was due to the solution technique adopted by the
network approach. Due to the inherent assumption of one-dimensional flow in the network approach,
the approach solves for total pressure as a non-directional parameter even at common nodes in cases
of two-dimensional flows. This leads to the software being used outside its inherent computational
domain characteristics which is flows in pipes. Thus, an extensive study can be introduced as future
work where a 1-D approach that solves for static pressure other than total pressure at the nodes is used
for comparison with the network approach. However, one of the advantages of the network approach

is that it eliminates solving the convective term in the direction of the flow.

In the development of the refined methodology some further insight was gained when analysing the
impact of accounting for the direct radiation originating from the flue gas in the final superheater region
towards the downstream final reheater heat exchanger. The refined modelling methodology
demonstrated some limitations in accounting for such a phenomenon. This was due to the number of

non-linear connections that would be necessary hence hindering the speed of the solver. This analysis
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illuminated that for better accounting of this direct radiation, refined models such as a zonal model
should be used. Such is recommended as future work for further development of the modelling
methodology. In this instance the convective pass will have to be modelled holistically accounting for

all the radiation in the spaces between the tubesheets of the different heat exchangers.
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Appendix A. Comparison of Lumped models to the Flownex model

APPENDIX A. COMPARISON TO RESULTS FROM LUMPED

PARAMETER MODELS — THE CALCULATIONS
A.1. Tube and shell geometrical information
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Fig. A-1: Geometrical information for the primary superheater heat exchanger [19].
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Fig. A-2: Aligned tube arrangements in a bank [123].
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Appendix A. Comparison of Lumped models to the Flownex model

D; == 32mm Dy, = 42mm
(Do e DI)

tw = - =5-mm

St = 104mm S; = 85mm

touterfoul ‘= 0.00102m tinnerfoul = Omm

Lsy = 24-5, = 2040-mm
Waoiler = 7920mm

148

Nsheets = T =74

Inlet tube length: Lintet = 5.34m

Outlet tube length: Loytlet = 4.46m
Number of passes: Npasses = 12

Liygpe = | for i€el.24

Ltubel. = Lintet + (i - 1) (

Ltube

Leube.total = O.SZLmbe = 58.8m

University of Cape Town

Deentre = 0.5-(D; + Do) = 37-mm

Number of tube sheets in SH1

Loutlet — LinletJ

2 Npasses =1
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Appendix A. Comparison of Lumped models to the Flownex model

Hfge = 4.46m

Hfg24 ‘= Hfge = 4460-mm

Loopfg height = | for i€23..1

0 = atan(

Ntybes = 2-Nsheets = 148

5.34m — 4.46m

= 23.962-°
1.98m

Hfgi = Hng_l + tan(8) -(SL)

Hyg

Hfgi = Hfg1 = 5.329-m

T 2 2
Affsi = Newpes 7 -Di” = 0.119-m

Pysi = Ngpes™D; = 14.879-m

2
Assroi = Hygi-( Wgi — Neheets-Do) = 24.449m

A-Avic
Pufal = —2L = 2328.471m

o

Vo’fg.total
Lg = = 2.886m

0.5-(Aggi + Astrge)

L
lengthratiogy = Lﬁ = 1.415
SH

Lta. = Lsk-lengthratiog; = 2.886m

University of Cape Town

T 2 2
Affse = Neubes - -Di” = 0-119:m

Pyse = Ngypes™Dj = 14.879-m

2
Affge = Hge-( Wige — Nsheets-Do) = 20.462m

4.A
Pufge = —2% _ 1948.808m

o

Ratio of model length to actual length to ensure
that volume of gas is calculated correctly within
FNX.

2 tubes per pass
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Appendix A. Comparison of Lumped models to the Flownex model

A.2. Material characterisation

Pipe material characteristics

k i’ w

Poipe = 7832—% kpipe(T) = [35.54 +0.004084-(T /°C) — 2.0891-10" (T /"C)z:l—
- 1) w

Ttest = 20 °C koipe| Ttest) = 35.613-——

m-K

Inner fouling layer characteristics
) w ) kg ) kJ
kinnerfoul = 0.07—— Pinnerfoul = 1700——  CPinnerfour = 1.085——
m-K m3 kg-K

Diji = Dj = 2-tipperfoul = 32-mm

Outer fouling layer characteristics

: w . kg ) kJ
kouterfoul = 0.07—— poyterfoul = 1700——  cpoyterfoul = 1.085 ——
m-K m3 kg-K
A.3. Steam conditions
ki
ms = 4622
s
psi = 9600kPa Ts = 337.7 °C
pse = 9600kP0 Tse = 400-9 °C
1 kg
l)Si := nn nn "“) e 44.378._3
Vsteam( Pei g™ s s m
nn nn nn kl
hsi = hsteam(psi , Tsi, ™ ,™ , ") = 2887.787-—
kg
- nn nn nn nn kj
CPpsj = CPsteam(Psi s disls: i B ) = 4.27-——
kg-K
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Appendix A. Comparison of Lumped models to the Flownex model

1 ki
P = = 35.934—2
v (p o un oo un ) 3
steam\ Mse > Ise» > 5 m
nn nn nn k’
hse = hsteam( Pse> Tse, " " » ) = 3108.754-k—g
CPse = CPsteam(Psea B e "") = 3.024-——

kg-K

Qs = mg-(hse — hs;) = 10208.655 -kW

o ) o
Ps = (Tse—Tsi) - kg-K
- nn nn nn nn T 5 kg
Hs -= I-"steaml:()-s( psi + Pse) > 0-5‘( Tsi + Tse) S S ] =23x10 m—s
nn nn nn nn W
ks == Asteam| 0.5(psi + Pse) , 0.5(Tsi + Tee) , ™ , ™ , ™, ™| = 0.065- —
CPs-Hs
Pr, = = 1.238
5
g = = g7 s = e
(Psi -Affsi ) s (pse -Affsi ) S

m
vs = 0.5-(vsi + vge) = 9.774—

)
. kg
ps = 0.5-(psi + pse) = 40.156 —
m
(ps'Vs'Di) 5
Re; = ————— = 5.4x10
Hs
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Appendix A. Comparison of Lumped models to the Flownex model

Gnielinski correlation

&6 = (1.8-log(Res) - 1.5)_2 = 0.013

£6 £
? -Res-Prs D. 3
Nu, g = J1+4]| —=— | |=1009.622
3 Ltube. total
’l £6 l 3
1+127- ? AP —4

ks w
hs g = Nusg-— = 2057.445-——

i maK
h,' = hs.G
A.4. Flue gas conditions
k
mpy = 6452
s
Pfgi = 100kPa Tfgi = 632.6 °C
Pfge = 100kPa Tipe = 5586 C
CPfa - = 2.139.
fg = =2.139-—
mig-( Trge = ngi) kg-K
Rig = 0 zsoi
flriees st kg-K
Pfgi k
Pfei = ﬁ = 0-442—1
fg ! fqi m
P, k
Prge = —2 = 0.481-Z
(ng'nge) m3
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Appendix A. Comparison of Lumped models to the Flownex model

Tsgi ki -5 k
Hpgi = ! ~ —2_ =3.995%10 " —
7  1.2618-10 16929 _ o MK s
1.2635x 10" + Thgi — Ttgi
K o2
T, k -5 k
Hfge = fg: —0_ - 3788x10 L
7  1.2618-10 1.6929  Bhuaiae m-s
1.2635x 10" + ————— .Tgre — ——— -Tpge
2
K
T =
ﬁ 1.0148
) - kK Tfgi w w
kigi ‘= 9.1446-10 ~-0.99958 = -————.—— = 0.063-——
K1.0148 m-K m-K
E 1.0148
. -5 K Tfge w w
kige = 9.1446-10 ~-0.99958 = -———.—— = 0.059-——
K1.0148 m-K m-K
m m
vgi = —2— = 5.9742
Pfai -Afffgi s
m m
Vfge = # = 6.554 —
Pfge -Afffge s

m
Vg = 0.5-(vpgi + vyge) = 6.264—

. kg
Pfg = 0.5-(prgi + Pfge) = 0.461—
m
. -5 kg
Hfg = 0.5-(pfgi + tifge) = 3.891x 10~ ——
m-s
w
kfg = 0.5-(kfgi + kige) = 0.061-——
m-K
CPfg-H
Pryy = —2 1% _ 1366
kfg
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Appendix A. Comparison of Lumped models to the Flownex model

Gnielinski correlations

Doo = Do + 2-toyterfoul
T
Lstream := ? 'Doo = 0-069m

Pfg Vfg-Lstream)
Refy) = (to Vo Ltream = 5136.921

Hfg

1

f— 3
Nufg.G.,aminar = 0.664- Refg_l‘Prfg = 52.803

0.037- Refg'lo's -Prfg
Nugg G.turb = 5 = 37.905

1+ 2.443-Refy) O'I-(Prfg o 1]

2 2
Nugg 6 = 0.3 + \/Nufg.G.Iaminar + Nugg G.turb = 65.3

&10.6 = (1.8-log(Refy;) — 1.5) ™% = 0.037

k w
hfg.G = NUfg_(.;-i - 90.342-—2
Drifg m°K

w

heo = hfg 6 = 90.342-——
2

m K
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Appendix A. Comparison of Lumped models to the Flownex model

A.5. Outside fouled wall temperature calculations

Tsi + Tse o
Ts_av := = 369-3‘ C
2
Tepi + T
Thov = % = 595.6-°C
w
ks = 0.07——
f m-K

w
159.‘@&'.:: kpipe( TS.OV) = 34.199- m

2
Aitube = TDj-Leype.total = 5.911m

1} w

Uitube = = 87.305-7
Do Doo m--K

Aitube -In| — Aitybe -In

1 D; Do
+ +
hs g 27"'kpipe'l-tube.total 2'T'kﬂ “Ltybe.total
Qs "
Tw = Teav+ = 502.957-°C

Ntubes - Ui tube -Aitube

A.6. Radiation heat transfer

(LS

T, i \

NN
S;

SONNONNNNNNNNNANNN

S S S S S S S
s1

Fig. A.3: Control volume of a tube in a tube bank [20].
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Appendix A. Comparison of Lumped models to the Flownex model

o= 5.67.10_8 - Stefan Boltzman constant
m -K
Ew =038
2'€w
Eeq = = 1.333
— é‘w

s =5, =0.085m s = Sr=0.104m
: T 2 2
Vpermeter = (31-52— z -Do ] = 0.007m

Apermeter = ™-Do = 0.132m

C=136
Vpermeter
5=C———— =0203m Geometric mean beam length
Apermeter
Do (4 s1°52
Scheck = C-— | —- —1|=0.203m
4 | w D 2
o

Weighted sum of grey gasses (Smith et al 1982)

atm = 101.325kPa

0 0 0
0
- -4
0.4201 | 1 0 6.508-10 -5.551.10
K= s b=
6.516 |atm 0 —-0.2504-10 *
131.9 _1 _a
0 2.718-10 -3.118-10

University of Cape Town

0

3.029-10

6.112-10" % -3.882.10”

1.221-10

7

74

7

0
-5.353-10
6.528-10

-1.612-10

11

11

11
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Appendix A. Comparison of Lumped models to the Flownex model

3 [ , (G-1)]
a1 = Z b”{?) = 0.397
j=1
3 [ 7, )0~
ay = Z by j- - = 0.213
j=1
3 [ 7.\~
a3 = Z b j- - = 0.093
j=1

ag=1-a1—ay—a3 = 0.297

_"'l.pg._
£, =1—e ™ = 0.081
1 L
)
£ =1-e " =073
2
)
_’.‘-3.pg._

)
3
g = Z ai(l—e '"J = 0.281
i=0

=In|1-¢ 1
i (—g) = 1.619—

S m
a-s = 0.329

< <
O -Egq-0-S (Tg —Tw)

hro = - 33.423. %
ro = = 33. —_—
a-s+ Eegq (Tg— TW) m2K
T -Eeq-A-S =3
Fug = ———— .0~ =0.264
a-s+ Eeq

w

hy = hgo + hp = 123.765-——
2

m K
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Appendix A. Comparison of Lumped models to the Flownex model
A.7. Overall heat transfer coefficient (UA)

2
Ao tube = TDoo-Ltybe.total = 8.135m

1
Ui tube.total =

k
- - . 57.72—39—
o 0o K
Aitube -In| — A tube -In| — °
1 D; D Aitube 1
— + +
h;

o

27"'kpipe‘l-l.‘ube.total 247‘kfl Ltybetotal  Ao.tube ho

Ntubes

kw
UAtybesheet = Ui tube.total -Ai.tube - = 0.682-—
sheets K

kw
UAnx = Ui tube.total -Ai tube -Ntubes = 50.497 - ——

A.8. Effectiveness NTU method — Part 1

w
= 1864.254-—
K

_ w
Cs == cps- = 2182.829-—

Nsheets

min( G Cs)

C, = — =0.854
¢ max( Cfg s Cs)

kw
UAtubesheet = 0-6827

UAtubesheet
NTU = —————

= 0.366
min( Gy Cs)
QS - - waw
QHx = = 137.955kW Calculated using given steam conditions
Nsheets

University of Cape Town
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Appendix A. Comparison of Lumped models to the Flownex model

A.8.1. Assume a counterflow heat exchanger

1 - exp[-NTU-(1- ¢,
Ef = exp[ ( C)] = 0.273 Effectiveness: Counterflow HX
1- C-exp[-NTU-(1-C) ]

ATgy = e -(Tpgi — Tsi) = 80.582-A°C ATgygnven = Tygi = Tgge = 78-A°C

Ttge.resultantef = Tfgi — Ang.cf = 552.018-°C Ttge.given = Tfge = 558.6-°C

m def — q
% = — 5 cppy- Ao = 150.226-kW %Errory = L% _ 89.%
sheets QHX

Aef . .
Tseresultantcf = Tsi + — = 406.522-°C Fedghien = Fop =009

s

“CPs
Nsheets
A.8.2. Assume a parallel flow heat exchanger
1-exp[-NTU-(1+ C.)]
Epf = = 0.266 Effectiveness: parallel flow HX
14+C
ATgypf = €pf-(Tfgi — Tsi) = 78.368-A°C ATg given = 74-A°C
° T eni— D08.6-"C
Tipessuhuntof = Tips— ATy or = 558332 ."C fge.given
m Gof — GHX
Qof = % -cpfg-ATfgof = 146.097 -kW %Errorps = Sk VAN =59-%
sheets QHx
: Gef . .
Tse resultantpf = Tsi + m— = 404.63-°C Tse given = 400.9-°C
s
“CPs
Nsheets
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Appendix A. Comparison of Lumped models to the Flownex model

A.8.3. Assume a cross flow heat exchanger

Cmax mixed, Cmin unmixed

1
Ecrossfm.u = (?] -[1 — eXpl:—Cc-(l - exp(—NTU))I] = 0.27
¢

ATfg crossfmu = Ecrossf.m.u'(rfgi = Tsi) = 79.53-A°C

nge.resultantcrossf.m.u = ngi - Ang.crossf.m.u = 553.07-°C

m

-cpfg- ATfg crossfmu = 148.265-kW

Gerossfm.u =
sheets

) Qcrossfm.u .
Tse.resultantcrossfmu = Tsi T m— = 405.623-°C
s

“CPs
Nsheets

Both fluids unmixed

= i

exp(-NTU-C.-n) - 1
Ce-n

ATfg crossfu = Scrossf.u'(ngi— Ts,-) = 77.11-AC

nge.msultantcrossf.u = ngi = Ang.crossf.u = 555.49-°C

m
Qcrossfu = 'Cpfg'Ang_crossf_u = 143.753-kW

sheets

) Gerossf.u .
Tse.resultant.crossfu = Tsi + - = 403.556-°C
s
CPs
Nsheets

University of Cape Town

Effectiveness: Cross flow HX

ATgg given = 74-A°C

nge_given = 558-6' oC

_ Gerossfm.u — QHX
BErroreossfmy = —————
qHx

Tse_given = 400-9' °C

Effectiveness: Cross flow HX

Ang.given = 74‘A°C

nge_given = 558.6' °C‘

Qcrossf.u — QHX
%ErrOrcrossfy = —————————
QHX

Tse.g,'ven = 400.9 - °C

=75-%

=4.2-%
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Appendix A. Comparison of Lumped models to the Flownex model

A.9. Flownex

Steam and flue gas temperature results that were extracted from the Flownex model with no

discretisation are shown below together with the resultant heat transfer for each tube in a pass.

397.393797 630.854414 6.475734094
397.4361909 627.4197773 6.330787975
390.8831236 624.0145121 6.366197743
390.8371868 620.6369988 6.227315389
384.5447741 617.2886246 6.257540238
384.5870671 613.9690613 6.119881041
378.556493 610.6799903 6.143823505
378.5109802 607.419829 6.012090991
372.7408073 604.189837 6.031323777
372.782771 600.9896815 5.900831498
367.2767574 597.8209975 5.913953841
| 367.2319059 | _ | 5%4.6822455 | | 5.789229733
Tsteam.Flownex = — ¢ Tfluegas.Flownex = 591.5744964 2 QFlownex = kw
5.798344595
362.0237995 588.4974044 5.674909837
339.9906161 585.3388193 6.102194089
339.9412232 582.099753 5.975015747
344.470647 578.958217 5.738517083
344.5228099 575.9125004 5.617718387
357.2415061 573.0127995 5.19403048
357.1931208 570.2550907 5.088267188
352.8534389 567.5261851 5.086621233
352.8988971 564.8258984 4.981544194
348.7718925 562.1554136 4.975468861
348.7246077 559.5133685 4.875506116
For each tube, the overall heat transfer coefficient (UA) was calculated as follows:
QFlownex
UAFlownex =
fluegas.Flownex — Tsteam. Flownex
Some of the results are shown below.
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Appendix A. Comparison of Lumped models to the Flownex model

0 0

0 0.028 8 0.026

1 0.028 9 0.026

2 0.027 10| 0.026

3 0.027 11| 0.025

B 0.027 12| 0.025

5 0.027 13| 0.025

6 0.026 kW 14| 0.025 kW
UAriownex = |7 | 0.026 T UAFiownex = (15| 0.025 T

8 0.026 16| 0.024

9 0.026 17| 0.024

10| 0.026 18| 0.024

11| 0.025 19( 0.024

12| 0.025 20| 0.024

13| 0.025 21| 0.024

14| 0.025 22| 0.023

15 23

Thus, the resultant average overall heat transfer coefficient (UA) from the Flownex model is

: kw
UAay.Flownex = UAElownex = 0-609'7

A.10. Effectiveness NTU method — Part 2

Using the overall heat transfer coefficient (UA) calculated from the Flownex model results, the process
of using different lumped heat exchanger arrangements assumptions in conjunction with the

effectiveness NTU method was repeated.

The Number of Transfer Units (NTU) were re-calculated as shown below.

) UAav.Flownex
NTU, = —_— = 0.327
mm( G Cs)

ATmax = ngi = Tsi

Gmax = min(Crg, Cs) - ATmax = 549.769-kW
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Appendix A. Comparison of Lumped models to the Flownex model

A.10.1. Assume a counterflow heat exchanger

1- exp[-NTU,-(1- C) ]
Ecfnew = ' = 0.251 Effectiveness: Counterflow HX
1- C.-exp[-NTU,-(1- C) ]

Qcfnew = Ecfnew -Gmax = 137.864-kW

Qcfnew — QHX
%Errorgfpewy = — = —0.07-%

QHX

A.10.2. Assume a parallel flow heat exchanger

1-exp| -NTU,-|1+ C,
Epfnew = [ 1 (; ( C):I = 0.245 Effectiveness: parallel flow HX
03 C

Gofnew = €pfnew Gmax = 134.739-kW

Qpfnew — QHX
%Errorpfpewy = —— = —2.3-%

QHX

A.10.3. Assume a cross flow heat exchanger

Cmax mixed, Cmin unmixed

Effectiveness: Cross flow HX

1
Somtmanes = | 1 ool (1ol ] - 026
c

Gerossfm.u.new = €crossfm.u.new "Gmax = 136.375-kW

_ Gerossfm.u.new — GHX
BEroreossf mu.new = =-11-%

QHx
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Appendix A. Comparison of Lumped models to the Flownex model

Both fluids unmixed

-0.22
n, = NTU,

Ceosgunes = Soh s vnmiced

exp(-NTU,-C.-n;) — 1 Effectiveness: Cross flow HX
=1—-exp = 0.24
Cc 'nr

Qerossfu.new = €crossfu.new -Gmax = 132.104-kW

_ Qcrossf.u.new — GHX
%ErrorcmssﬁU.new - = —4.2‘%

AHx
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Appendix B. Flue gas side modelling

APPENDIX B. FLUE GAS SIDE MODELLING

B.1. Inviscid flow: Network approach

Consider the case shown in Fig. B.1, with the known inputs as follows;
pr =0Pa

py =0Pa

Fig. B.1: Set-up for a 2-D technique that solves for total pressure (1-D network approach).

The total pressure at the outlet nodes is given by
_ +l 2 _0 5 2
Por = Pe 210ME_ DUg

1
Pon = Py +§p"12v :O-5V12v

Solving for total pressure on the common node P for this case which has no resistance to flow implies

that

Por = Poe = Pon
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Appendix B. Flue gas side modelling

Combining the two equations, it can be concluded that

Thus, with the technique that solves for total pressure, the flow equally splits between the two legs. A

summary of the results from the basic fluid mechanics calculations are as follows;

Pow =4.5Pa  p, =4.5Pa p,,=4.5Pa p,, =45Pa p,, =4.5Pa

u, :%x5 +%x1+%x3+%><3 =3.667 —
S

uWZS? v521? uP:3.667? uE:3? VN:3?

py =—8Pa  p;=4Pa p,=-222Pa p,=0Pa p, =0Pa

These results are summarised in Table B.1.

Table B.1: A summary results from the network approach for a steady state case with inviscid flow.

Details Symbols | Units Values
x outlet velocity u, m/s 3

y outlet velocity v, m/s 3
Internal static pressure Dy Pa -2.22
Inlet x static pressure Dw Pa -8
Inlet y static pressure Ps Pa 4
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APPENDIX C. RADIATION MODEL — VDI HEAT ATLAS

General inputs

Emissivity of the tube wall:

Stefan Boltzman constant:

Inlet flue gas temperature:

Outlet flue gas temperature:

Average flue gas temperature:

Average wall temperature of
the outer fouling layer:

Mean beam length
Outer tube diameter:

Transverse pitch:
Longitudinal pitch:
Radiation constant:

Mean beam length:

Gas specification

Total pressure of the system:
Fraction of CO?2 in flue gas:
Partial pressure of CO2:

Fraction of H20 in flue gas:

Partial pressure of H20:

University of Cape Town

Appendix C. Radiation model -VDI heat atlas

ey = 0.7

o= 567-10"%.

msz4

TgFSHm = 1059.07 °C
Ty FSH.omt = 942.608°C

T -+ T
T, = ki - gFSHowt _ 1 73089k

T, = 790.826 K

Do = 0.0445m
51 = 0.0455m
sy = 0.8571m
Crad =36
D 4 S1-S

o 4 =172

Lmb = CradT ; - > —1|=0964m
o
p; = lbar

Xcop = 21.424%
PCO2 = XcO2Py = 0-214-bar
XHO = 4.545%
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Emissivity of gases and particles

Emissivity of gases
Carbon dioxide:

Gas temperature:
Emissivity of Carbon dioxide:

Water vapour:

Gas temperature:

Emissivity of Water Vapour
uncorrected:

Pressure for correction:

Correction factor f:
Emissivity of Water Vapour:
Pressure for interphase:
Combined :

Gas mixtures interphase
Emissivity of the gas mixture:

Emission coefficient for the gas phase:

Emissivity of particles

Density of the fly ash material:

Diameter of the particle:

Mean relative absorption efficiency:

Particle load:

Specific projection area for the particle:

Mean relative backscattering efficiency:

University of Cape Town

Appendix C. Radiation model -VDI heat atlas

pcoz Lmb = 0.207-bar-m
Tg = 1274K

Eco2 =013

Tg = 1274K
= 6102
€H20 uncorrected = ©
PH20 * Pt
pf.H20 = 5 = 0.523-bar
fino = 1.0

€420 = TH20 " H20 uncorrected = 0-06

PH20

— - 0175
PCO2 * PH20

Pinterphase =

PLeomb = (PCO2 * PH20 ) Linb = 0-25-barm

Aeg = 0.0125

Eg = EHzo + Eco2 = A&'g

n(1-¢ q
B el Gk 1 8 0.203 —
mb m
_ kg
pp = 2963—=
m
4, = 16107 e
Qups = 032
neae ke
L, = 0.046—=
m
3 m2
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Appendix C. Radiation model -VDI heat atlas

: 2'Qbsc
Auxiliary term: Tp = 1+ = 2385
Qabs
A[,p — 1

Auxiliary term: B = = 0.409

p Yo+ 1

P
Opncal thickness for emission for (bp = Qabs'Ap 'Lp'Lmb'Afp
particles: |
1—exp(—P
Emissivity of particles: & = (1 - ﬁp)'[ N ( _ﬁ)) ]
Bp-exp(~p)

Optical thickness for emission for the gas ,_ =
solid mixture: q’emi_gp a (QabS'A'P 'Lp - Kemi.g) Lob- ip

1- exp(—(l’emi_gp)
1+ [5p-exp(-(l’

Emissivity of gas + particle mixture: €gp = (1 = Bp) [ : )
emi gp

Absorptivity of gases and particles

Carbon dioxide:
Wall temperature:
Emissivity of Carbon dioxide:

Correction factor

Absorptivity of Carbon dioxide:

Water vapour:

Wall temperature:
Emissivity of Water Vapour
uncorrected:

Pressure for correction:
Correction factor £

Emissivity of Water Vapour:

Absorptivity of Water vapour:

University of Cape Town

PCO2 Lmb = 0.207-bar-m
T, = 791K
Ecoz.w =10.125

fco2w =1

0.65
Tg
Qco2 = fcoz.w'[T—] “€CO2.w

w
pH20'Lmb = 0.044-bar-m
T, = 791K

- . -2
€H20 uncorrectedw = 9-10

szo =1

€H20.w = TH20 "€H20 uncorrected w = 0-09

w

Tg 045
OH20 = T_ “€H20.w
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Appendix C. Radiation model -VDI heat atlas
= 0.175

Pressure for interphase: Pinterphase

Combined : pL = 0.25-bar-m

comb
Gas mixtures emissivity interphase at wall: Aew = 0.007
Absorption of the gas: Qg = 00 + Aco2 ~ Ay

—ati-
(—ﬂg) = 0_333i

mb L

K

Absorption coefficient for the gas phase: abs.g =

gigc;lb{ tl:lincl?ess for emission for the gas D g = (Qabs' Ap 'Lp +K abs.g) 'Lmb"‘l'p

1- —q
Absorptivity of the gas + particle mixture: Ogy = ( 1 ﬁp) [ exp( >abs.gp) ]

1+ Bp'e"p(_‘babs.gp)

Heat transfer coefficient

S g -T4—Ol -T 4 W
hq = i of 2L B ¥ I_9989
ad = +e.,—0Q . -E. T="T. ’
egp W gp "W g W m -K
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APPENDIX D. MASS AND ENERGY BALANCE

D.1. Coal composition

NOx fraction during combustion:

Mass fraction
Carbon: mf~ = 43.94%
Hydrogen: mfyy = 2.5%
Oxygen: mfy = 9.61%
Nitrogen: mfyy = 1.11%
Sulphur: mfg = 1%
Water: mffpyo = 10.51%
Ash: mf, = 31.33%

Confirming mass fractions

fNOX = 70%
Calorific value:
kJ
CVe = 32778.15—
kg
kJ
CVy = 119931.72 —
kg
CVgp = 0H
kg
C\7N = 0E
kg

CVg = 9257.524£

kg
kJ
CV =0—
H20 kg
kJ
CV =0—

Appendix D. Mass and Energy Balance

Stoichiometric
coefficient

StC z=ili]

1

Styy = —

H™
Stg = —=

1
StN = fNox 3
Stg = 1.0
StHzo =10
Stash = 0

mftot = nfC+mfo+me+mfN+me+me20+mfash = 1

D.2. Mass balance and flue gas composition

Unburnt carbon

Mass of Unburnt Carbon
per kg of Coal:

Mass of Unburnt Carbon per kg of
Cabon:

Energy loss of Unburnt Carbon per kJ
of Energy input:

University of Cape Town

C' = mf, g {(%Cg, %FA) + (%Cpy%BA) | = 0313-%

G = i = 0.713-%
wf
CVn-C
&
C" = - = 0.626-%
CVeoal
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Appendix D. Mass and Energy Balance

Air Ratios
Theoretical Air/Fuel Ratio:

M. mf~ — C'
TAR = —— 1§ (st ~C)

TAR = 5523
Dry Air Required at Furnace:

Humid Air Ratio:

Mole Fractions

New Carbon Mass Fraction:

Moles of Carbon:

Moles of Hydrogen:

Moles of Oxygen:

Moles of Nitrogen

Moles of Sulphur:

University of Cape Town

%O | C  Mc

‘ mf(y mfyy mfy mfg
+ St + Sty —— + =Sty —— + Stg-——
M My 2 NM Mg

b - | .

DAR := TAR (1 +EApy g jnlet) = 6481

HAR = (1 + w) DAR = 6.492

mfe = mic - C' = 43.627.%

o M
MC Iﬁ‘cu
mfyy M

gy — o — o U683
MH mev
mfy M

ng = P i
Mo mfc
mfyy M

my = = 0.022
My mi
mfe M,

S €

ng = ——— = 0.009

Mg mfe
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Appendix D. Mass and Energy Balance
Flue gas constituents per kg Coal (FRconst)
Mass Flow Rate of CO, per kg Coal: FRcop = ng-———— mf = 1.599

Mass Flow Rate of H,O per kg Coal:

1 Mmo
FRH.20 = - D lnfcv + DAR-w+ mezo = 0.339
2 Mc
Mo2
Mass Flow Rate of O, per kg Coal: FRp, = EA AH.fg.hlet'TAR'%OZ.air'M_ =10.222
“air

Mass Flow Rate of N, per kg Coal:

M M

1 N2 N2
FRyp = 5(1 - fNox)-nN-M—-me» - DAR-%Nz_air-M—' = 4901
= i
Mno
Mass Flow Rate of NO per kg Coal: FRNO = fNOx nN\/I— -mf = 0.017
B -
Mso2
Mass Flow Rate of SO, per kg Coal: FRg) = ng- = -mf = 0.02
M,
Mass Flow Rate of Argon per kg Coal: FRp, = DAR-%Ar,. -—— = 0.082
~air

Mass Flow Rate of Fly Ash per kg Coal: FRpp = mf, -%FA~( 1- %Cfa) = 0.279

Mass Flow Rate of Unburnt Carbon in Coal: FRUBC = mfash -(%Cfa-%FA) = 0.003

Mass Flow Rate of Flue Gas per kg Coal:

Checking if flue gas mole fractions are

~S = 0 =
R FGR ek = HAR + 1 — mf, -%BA = 7.461
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Appendix D. Mass and Energy Balance

Flue gas constituents per kg Flue Gas (mfconst.fg)

F
Mass fraction of CO2: meOZ.fg = §§§2 =21424-%
FR,
H20
Mass fraction of H20: me20_fg = TR = 4545-%
FRo)
Mass fraction of O2: mf, = — = 2976-%
FRypp
Mass fraction of N2: mf; = —— = 65.684-%
N2f 7 FGr ’
FRnO
Mass fraction of NO: mfNO fo = ﬁ = 0.223-%
, FRgo2
Mass fraction of SO2: l’nfso2fg = ﬁ- = 0.268-%
FRp,
Mass fraction of Argon: mfp fg = TR = 1.102-%
FRpp
Mass fraction of Fly Ash: mff Afg = ﬁ = 3.741-%
FR.
UBC
Mass fraction of UBC: meBCfg = W = 0.038-%
Flue Gas molar mass
. kg
Mgg = mfg) g-Mop + 102 £ Mco2 + N2 fg MN2 = S

+mfg0) f5-Ms02 + MH20 f5 MH20 + MNO £ MNO * MfuBC 5 Mc

D.3. Energy balance

Enthalpy of flue gas using NIST Data:

hfg 1) = mfozfg -h (Constantsh s 02 S T) + mfcozfg -h (Constantsh X C02 S T)

+ mfN2fg ‘h (Constantsh N ,T) + meOZ.fg ‘h (Constantsh ,SO, ,T)

+ 061> 0 fg Prvater (D + 1y c £ bUBC (D + mfEA g5 bEash (D -
+h (Constantsh ,NO, T) mfNofg
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Appendix D. Mass and Energy Balance

Moist Air Enthalpy:
T—0°C T-0°C kJ
h T,w) = 1.006- + w{ 2501+ 186 ——— |||-—
ASHRAE()[(KJ[ (Kmkg

hoist air (T- @) = BASHRAE (T, w)

Net Energy recieved by Steam:

Ein = M5 PrimRH Bs PrimRH in + ®'s econ Ps.econin -
+m'g ATT] Bg ATT1 + @5 ATT2 B ATT2 + ™5 ATT3 Bs ATT3

E out = M FSH B FSH.out ™ ™'s FRH s FRH out = 3424.97-MW

E ;, = 1445.628 MW

AQgteam = Eg out — Es.

Mass flow rate of coal

_ AQgteam ~ Qredits * (Bmoist air  Tair AH.out> ) = Bmoist air( Tamb> ) ) ™seal air

w' g = - 976162

(1 - qradloss)'cvcoal =
+hcoa1(Tamb)

+HAR'%Airing'hmoist.air(Tamb’ “’)

+HAR (1 - %Aityno ) bygice i Tair AHLout> %)

+-FGR gy (Tgy Afin) -

+-mf g -%BA-hE A o (Tha exit)

D.4. Mass flow calculations
Mass flow rate of total humid air into the boiler

- ; kg
Inchuding ingress air m'p, = HARm' = 633A723T

Mass flow rate of humid air into the furnace

. ) - kg
i 7= HAR-(I — %A:rhg)-mcoal = 621.936—5-

Mass flow rate of ingress air

: . . . kg

Mass flow rate of flue gas at the economizer exit

mig, = FGR-m oy = 7284158
S
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Appendix E. Gurvich thermal model

APPENDIX E. GURVICH THERMAL MODEL

E.1. Flame temperature calculation
. MJ
CV ool = 16.41-k—g

Calorific value (HHV):
: kg
mcoal = 97.616?

Mass flow rate of coal:
Enthalpy of coal: h o T — 60.706 kJ
meoahk. coal( amb) = OV k_g
Humid air ratio: HAR = 6.492

Mass flow rate of hot air into the furnace: m'haﬁnnac e = 621936 E
s

i ; kg
Mass flow rate of flue gas: Mo furnace = 716.628 ?
Bpa frnace *= Bmoist.air  Tair A/H.out )

Enthaply of hot air:
B fince = 294.705-k—
” g

_ ™'ha furnace Pha farnace + m'coal'(hcoal = CVcoal)
hfame = y .
M ha farace * ™ coal

Flame enthalpy:

kJ
= 2489.18.—
hﬂame kg

Assumed only for this
short calculation. No

Calculation of Flame Temperature:
T9ame num = 1800°C
need to update.

Initial Guess Value:
(D= hfg(T) ~ bjame
Tame = root (f(Tﬂame.num) ’Tﬂame.mnn)

Tame = 1983.55-°C

E.2. Furnace geometry

Width of the furnace (at burner level): ; o
( el) Weynace = 24.003m
Depth of the furnace (at burner leve )
P ( D Beyrmace = 13.716m

University of Cape Town
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Height at the middle of the hopper:

Height of the burner rows:
Row 1:
Row 2:
Row 3:

Burner height:

Height at the bottom of the platen superhater:

Height at the roof:

Height at the middle of the platen superheter;

Height from dry hopper to center of furnace exit:

Burner relative height:

Parameter:

Relative height of flame center:

Area of the furnace:

Volume of the furnace:

University of Cape Town

by :

Appendix E. Gurvich thermal model

hhopper = 11084mm
howl = 19360mm
hrow2 = 26675mm

Brow3 = 33390mm

B hlrowl + hrow2 + hrow3 B
= 3 opper

by, = 15.391m

hplat.bot = 53400mm

hroof = 71385mm

 byjatbot T Broof
Bplatmid =~ 5
hhp = hplat.mid - hhopper
hhp = 51.308m
xb = i =03
hyp
Ax = 0.05

Xfame = Xp + Ax = 035

4281m>

Aﬁlmace -

15794m>

Vﬁmlace -
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Appendix E. Gurvich thermal model

E.3. Furnace Exit Gas Temperature and Flame radiation

Calculation of Furnace Exit Temperature:

1270:52°C Trial and error

hFE d

KJ
be (Trg) = 157332

£(D) = b (T) — bg

Tgg. = root(f(Tg). Tpg) = 1270.52-°C

Theoretical combustion temperature: Tgame = 1983.55-°C
Furnace outlet gas temperature: Tgg = 1270.52.°C
Mean overall heat capacity of the c __ Bfame ~ PFE m'fg-ﬁmce
combustion products: Pfurnace = T —Ter y
flame ~ “FE M eoal
kJ
C =943.——
Pfurnace ke K

Excess air in the furnace: EAg nace = 15.0%

Coal ash (as-received basis) Xash = 31.33%

Fly ash ratio: %FA = 90-%
Flue gas pressure: Pfamace = 10°Pa
Reference pressure: po = 9.81 -104Pa
Diameter of ash particles: dfa = 16pm
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Appendix E. Gurvich thermal model

Calculations
Excess air coefficient: BA  ifumnce — 1 + B8 e = 115
Water vapor volume (normal state):
VH20 fumace = YoH20 * 0'0161'(EAcoefﬁxnace - 1)-\70
3
VH20 furnace = 0-341 =
kg
Flue gas volume (normal state):

Vg.fumace = VO.g + (EAcoeffumace - lv)‘VO +0.0161 '(EAcoef.ﬁnnace - 1)"’0

3
m
Ve fumace = 59515
V;
RO2
'RO2furmace = 3, -~ 0148
g furnace
VH20 furnace
Vohime fraction of water vapor: TH20 fumace = 7— = 0.091
b g furnace
Volume fraction of triatomic gas: Tnfurnace = TH20 furnace * TRO2 furnace = 0-23%
Partial pressure of triatomic gas: Baiince = Sifiniice Phiie = S9MD-4S
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Appendix E. Gurvich thermal model

Flue gas mass excluding the fly ash:

. kg
Ggfumace =l-xp+ 1'306'1*:/“<:oeffumace'VO'_3
m
Gg.fumace = 7.911
Flu Sencd ) Ggfumace kg
e gas density: Pg fumace = = = 1.329—-5
g furnace m
Vﬁnnace
Effective radiation layer thickness Lo — $————
(Mean beam length): Afurnace
Lb famace = 13-282m
Gas coefficient:
078+ 1.6r fiana L -
kg furnace = e i ST -(1 - 037 0?;]-10"’. lp
Pp furnace Lmb.ﬁ.lmace i ’ el
Po m
k = 6838x10"/
g furnace = % “mPa

. . . - 5 1
Extinction coefficient of the gas: Kg.fumac e = kg.fumace T farnace Pfarnace = 0-016 ™

Fly ash concentration: M6 farnace = ——— = 0.0356

Fly ash coefficient:

University of Cape Town
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Extinction coefficient for fly ash:

Appendix E. Gurvich thermal model

1
Kfa furnace = Xfa furnace ‘Mfa furnace Pfurnace = 0'186;

Dimensionless number for bitumite coal:

Dimensionless number for pulverized
coal furnace combustion:

Extinction coefficient of the coal flame:

T |
k. =10".
<0 m-Pa
Xl =005
X2 =101

Ko = X1 %Koo Phurnace

1
Kgo = 0.05—
Extinction coefficient of the ) 1
coal flame: Kgame = Kgfumace + K¢, farnace + Ko = 0233 ;
Coal flame emissivity: flame = 1~ e"p(_Kﬂame'Lmbfmnace)

Furnace configuartion factor:

Water wall fouling factor:

Average thermal efficiency coefficient:

Average thermal efficiency coefficent:

Flame center modification factor:

Heat loss due to furnace wall radiation and
convection:

Heat preservation coefficent:

University of Cape Town

€fame = 0.965
Xfurnace = 1
(slag =045
Vave = Cslag Xfurnace = 043
__ “flame
“furnace = als '
llave'(l il E‘ﬂame) * €flame
Sfurnace = 0-984
B :=0.56 C=05

M := B - C-xgype = 0.385

Grad loss = 0-9-%

Pfurnace = 1 ~ 9radloss = 0-991
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Appendix E. Gurvich thermal model

Stefan boltzmann constant: op = 567-10° 8
2.4
m~-K
) Tﬂame
TEE calc = 0.6

3 = Qi
(“O'Tﬂame “furnace Afurnace ave)

‘Pfarnace 'mlcoal 'Cpﬁnnace

M-

Remember to
update initial
guess value

Fumace oulet gas temperature error: Errory FE = TrE ~ TFE calc

Eﬂ'Ol'TFE = —0.072-A°C

TEE cale = 1270.59°C

Furnace outlet enthalpy: bEE cale = hfg(TFE.c alc)
kJ
bEE calc = 157341
Furnace outlet gas enthalpy error: Errory, pg = bpg — BFE calc
kJ
Error, = —0.091-—
hFE ke
& m'fg.ﬁ.lmace
Radiative absorbed heat of 9 furnace = P& ce'(hﬂ - hFE.calc)' S
furnace: ‘ M coal
K
I 6662.4-k—g
Thermal load of radiation heating surface: "
Q - Mcoal Y furnace
Calculated value: r furnace -~
culated value A .
kW
Q xnnce = 151.917-—2
m
E.4. Platen superheater geometry
Platen superheater height hotat = Broof ~ Bplatbot = 17-985m
Platen superheater depth: bplat = 1053mm + 914mm + 2145mm + 914mm + 1053mm
bplat = 6.079m
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Platen superheater avarage
transverse spacing:

Platen superheater average
longitudinal spacing:

Outer diameter of the tubes:

Average thickness of the tubes:

Inner diameter of the tubes:

Number of rows:

Average spacing in the longitudinal direction:

Number of tubesheets/elements in parallel:

Number of tubes per tubesheet:

Inlet radiation area:

Qutlet radiation area:

Configuration factor from inlet to outlet:

Superheater area exposed to radiation:

University of Cape Town

Appendix E. Gurvich thermal model

STplat = 1143mm

St = 3%mm
ODplat = 38mm
_ 3.6mm + 5.6mm + 8.8mm
tp]at ’ 3
tplat = 6-mm

]Dplat = ODplat = 2tplat = 26-mm

Nrows plat = 2822 = 112

b
Sp gy = — P = 54.766.mm

P N =
rows.plat

) Wurnace

Ntubesheets.plat =3 -1=20
T plat

Ntubes.tubesheet_p]at =282 =56

Aradplatin = (Bplat * Pplat ) ST plat Neubesheets plat

Avad platin = 550.103m”

Arad.plat.out - hpl::lt 'ST.plat'Nmbesheets.plat

Avad plat ot = 411.137m?
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Appendix E. Gurvich thermal model

E.5. Platen superheater thermal calculations

Furnace exit temperature: Tgg = 1270.52.°C
W'ty furna KJ
a g ce kI
Furnace exit enthalpy: hEE plat = hFE.—m' » = 11550.22-kg
co

Qroofplat = 9-754MW  Trial and error

Quw plat = 14.178MW Trial and error

Tg plat.out = 1059.07°C Trial and error

Average platen superheater flue gas T a Tpg+ Tg.plat.out
temperature: gplatave 2
Tg.plat.ave = 1164.8-°C
L o 1.8
Mean beam length for the platen superheater: mb.plat -~ 1 1 1
+ -
[ bplat  Bplat ST-plat]

Excess air coefficient: EAcoef. conv.pass = 1+EA AH fg inlet = 1.174

Water vapor volume (normal state):
VH2O.conv.pass = Vom0 +0.0161 '(EAcoef.com'.pass - 1)'\"0

3

7 y m
v H2O.conv.pass — 0'542k_g
Flue gas volume (normal state):

vg.conv_pa_;s = vO.g : (EAcoef.conv_pass N l)'VO +0.0161 '(EAcoef.conv_pass = 1)‘\"0

3
. m
Vg conv pass = 69661~
\.7
RO2
"RO2convpass =y 0.145
g.conv.pass

) VH2O.cor1v.pass
rHZO.conv.pass = —V = 0.089
g.conv.pass

Volume fraction of water vapor:

Volume fraction of triatomic gas:

Tn.conv.pass = TH20.conv.pass T TRO2.conv.pass = 0-235
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Appendix E. Gurvich thermal model

Partial pressure of triatomic gas:

Pn.conv.pass = 'n.conv.pass Pconv.pass = 23.481 kPa
Flue gas mass excluding the fly ash:
. kg
Gg.conv.pass =1-xp t 1'306'EAcoef.conv.pass'VO'_3
m
Gg.conv.pass = 8.059
G
- __ gcomv.pass kg
Flue gas density: Pg conv.pass = ~ 1.328—3
g.conv.pass m
Gas coefficient:

_ 0.78 + 1'6'rH20.conv.pass
gplat — 3
J Pn conv.pass “mb plat

k

T : 51
7 g-plat.a‘e]_lo->

-01|{1-03
1000-K

.m-Pa

P m
—6: A
k =6418x10 -
gplat m-Pa
" . 1
Extinction coefficient of the gas: Kg.plat = Kg plat Tn conv.pass Pconv.pass = 0.151 ;
Fly ash concentration: Hfa conv.pass = = 0.035
Gg.conv.pass
4300 p —
Fly ash coefficient: . . R ke 10-5. 1
aplat - 1 m-Pa
3
2 2
Tg.plat.ave E
K pm
ke g = 7,062 1075.—
aplat ) m-Pa
Extinction coefficient for fly ash: :
Kfa_p]at = kfa_plat""'faconv.pass'pconv.pass = 0'247;

Radiant absorption coefficient of flue gas radiation:
Kplat g (kg.plat Th.conv.pass + k'faplz\t'”fzxconv.pass)

1
= 3.978.
Kplat m -MPa

Flue gas emissivity: Cgplat = 1~ exl’(_Kplat “Pconv.pass 'Lmb.plat)

Egplat = 048
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Appendix E. Gurvich thermal model

Coefficient considering reradiation: ﬁplat =092 from figure D6
- -0.5-
Bt ob o Xoiat = (troof ~ Bhopper) ~ 05 hpiar
00! er
Byoof — hhopp
Xplat = 0.851
Thermal load distribution coefficient: Nplat = 0.78
TN S—
Radiation heat flow of panel zone: 9 platin = plat % o eoe
Afurnace
= 118.495 o
A platin = T
m
c A caative3 S ) ) kW
orrected radiative intensity of panel zone: 9 plat in.cor = Bplat'qr.plat.in — 109.016-—2
m
Radiation heat absorbed by panel directly . _ SrplatincorArad platin
from furnace: A platin = m' oy

y kJ
qr.plat.in = 614344 k_g

q'r.plat.in'(1 - eg.plat)*"plat

Furnace radiation heat leaked out of panel: q'r.plat.o ot =

Polat
q'r.plat.out = 32.362-k—g
Direct heat transfer from the furnace - photons
Radiation heat absorbed by panel: q'r.plat = q'r.plat.in - q'r.plat.o ut

, KJ
Ut = 581982

Qr.plat = cl'1'.plat'm'coal
Qq plat = 56811 MW

Direct heat transfer from the flue gas in the platen superheater into the final superhater:

Inlet area of heating surface behind the Arad FSHin = Arad platout = 411.137m’
platen superheater: ) )
Flue gas absorptivity in the platen: Og plat = Egplat = 0.48
Correction factor of fuel: =05
4

o - o -T s -G

Platen gas direct radiation into FSH: qq = i AradF'SH.m solstave
M ¢oal
q = 245.027-H
kg
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Appendix E. Gurvich thermal model

Direct heat to the FSH from flue gas in Qg. direct FSH = 9™ coal

platen:

Qg direct FSH = 23-919-MW
Direct heat to the FSH from photons - '
from the furnace leaking out of the Qphot direct FSH = drplat out ™ coal
platen:

Qphot direct FSH = 3-159-MW

Total direct radiation into FSH Qrad direct FSH = Qg direct FSH * Qphot direct FSH

Value Qrad direct FSH = 27-078-MW
From energy balance
Inlet steam conditions
Temperature: Ts.plat.in = 386°C
Pressure: Psplatin = 18.02MPa
Enthalpy: hs.plat.in = Bgteam (ps.plat.in ’Ts.plat.in’" - ’")
kJ
hs.plat.in = 2804.66-k—
Outlet steam conditions g
Temperature: Ts.plat.out = 480°C
Pressure: Ps plat out = 18.02MPa
Enﬂlalpy: u wowm wwm
hs.plat.out = Bgteam (ps.plat.out ’Ts.plat.out’ > )
kJ
hs.plat.out = 3205.37-k—g
Energy absorbed by the platen: Yplat = Bs plat out ~ hs.plat.in
kJ
Gt = 400715
: ; : kg
Mass flow rate of steam: s pfat = 504.667 T
Total heat absorbed the platen Qplat.tot = m's.plat'qplat = 202.228-MW

Comcioit gt Qe ™ Qplveor ™ e

Qe plat = 145417:MW

If steam conditions are not known, this value is assumed then adjusted via trial and error
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Appendix E. Gurvich thermal model

Flue gas exit temperature for the platen superheater

Energy balance:
" ) W'y BEE cale T Mair ing 'hmoist.air(Tamb’ “’) ~ Q¢ plat ~ Quww plat ~ Roof plat ~ Qg direct FSH
plat.out.calc -~ ;
gp m'g

hS-Plat.om_calc = 1308.87.1':_g

Initial Guess Vae ~ 1gplatout = 1039-07-°C
iD= hfg(T) - hg.plat.out.calc

Tg.plat.out.calc == root (f(Tg.plat.out) ’Tg.plat.out) = 1058.473-°C

_ E““T.g.plat.om e Tg.plat.out . Tg.plat.om.calc

Remember to update

— A2
EMMor] g piat out = 0-597-A°C e

Heat transfer to the roof zone above the platen superheater

Ash deposit layer thickness: t = 0.8mm
e = assumed and adjusted
Ash deposit layer thermal conductivity: >‘oﬂ.plat = 0.07 p—
tofl plat m’K
Ash deposit coefficient: Eoch lat = P -0011——
P PN W
ofl.plat
Outer diameter of the tubes in the platen OD, i+ = 38-mm
P
superheater:
Average longitudinal spacing: sL.plat = 54.766-mm
Density of the platen superheater: Pgplat = Pg conv.pass
Inlet flow area: Aflow platin = Nmbesheets.plat'ST.plat'(hplat + bplat)
Agow platin = 530.103m’
Outlet flow area Aﬂow.plat.out = Ntubesheets.plat'ST.plat 'b'plat
2
Aflow plat.out = 411.137m
. m'gy m
Velocity of the flue gas at the Vg platin = = 0.997—
inlet of the platen: o Pg plat Aflow plat in o
Velocity of the flue gas at the . — g =i 334m
utlet of the platen: Vgplat.out = ™
. Pg plat Aflow.plat.out i
v, - +V
Average flue gas velocity in the platen: Vg plat = pplatin z i = 1.165 i
g s
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Appendix E. Gurvich thermal model

Thermal conductivity of the flue gas:
Interpolation values: T, = 1100°C A =117
T, = 1200°C N =126
T, -T
gplatave ~ 1 -2 W
A A+ M —N)|-10 T-—
g.plat [ 1 ( Ty~ Ty J( 2 lﬂ m K
>‘g.plat = 12-m—_
Kinematic viscosity: 5 5
—_5m ] —5m
Interpolation values: vy = 180-10 ™ vy = 211-10 ™
g § i
gplatave ~ "1
Vgplat —[V1+( Tz—Tl ](Uz—vl)
3 m2
Vg.plat =2x 10 T
Prandt number Pry =057 Pry =056
T T
gplatave ~ "1
Prgplat ave |:Pr1 ¥ ( T,-Ty J (Pr2 B Prl)}
Prg.plat.ave 0.56
Flue gas prandtl number: Prg.plat = (0.94 + 0'56'rH20.conv.pass)'Prg.plat.ave = 0.558

T

Correction factor of tube rows: Cz.plat =091+ 0_0125-(N ows.plat ~ 2) = 2.285

Relative transverse pitch:

Relative vertical pitch:

S

o) = —P= _ 30,079
D1
SL plat

oy = —2 = 1441
D1

Correction factor for geometric arrangement:

Flue gas composition and temperature

correction coefficient:

The flue gas side Reynolds number:

3712
2
Csplat = | 1+(2-01-3)- = = 0.198

Vg plat

Reg ot = = 22.129

University of Cape Town

Cyv.plat = 092+ 07261150 conv.pass = 0-985
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Appendix E. Gurvich thermal model

Flue gas Nusselt number:
Ntg pjar = 024(Reg ptar) " Prgpiae - -Cy piat Cis plat C
giplat =" g.plat g.plat z.plat s plat ~w.plat
Flue gas convection side coefficient h _ Dgplat N 178—_
g.conv.plat -~ “NUg plat = -
ODplat m2-K
Tube wall fouling emissivity: € = 07
- " Ts.plat.in ¥ Ts.plat.out
Average steam temperature: s.plat -~ 2
Ts.plat = 433.°C
Correction coefficient of tube diamter: C dplat = 1.04 Figure D8

Average free flow area on the steam side:
) 2
Aﬁ'.s.plat = Z 'lelat 'Ntubes.tubesheet.plat 'Ntubesheets.plat

Agr ¢ plat = 0595m”

Steam conditions for steam heat transfer coefficient

Density of the steam in the platen superheater:

1
Inlet density: Ps platin = ——y
. . Vsteam(ps.plat.in’Ts.plat.in’ > )
ensity P : 1
Outl td i ' S lat.Ou‘t — o "mooun o owm
© . ‘steam(ps.plat.out’Ts.plat.out’ > )
: Psplatin * Ps plat.out kg
Average density: ps.plat = - = 77_223;

Dynamic viscosity of the steam in the platen superheater:

Inlet dynamic viscosity: Hs platin = p‘steam(ps.plat.in ’Ts_plat,in’"" R ,"")
Outlet dynamic viscosity: ~ Hs.plat.out = ”steam(ps.plat.out Tsplatout> » o ’“)
0 o+ &
s.plat.in s.plat.out — 2697 x 10 )g
2 s-m

Average dynamic viscosity: Hs plat =

Thermal conductivity of the steam in the platen superheater:

o ww nn)

Ty T TTTIT Y T SECT D.TTEN  TETRE D Wiategtegt

woown wm mv)

Outlet thermal conductivity: >‘s.plat.out = >‘steam(ps.plat.out ’Ts.plat.out’ I

Average thermal conductivity: >‘s.plat =
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Appendix E. Gurvich thermal model

Specific heat of the steam in the platen superheater:

Inlet specific heat: Cps.plat.in = Cpsteam(ps.plat.in ’Ts.plat.in T ")
Outlet specific heat: Cps plat out = Cpsteam(ps.plat.out Tsplatout:” o ’")
Cps platin T CPs plat
erage spec _ Psplatin* Popaton oo 10
Average specific heat: Cps.plat = ” = 4.852-kg.K
-
Steam velocity in the steam: Vs plat = e = 1099
Aff s plat s plat &
P v, -ID, o
Reynolds number of the steam: Res.plat = il o = 8.181x 10’
Hs plat
Cpg plat ™
Prandtl number of the steam: Prs_p]at = e Rl = 1464
>‘s.plat
Heat transfer coefficient from tube wall to steam:
Nusselt number on the steam side: N“s.plat = 0.023 'Cdplat'Res.platO.s'Prs.platOA
Heat transfer coefficient on the steam side: hs.plat = Nus.plat'm = 5148--‘;—
I:Dplat m~-K
Platen superheater configuration factor:
S S
L plat . T.plat
= 1441 eplat = = 571.5-mm Figure D5
plat
*plat.conf 1

Platen superheater total heating area:

Ag.plat.conv = Xplat.conf 'hplat 'bplat 'Ntubesheets.plat 2
Fouling layer temperature of tube:

1 J.(Qr.plat i Qc.plat)

T =T +|E +
w.plat s.plat ash plat
P P ( P hs.plat A

g.plat.conv

Ty plat = 970465-C

Radiation heat transfer coefficient: ~ .
4
1— Tw.plat
b N Al | 3| \Tssheme 22—
radplat = 90 | “g.plat”*g.plat.ave - = :
g 2 &P & 1— Tw.plat m2-K
i i Tg.plat.ave Il
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Appendix E. Gurvich thermal model

Utilization coefficient of the panel: (g.plat =085

Flue gas side heat transfer coefficient:

(ﬂ-onplaI ] -
h = § | —h +h = 192219 ——
plat plat .conv.plat -rad plat
g.p e ET R g radp =
Overall heat transfer -
. o s gPp
coeffiecient for the platen Uplat = 3
superheater: 1.plat 1
1+|1+ Q . F:ashplat T _h 'hg.plat
c.plat s.plat
W
Uplat = 46.803-T
m~-K
Maximum temperature difference: ATmax.plat =Tgg - Ts.plat.in = 884.52-A°C

Minimum temperature difference: ATmin.plat f= Tg_p]at.out - Ts.plat.out = 579.07-A°C

Logarithmic mean AT a ATmax.plat - ATmin.plat
; : LMTD plat -~
temperature difference: p ATmax.plat

ATL\TD plat = 721.044K

Convective heat

transferred by the panel: Qc.plat.calc = Vplat 'Ag.plat.conv ‘ATLMTD.plat
Qe plat.cale = 147.585-MW
Error: E““Q.c.plat = Qc.plat . Qc.plat.calc

E““Q.c.plat = -2.168- MW

In this case since the heat transferred into the steam is known from the given conditions in the
C-schedule then it is not assumed first and then corrected by iteration. Some terms still need to
be adjusted on the calculated values of convection and radiation heat transfer to reduce the
error

Heat absorbed by the waterwalls around the platen superheater

Diameter of tubes in the water wall: OD,, = 63.5mm
Spacing of the waterwalls: Sww = 72.6mm
Distance from the platen 5 o i
superheater backside to furnace noseplat

nose:

Side water wall heating around the platen:
0.57-ODyy, + ( Sy — OD

Aww.plat = hikad 5 ww) [ otat (bptat + dnose plat) |2 = 377.119m’

ww

University of Cape Town

237



Steam conditions
Pressure:

Water temperature of water
wall at palten side:

Average temperature difference of heat
transfer:

Error on heat transferred by water wall:

Appendix E. Gurvich thermal model

Pww.out = 19-01MPa

T T

steam(pww.om’ > )

361.51-°C

s.ww.plat =

Ts.ww.plat =

AT, g =T T

gplatave ~ *s.ww.plat

AT,y = 803.28K

wa.plat.calc = Uplat 'Aww.plat 'ATs.plat

,  Remember to
Quvw plat cale = 14178 MW ital

guess value

Error ww.plat = Quww.plat ~ Qww.plat.cale

E"“Q.ww.plat = -251.11W

Roof infront of the platen superheater

Length of the roof in the furnace:

Area of roof in furnace

Relative height of furnace roof cover:

Thermal load distribution coefficient:

Furnace radiation heat flux absorbed
by furnace roof cover:

Furnace radiation heat absorbed by
the furnace roof cover:

University of Cape Town

Lroof_ﬁmce = 2056mm

Asoof farnace = Wumace Lroof famace
Aroof farnace = 49'35m2
xi= 1

Nroof furnace -= 0-633
”roofﬁn’nace'qr.ﬁnnace'm'coal

Qroof farnace =
Aﬁlmace
kW
Qroof farnace = 96'163'_2
m

ﬁplat “Qpoof furnace Aroof furace

Aroof furnace = .
m 5al

kJ

Groof furnace = 44'727-k_g
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Mass flow rate of water/steam in the
water walls:

Primary desuperheating water flow rate:

Mass flow are of water/steam in the
roof cover above the furnace:

Steam enthalpy increment of furnace
roof cover:

Appendix E. Gurvich thermal model

m' = 487.841 ke

S.Ww

Al T . o Al
M roof furnace =~ Psww ~ Psds

W' o2l “Groof furnace
m's v oof furnace

Ahs.roof.fm'nace =

kJ
Ahg roof fanace = 893 k_g

Saturated steam temperature of
the drum outlet:

Quality of the steam out of the drum:

T roof fumace in == 361°C

X dgrum out = 0-96
Dry saturated steam enthalpy of drum outlet:

by roof furnace.in = hsteam("” +Ts roof fumnacein> ’xdﬂnn.out’"")

kJ

by roof farnace in = 2442.58 k_g

Furnace roof steam pressure: Pgrum = 19.01MPa

Steam enthalpy of furnace roof inlet to the roof of the platen superheater:

hs.roof.ﬁn'nace.out = hs.roof.fumace.in + Ahs.roof.fumace

kJ
by v oof furnace.out = 2451.33 k_

g
Heat absorbed by the roof around the platen superheater
Steam temperature of roof T v g " w
inlet into the roof of the s.roof plat.in -~ steam(pdrum > »Ysroof furnace.out ° )
platen superheater:
L sodbplaila = I61.515:°C
Steam enthalpy increment ) Qroof.plat kJ
Ah, folat = —————— = 19.994.—
of platen superheater roof: S.TOOLp W' o of furnace kg
Steam enthalpy at the outlet of i o e Ah
platen superheater roof s.roof plat.out -~ “s.roof furnace.out + s.roof plat
kJ
hs.roof.plat.out = 2471.53‘1(—g
Steam temperature at the T =T " -
outlet of platen s.roof plat.out = steam(ps.Pn'mSH.in > »Ysroof plat.out » )
superheater roof:
Ts.toof.plat.om = 361.556-°C
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Appendix E. Gurvich thermal model

i gl )

Average steam temperature - _ “sroofplatin ™ “s.roof plat.out

at the roof of the platen: s.roof plat.ave -~ >
Ts.roof.plat.ave: = 361.535-°C

Average temperature AT - T e

difference of heat transfer on roof plat -~ g plat.ave s.roof plat.ave

the roof of the platen

superheater: ATroof.plat = 803.26-A°C

Geometry for heat transfer area calculation
Roof tube spacing:

Outer diameter of the tubes in the roof:

Roof tube length within the
platen superehater zone:

Width of the furnace:

Sioof = 73mm
OD, ¢ = 70mm
Lroof.plat = bplat + 60056.])131 = 6.993m

Wiyrnace = 24-003m

Heating area of the roof tubes above the platen superheater:

OS“ODrOOf + (SI'OOf o ODroof)

7

Aroof plat =
o Sroof

{ 2
'(“ﬁmace'l‘roof.plat) = 259.725m

Qroof.plat.calc = Uplat 'Aroof.plat 'ATroof.plat

Qroof.plat.calc =9.764-MW  pemember to

update initial
guess value

Error of the heat transferred into the E"orQ.roof.plat = Qroof.plat = Qroof.plat. —

roof of the platen superheater:

4,
E““Q.roof.plat =-1046x10 W

Absorbed convection heat of furnace roof at
panel zone:

University of Cape Town

Q\v\v_platlcalc = 14.178- MW

Error Q.ww.plat = —-025kW

Qroof.plat.calc = 9.764- MW

E““Q.roof.plat = —10461{“'

TS-Plat.out_calc = 1058.47-°C
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Appendix E. Gurvich thermal model

E.6. Cascading direct radiation through the Final superheater

Direct radiation into the final superheater: Qad direct FSH = 27-078-MW

Cascading the energy over the superheater rows:

: 2
S S
\ OD OD OD i gl
a(OD’ST) e B s asﬁ{_] + [ ] -1

280 8¢ Sy op) ~ oD
Outer diamter of the tubes: ODggy = 44.5mm
Pitch across depth: St FsHg = 0-8571m
Fraction of energy from the first row: apgyH = a(ODFSH, ST.FSH) = 0.08
Number of rows: Nrows.FSH =234

Direct radiation energy consumed per row:

Qraddirect.FSH.casc = |for i€ 0"Nrows.FSH =
%, < 2FSH
i—-1

«||1- -a fi>0
a, ZO 2% |[aFsH
i=

Qrad direct FSH'%

7
Rows 1-8: Qrad direct FSH.1to8 = Z Qracldirect.I-TSH.casci = 13200.81-kW
i=0
15
Rows9-16:  Qpaq drect FSHOt016 = ), rad direct FSH casc, = 6765 18kW
i=8
24
Rows 17-25: Qrad direct FSH.17t025 = Z Qraddirect.l-’SH.casci = 3759.21-kW
i=16
33

Rows 26-34: Qrad direct FSH.26t034 = Z Qraddirect.FSH.casci = 1772.09-kW
i=25
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Rows 35-43:  Qpaq direct FSH.35t043 -

Rows 44-52:  Qad direct FSH.44t052 =

Rows 53-60:  Qpaq direct FSH.53t060 =

Rows 61-68: Qa9 direct FSH.61t068 =

E.7. Final superheater geometry

Area of heating surfaces:

Outer tube diameter:

Transverse pitch:

Longitudinal pitch:

Thickness of the outer fouling layer:

Number of tubes per tubesheet:

Number of tubesheets:

Height at the inlet of the FSH:

Height at the outlet of the FSH:

Final superheater depth:

University of Cape Town

43-1

Z Qracldirect.l-TSH.casci = 835.36-kW

i1=35-1

52-1

Z Qraddirect.l-TSH.cas.c:i = 393.79.kW

1=44-1

60-1

Z Qracldirect.FSH.casci
1=53-1

68-1

Z Qracldirect.FSH.casci
i=61-1

Appendix E. Gurvich thermal model

171.2. kW

= 87.74 kW

, 2
Apgy = 3660m

ODFSH = 44.5-!!]111

toﬂ.FSH = 0.9mm

Niubes tubesheet FSH = 34

Niubesheets FSH = 28

hpsH out = 71305mm — 57018mm

bFSH = 1958.5mm + 1962mm
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E.8. Final superheater thermal calculations

Inlet flue gas temperature: Tg.FSH.in = Tg.plat.out = 1059.07-°C
Outlet flue gas temperature: Tg.FSH.out = 942.608 °C

Inlet steam temperature: T psHin = 473-581-°C

Inlet steam pressure: PsFSHin = 17-88-MPa

Outlet steam temperature: T, FSH out = 340-°C

Outlet steam pressure: Ps FSH out = 17-35-MPa

Steam mass flow rate: m'y gy = 309.763 ke

S

Heat transfer to the surroundings walls Quw.FsH = 23-234MW  trial and error

Exit enthalpy of the final superheater:

" (™' hg plat out.cale * Qrad direct FSH ~ QFSH ~ Quw FSH)
FSHout = .
. me

kJ
— hg FSH out = 116676
Flue gas exit temperature calculation

Initial guess value: T = 1059.07-°C

g.plat.out
(D= hfg(T) - hg.FSH.out
Tg.FSH.out.calc == root (f(Tg.plat.out) ’Tg.plat.out) = 942.515-°C

Errorr o FSH out = Tg FSH.out ~ Tg FSH.out.cale = 0-093-A°C

Constants
Emissivity of fouling on the tube wall SoflFSH = 0.7
) W
Thermal condutivity of the outer fouling layer: Nofl FSH = 007 .
=9 W
Stefan- Boltzmann constant: op = 3.67x10 - 2.4
m~-K
Radiation constant: Cra d= 3.6
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Calculations
Flue gas pressure:
Flue gas density:

Average flue gas temperature:

Average temperature difference
of the working fluid:

Ash deposition coefficient of the heating surface:

Outer diameter of the fouled layer:

Effective radiation layer thickness
(Mean beam length):
Lmb FsH = Crad’

Gas coefficient:
0.78 + 1'6'rH2O.conv.pass B

Appendix E. Gurvich thermal model

Pconv.pass = 100-kPa
kg
Pg.conv.pass = 1'328_3
m
 Tgrsrin+ T FSH out
Tg.FSH.ave i 2

Ty FSHave = 1273.99K

Ts FsHin * Ts FSH.out
2

T FSHave =
T FSH.ave = 306.79-°C

tofl FSH
Nofl FSH

Eash FSH =

4 m2-K
Eash.FSH =1286x10" T

OD, rsH = ODgsH *+ 2t FSH

ODofiFsH [ 4 STFSHSLFsH 1

0.1 -[l - 0.37-

4 2
OD.q FSH

ToFSHave | . _5 1

s

k = ]
¢ FSH : ;
Lub FSH 1000-K m Pa
Tn.convpass T
6 1
k =995%x10 -
B m-Pa
1
K¢ FSH = Kg FSH Tn.conv.pass Pconv.pass = 0-234;
m3
4300 p o
Fly ash coefficient: ke _ BERVDRS 10- 5 1
aFSH - N — e

[(

kszFSH = 7.655 % 10— &

1
Kfa FSH = kfahFSH'“facouv.pass'pcou\«'.pass = 0.268 —

University of Cape Town

Tg.FSH.ave
K

%
pm

15}

m-Pa

>

m
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. . . ) 1
Extinction coefficient of the flue gas: Kgsyg = Kg.FSH + K¢, psyg = 0501 ;
Flue gas emissivity: egFsH = 1~ exp(—KpsyLup FsH)

CoflFSH T 1
_ Frad_I-7SH.Flownex = (— 'ag.FSH = 0315

2

Outside radiation heat transfer coefficient

_ 63mm + 8.8mm + 11mm
Average thickness of the tube: 'FSH.ave = 3

Average inner diameter: IDpgy = ODggy — 2tESH ave = 27-1-mm

Average free flow area on the steam side:

_w 2
AFsFSH = Z'IDFSH ‘Niubes tubesheet FSH Ntubesheets FSH

AFSFSH = 0.549m”

Steam conditions for steam heat transfer coefficient
Thermal condictivity of the steam in the final superheater:

Inlet steam thermal

e N FSHin = Nteam(PsFSHin - TsFsHa """ ")
W
. = 0.084.—
s FSH.in K
Outlet steam thermal . o o e e
e N FSH.out = Nsteam(Ps FSH.out - Ts FSH.out:" """ )
%
= 0.087. —
s FSH.out oK
.+
Average steam thermal X FSH ave = NS FSHin * S FSH ou = 0.086i
o ; : 2 m-K
conductivity:

Density of the steam in the final superheater:
1

Inlet density: PsFSH.in = = o 5
¢ Veteamn (Ps FSHLin - TsFSHin ™ ™ > )
1

Outlet density: Ps FSH.out = S————
¢ Vsteam (ps.FSH.out’Ts.FSH.cmt’ pis D )
; PsFSH.in © Ps FSH.out kg
Average density: Ps FSH ave = > = 57.697—3—
m
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Dynamic viscosity of the steam in the final superheater:

Inlet dynamic viscosity: HsFSH.in = “steam(ps.FSH.h’Ts.FSH.hl’ o |3

Outlet dynamic ,_ T R —

WS Ms FSH.out = Msteam (Ps FSH.out - Ts FSH.out- )
Average dynamic _ Hs FSH.in T Ms FSH.out — 2.994x 10 5 kg
viscosity: lJ‘S.FSI’I.Z"’C - 2 - - Sih

Specific heat of the steam in the final superheater:

Nalk opaciic hni PsFSH.in = CPsteam(Ps FSHin - Ts FSHin> "> " ")
Outlet specific heat:  Cp; popy oyt = Cpsteam(ps.l-TSH.out TsFSHout: ’")
Cp .+ Cp
Average specific heat:  Cp, psH ave = e = S = 3.102-%
g.
"
Steam velocity in the steam: Vs FSH ave = sEHg = 16.09 =
Af s FSH Ps FSH.ave s
p oV t
Reynolds number of the steam: Re. psyg = sFSHave Vs FSHave IDFSH

Prandtl number of the steam:

Steam side Reynolds number:

Convective heat transfer coefficient on the steam hs.FSH =

side:

Duty of the final superheater:

Temperature of the outer fouling layer:

TofiFsH = Ts FSH. ave

University of Cape Town

Hs FSH.ave
Re, pspy = 8403 x 10°

CPs FSH.ave Ms FSH.ave

PrS.FSH = = 1.083

As FSH.ave

. 0.8 0.4

Nug poyy = 0.023-Reg pgyy — -Prg gy

Xs FSH.ave NUs FSH

Dgsy
3 W
m~-K

Qpsy = 107.358- MW

Q

sFSH ) \ AFSH
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Radition heat transfer coeffiecient:
4

. <+
_ SoflFSHY 1) ToFSHave ~ ToflFSH

BgradFSH = 90| — 5 |SgFSH| 7 -
gFSH.ave ~ “oflLFSH

-
hg rad FSH = 82941 ——
m°K
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Appendix F. Inputs for the real plant data cases

APPENDIX F. INPUTS FOR THE REAL PLANT DATA CASES

F.1. Inputs for the full load case

Parameter Units Full load
Excess air % 20
Ingress air % 7
Unburnt carbon per kg of coal % 0.8
Coal CV (HHV) MJ/kg 15.07
Feedwater flow rate kg/s 428.6
Cold reheat flow rate kg/s 411.2
EC water inlet temperature °C 247.5
EC water outlet temperature °C 293.1
Drum outlet temperature °C 352
RH 1 steam inlet temperature °C 334.1
RH 2 steam outlet temperature °C 534.9
ATT 1 spray water temperature °C 247.5
ATT 2 spray water temperature °C 247.5
ATT 3 spray water temperature °C 148.6
EC flue gas outlet temperature °C 397.9
Ambient air temperature °C 44
AH air outlet temperature °C 284.4
FW inlet pressure MPa 16.9
Drum pressure MPa 16.9
SH 2 steam inlet pressure MPa 16.45
SH 3 steam inlet pressure MPa 16.3
SH 3 outlet pressure MPa 16.2
RH 1 inlet pressure MPa 3.7
RH 2 outlet pressure MPa 3.7
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F.2. Inputs for the 65% full load case

Parameter Units 65% full load
Excess air % 20
Ingress air % 7
Unburnt carbon per kg of coal % 0.8
Coal CV (HHV) MJ/kg 15.07
Feedwater flow rate kg/s 262.4
Cold reheat flow rate kg/s 251.7
EC water inlet temperature °C 247.5
EC water outlet temperature °C 2733
Drum outlet temperature °C 413.4
RH 1 steam inlet temperature °C 323.1
RH 2 steam outlet temperature °C 531.8
ATT 1 spray water temperature °C 247.5
ATT 2 spray water temperature °C 247.5
ATT 3 spray water temperature °C 148.5
EC flue gas outlet temperature °C 369.7
Ambient air temperature °C 44
AH air outlet temperature °C 268.2
FW inlet pressure MPa 16.1
Drum pressure MPa 15.9
SH 2 steam inlet pressure MPa 15.9
SH 3 steam inlet pressure MPa 15.9
SH 3 outlet pressure MPa 15.9
RH 1 inlet pressure MPa 2.4
RH 2 outlet pressure MPa 2.4
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